

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bennett, Patrick

Doctoral Thesis Essays in Education, Crime, and Job Displacement

PhD Series, No. 18.2016

Provided in Cooperation with: Copenhagen Business School (CBS)

Suggested Citation: Bennett, Patrick (2016) : Essays in Education, Crime, and Job Displacement, PhD Series, No. 18.2016, ISBN 9788793483019, Copenhagen Business School (CBS), Frederiksberg, https://hdl.handle.net/10398/9308

This Version is available at: https://hdl.handle.net/10419/208972

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/3.0/



WWW.ECONSTOR.EU

COPENHAGEN BUSINESS SCHOOL

SOLBJERG PLADS 3 DK-2000 FREDERIKSBERG DANMARK

WWW.CBS.DK

ISSN 0906-6934

Print ISBN: 978-87-93483-00-2 Online ISBN: 978-87-93483-01-9



ESSAYS IN EDUCATION, CRIME, AND JOB DISPLACEMENT

Patrick Bennett ESSAYS IN EDUCATION, CRIME, AND JOB DISPLACEMENT

The PhD School of Economics and Management

CBS K COPENHAGEN BUSINESS SCHOOL

PhD Series 18.2016

Essays in Education, Crime, and Job Displacement

Patrick Bennett

Supervisors: Birthe Larsen and Lisbeth la Cour

PhD School in Economics and Management Copenhagen Business School Patrick Bennett Essays in Education, Crime, and Job Displacement

1st edition 2016 PhD Series 18.2016

© Patrick Bennett

ISSN 0906-6934

Print ISBN: 978-87-93483-00-2 Online ISBN:978-87-93483-01-9

"The Doctoral School of Economics and Management is an active national and international research environment at CBS for research degree students who deal with economics and management at business, industry and country level in a theoretical and empirical manner".

All rights reserved.

No parts of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or by any information storage or retrieval system, without permission in writing from the publisher.

Forward

This Ph.D. thesis has been written over the course of my doctoral studies in the Department of Economics at the Copenhagen Business School. I am very grateful for the financial support provided by the Copenhagen Business School throughout my Ph.D. First and foremost, I wish to thank my two supervisors, Birthe Larsen and Lisbeth la Cour, for their numerous comments and suggestions, for our co-authorship, and more generally for always being so supportive and always being available. I wish to thank Amine Ouazad not only for our co-authorship but for providing so much guidance and insight throughout our relationship. During 2014, I was fortunate enough to visit the Swedish Institute for Social Research (SOFI), and I am grateful to Matthew Lindquist not only for the invitation but for all his support and to everyone else who provided feedback and insight during my stay at SOFI. I also wish to acknowledge the financial support from Otto Mønsteds Fond and Christian og Ottilia Brorsons Rejselegat which made my visit to SOFI possible. I am very grateful to Marcus Asplund for his detailed comments and work on the first chapter of my thesis. I also wish to thank all of those who have taught me throughout my academic career both in the UK and Denmark.

I am very thankful for the thorough and invaluable comments I received during my Closing Seminar from Dario Pozzoli and Anna Piil Damm. Special thanks go to all of my colleagues at the Copenhagen Business School for the support at various seminars and workshops, but in particular Fane Groes for always helping with data questions, Moira Daly for always going the extra mile in helping me throughout my Ph.D., Jimmy Martínez-Correa for his support during the start of my Ph.D., and Dario Pozzoli for his encouragement throughout the last months of my Ph.D. I wish to also thank my Ph.D. colleagues not only for inspiring ideas and discussions but also for putting up with some of my less than desirable work habits—your presence will be missed. I also wish to thank the anonymous twins who revealed themselves at various stages in my relationships with them for providing anecdotal evidence on the twin lifestyle.

Finally, I wish to thank mother for (quite correctly) teaching me to always print, my father for paving the way for Bennett graduates in Economics, and both of them for all that they have done. Thank you to my grandparents for always looking out for me and their constant support throughout my academic career. A special thanks goes to all of my high school English teachers who, although they may never even see this, certainly deserve praise and appreciation for their tireless and underpaid educational efforts. Special thanks also go to Fabio Aricò for his words of encouragement, support, and assistance over the years, without which I wouldn't be anywhere close to where I am today. Lastly, but certainly not least, I wish to thank my wife Anna for her endless support and for dealing with the countless late nights, weekends, and trips abroad required throughout my Ph.D. And to anyone who I mistakingly left out, please know I appreciate your efforts and please accept my sincere apologies for the omission.

Abstract

With a limited budget and resources, governments must decide how to allocate funds across a variety of factors which benefit society such as education, crime deterrence, and public safety. Each increase in spending on one area comes with the knowledge that this money cannot be spent on social problems in another area. As such, externalities and unexpected spillover effects impact the costs and benefits of public spending to society and may have large and meaningful implications on how to most effectively allocate resources across a multitude of outcomes. For example, an increase in education corresponds to an increase in the opportunity cost of engaging in criminal activity, decreasing the probability an individual commits crime. Likewise, loss of employment decreases the opportunity cost of engaging in criminal activity, increasing the probability an individual commits crime. Discrimination towards immigrants can impact their employment prospects which, in turn, impacts their decision to further pursue education. Identifying how these individual level factors have an impact on society is key to informing and designing effective public policy.

This Ph.D. thesis, entitled "Essays in Education, Crime, and Job Displacement", analyzes the determinants and social implications of these three factors. While independent, each essay within this thesis examines the impact of factors such as education, in terms of reduced crime, job loss, in terms of increased crime, and discrimination, in terms of its impact on the educational attainment of immigrants, on society.

The first essay of my thesis, "The Heterogeneous Effects of Education on Crime: Evidence from Danish Administrative Twin Data" examines whether education is equally effective in reducing crime for everyone. Education reduces crime, but this paper is the first to directly examine heterogeneous effects, in particular how the crime reducing capabilities of education depend on specific individuals and factors. I make use of twins contained in detailed Danish Register Data to control for characteristics, both observable and unobservable, which are common between twins. I focus on twin data as identification using twins provides many advantages—giving the freedom to explore the impact of many educational qualifications across the entire education distribution and directly estimating the effects of juvenile crime on educational attainment to examine reverse causality. I find that family factors are important, where education reduces crime for males of low educated families, with less effects seen for individuals coming from highly educated families. Environmental factors are found to be less important—education lowers crime irrespective of the levels of crime in childhood neighborhoods. I find that the completion of high school lowers crime considerably while, contrary to expectations, the completion of vocational education is found to have no crime reducing effects. The role of juvenile crime is examined in great detail, revealing that taking into account participation in juvenile crime is important but that education reduces adult crime above and beyond what is explained through juvenile crime. My results imply that these specific "at risk" individuals obtain education which is less than socially optimal, as, in addition to the private benefits to education, the social benefits of education are large.

The second essay, entitled "Job Displacement and Crime" investigates, together with Amine Ouazad, the individual level impacts of job loss on crime. While unemployment measured at an aggregate level causes crime, the extent to which a transition into unemployment increases crime is less clear, as previous literature has focused on measures of unemployment at the state or county level. We identify the impact of sudden and unexpected job loss on crime, using detailed police, employer, and employee data. Following Jacobson, LaLonde and Sullivan (1993), we examine high tenured workers with strong labor market ties to their firms. For these individuals, job loss occurring in a mass layoff event is likely sudden and unanticipated. We find that displaced individuals are significantly more likely to be convicted in the time following job loss, but importantly, not in the time before job loss. These effects are primarily seen for property crimes and are concentrated for individuals at the lower ends of the educational distribution. These effects are long lasting, particularly for individuals with less than high school education. Individuals educated to the university level or beyond are more resilient, in terms of criminality, to job loss. We examine a possible intergenerational effect where father's job loss may impact the criminality of children, finding some evidence that sons are marginally more likely to commit crime in the short-run following father's displacement. We see sizable, significant, and long lasting effects for individuals who live alone. Our results are robust to changing our mass-layoff criteria including increasing the number of employees who must lose their job to cause a mass-layoff event and altering how we define a mass-layoff event. We argue that neither employers nor employees fully internalize the social costs of job loss, which justifies an active role of policies which incentivize unemployed to transition back into formal employment or additional taxation of employers.

The final essay of my thesis, "Negative Attitudes, Network and Education", investigates, to-

gether with Lisbeth la Cour, Birthe Larsen, and Gisela Waisman, what factors can explain the educational gap that exists between natives and immigrants. We examine, both theoretically and empirically, the importance of two specific factors: negative attitudes towards immigrants and networking amongst immigrants. Theoretically, we formulate a Becker-style taste discrimination model within a search and wage bargaining setting where the educational decision of natives and immigrants is endogenous. We show that the education an immigrant obtains depends on negative attitudes, which directly influence their employment prospects. If all immigrants are equally affected by discrimination, immigrants obtain less education than natives. If only low skilled immigrants are affected by negative attitudes, immigrants obtain more education than natives to improve their employment prospects. We find that more immigration increases the fraction of educated immigrants via networking which also improves immigrant employment prospects. Empirically, we analyze the educational decision of young immigrants, specifically whether they decide to attend high school. We find evidence that negative attitudes against immigrants increase the likelihood male immigrants attend high school, supporting the case when discrimination is against only low skilled immigrants. We find evidence supporting our theoretical findings that networking amongst immigrants, in the form of a higher fraction of own nationality immigrants for females and a higher fraction of own nationality immigrants who are employed for males, increases the likelihood young immigrants attend high school. Due to the fact that immigrant's can selectively locate, we examine how our results change taking into account unobservable factors which could drive these decisions. We show that, under reasonable assumptions regarding the role of unobservables, that while unobservables do explain a portion of our results, they are unable to completely explain the positive impact of negative attitudes on education we see for males. Our results are also robust to excluding those families who have recently relocated and are explained almost entirely by an impact on 1st generation immigrants.

Resumé (Abstract - Danish)

Med et begrænset budget og ressourcer må regeringer beslutte sig for hvordan de vil fordele midler til forskellige faktorer, så som uddannelse, bekæmpelse af kriminalitet og sikkerhed. Enhver stigning i udgifter til et område betyder at disse penge ikke kan bruges på sociale udfordringer på et andet område. På den måde har eksternaliteter og uforudsete spill-over effekter indvirkning på omkostninger og gevinster fra offentlige udgifter for samfundet og kan have længerevarende og betydningsfulde implikationer for hvor effektivt ressourcerne er allokeret. For eksempel, en stigning i uddannelse svarer til en stigning i alternativomkostningerne ved at engagere sig i kriminel aktivitet, hvilket sænker sandsynligheden for at et individ bliver kriminel. På samme måde, sænker et beskæftigelsestab alternativomkostningerne ved at en person begår kriminalitet. Diskrimination mod indvandrere kan påvirke deres beskæftigelses chancer, hvilket videre kan påvirke deres beslutning om at tage videre uddannelse. En identifikation af hvordan disse individuelle faktorer kan påvirke samfundet er nøglen til at informere og designe effektiv offentlig politik.

Denne Ph.D. afhandling, 'Essays in Education, Crime, and Job Displacement' (Essays om uddannelse, kriminalitet og arbejdstab) analyserer determinanter og sociale implikationer af disse tre faktorer. Selvom de er uafhængige, undersøger hvert essay hvorledes faktorer som uddannelse, i form af reduceret kriminalitet, arbejdstab, i form af større kriminalitet, og diskrimination, i form af dens påvirkning af indvandreres uddannelsesniveau, påvirker samfundet.

Det første essay af min afhandling: 'The Heterogeneous Effects of Education and Crime: Evidence from Danish Administrative Twin Data' (Heterogene effekter af uddannelse på kriminalitet: Evidens fra Dansk register tvillingedata) undersøger hvorvidt uddannelse er lige effektivt for alle med henblik på at reducere kriminalitet. Uddannelse begrænser kriminalitet, men dette papir er det første som direkte undersøger heterogene effekter, dvs. hvorvidt uddannelse begrænser kriminalitet afhænger af specifikke individer og faktorer. Jeg benytter mig af tvillingepar i detaljeret dansk register data til at kontrollere for karakteristika, både observerbare og ikke-observerbare, som er fælles for tvillinger. Jeg fokuserer på tvillingedata da identifikation via tvillinger har mange fordele – hvilket giver friheden til at udforske implikationen af forskellige uddannelseskvalifikationer - og estimerer direkte indvirkningen af ungdomskriminalitet på uddannelsesniveau med henblik på at undersøge om der er omvendt kausalitet. Jeg finder at familiære faktorer er vigtige, hvor uddannelse reducerer kriminalitet for mænd fra lavt-uddannede familier, hvorimod der er mindre effekt for individer fra højtuddannede familier. Samfundsmiljømæssige faktorer findes at have mindre betydning – uddannelse reducerer kriminalitet uafhængig af kriminaliteten i nabolaget, hvor personen voksede op. Jeg finder at en studentereksamen reducerer kriminalitet betydeligt, hvorimod, i modsætning til hvad vi ville forvente, at en uddannelse som faglært ikke sænker kriminalitet. Betydningen af ungdomskriminalitet undersøges detaljeret og det afsløres at ungdomskriminalitet er vigtig men at uddannelse reducerer kriminalitet mere end hvad kan forklares ved ungdomskriminalitet. Mine resultater betyder at disse individer i risikogruppen opnår et uddannelsesniveau som er under det, som ville være samfundsmæssigt optimalt, da udover de individuelle gevinster ved at uddanne sig, er der tillige nogle store samfundsmæssige gevinster.

I det andet essay: 'Job Displacement and Crime' (Arbejdstab og kriminalitet) undersøger jeg sammen med Amine Ouazad, de individuelle effekter af arbejdstab på kriminalitet. Hvor arbejdsløshed på aggregeret plan medfører kriminalitet, er det mere uklart om et job tab og dermed transition til arbejdsløshed hæver kriminalitet på individ niveau. Dette er tilfældet da tidligere litteratur har fokuseret på arbejdsløshed på lande eller regionsplan. Vi identificerer påvirkningen af et pludseligt og uventet arbejdstab på kriminalitet, ved at benytte os af detaljeret kriminalitetsregister, virksomhedsdata og individ og dermed ansættelsesdata. Idet vi følger Jacobson, Lalonde og Sullivan (1993) ser vi på individer med lang arbejdserfaring i den virksomhed de er ansat i. For disse personer er et arbejdstab som følge af en massefyring sandsynligvis pludselig og uventet. Vi finder at disse personer, som mister deres job, med større sandsynlighed bliver dømt for kriminalitet i tiden derefter, og dette selvom de ikke har begået kriminalitet i tiden før de mistede deres arbejde. Disse effekter ses primært for ejendomsforbrydelser og er koncentreret i denne lave ende af uddannelsesfordelingen. Disse effekter er langvarige, især for individer med mindre uddannelse end studentereksamen eller lignende. Personer med længere uddannelse, som pludseligt mister deres arbejde er mindre påvirkede med hensyn kriminalitet. Vi undersøger en mulig generationseffekt, hvor farens beskæftigelsestab kunne tænkes at påvirke børnenes kriminalitet, og vi finder noget evidens for at sønner begår marginalt mere kriminalitet lige efter farens beskæftigelsestab. Vi finder betydelige og langvarige effekter for individer, som bor alene. Vores resultater er robuste overfor at ændre massefyringskriteriet således at flere skal have mistet deres job og ser på en ændring af andelen ansatte, der mister deres job. Vi argumenterer for at hverken arbejdsgiveren eller de ansatte fuldt ud internaliserer de sociale omkostninger ved beskæftigelsestab, hvilket retfærdiggør en aktiv rolle for politik med henblik på at intensivere transitionen fra arbejdsløshed tilbage i beskæftigelse og/eller en yderligere beskatning af virksomheder.

Det sidste essay: 'Negative Attitudes, Network and Education' (Negative holdninger, netværk og uddannelse) undersøger, sammen med Birthe Larsen, Lisbeth la Cour og Gisela Waisman, hvilke faktorer der kan forklare uddannelsesgabet mellem indfødte og indvandrere. Vi undersøger, både teoretisk og empirisk, vigtigheden af to specifikke faktorer: negative holdninger overfor indvandrere og indvandreres netværk. Teoretisk formulerer vi en Becker smagsdiskriminationsmodel i en søge-matching forhandlingsmodel, hvor uddannelsesbeslutningen for indfødte og indvandrere er endogen. Vi viser at det uddannelsesniveau, som indvandrere opnår afhænger af negative holdninger, da de direkte påvirker forventet beskæftigelse. Hvis alle indvandrere oplever samme diskrimination vil indvandrere opnå mindre uddannelse end indfødte. Hvis derimod kun lavt-uddannede indvandrere påvirkes af negative holdninger, vil indvandrere opnå mere uddannelse end indfødte, da de derved forøger deres beskæftigelseschancer og løn. Vi finder at mere indvandring forøger andelen af uddannede indvandrere gennem netværkseffekter, da det forøger deres beskæftigelseschancer og løn. Empirisk analyserer vi uddannelsesvalget for unge indvandrere med hensyn til om de fortsætter på gymnasiet eller lignende uddannelse. Vi finder at negative holdninger overfor indvandrere øger sandsynligheden for at mandlige indvandrere går i gymnasiet, hvilket stemmer overens med modellen, hvor diskrimination kun finder sted mod lavt-uddannede indvandrere. For kvinderne finder vi dokumentation for modellens forudsigelse at indvandreres netværk, i form af en højere andel af egen nationalitet øger sandsynligheden for at den unge indvandrer går i gymnasiet eller lignende. For de unge indvandrermænd finder vi at en højere andel ansatte af egen nationalitet øger sandsynligheden for at den unge indvandrer går i gymnasiet eller lignende. På grund af indvandreres mobilitet, undersøger vi hvordan vores resultater ændrer sig hvis vi tager hensyn til ikke-observerbare faktorer, der kunne påvirke disse beslutninger. Vi viser, at under rimelige antagelser om den rolle som ikke-observerbare faktorer spiller, kan de ikke fuldt ud forklare den positive indvirkning af negative holdninger på uddannelse, som vi ser for mænd. Vores resultater er også robuste i forhold til at udelade de familier, der er flyttet for nylig og forklares næsten udelukkende af en indvirkning på 1. generations indvandrere.

Contents

Forward	3
Abstract	5
Resumé (Abstract - Danish)	9
Introduction	15
References	18
Chapter 1 - The Heterogeneous Effects of Education on Crime: Evidence from	
Danish Administrative Twin Data	19
Chapter 2 - Job Displacement and Crime	75
Chapter 3 - Negative Attitudes, Network and Education	117
Conclusion	167

Introduction

This Ph.D. thesis is composed of three chapters and ends with a general conclusion for all three chapters. It should be noted that while all three chapters are independent research papers and can be read as such, they all address important topics within the field of Labor Economics. More specifically, all three chapters analyze the determinants and social implications of education, crime, and job displacement. The first chapter finds that education is not equally effective in lowering crime for everyone, and identifies specific individuals for whom we can expect education to have an impact on criminal activity. The second chapter finds that job displacement, sudden and unexpected job loss in a mass-layoff event, increases the probability of committing crime and that the magnitude and longevity of these effects depend on education, household factors, and post-displacement employment outcomes. The third chapter proposes two potential explanations of the educational gap that exists between immigrants and natives, negative attitudes towards immigrants and networking, finding that while networking amongst immigrants can reduce this gap, negative attitudes motivate immigrants to pursue further education and, as such, are unable to explain this gap in education.

All three of these chapters are motivated by and build upon existing literature and make unique and novel contributions within their respective fields. Over the last decade, economists have established that education has a negative and significant impact on an individual's propensity to commit crime. There are numerous studies establishing this fact (Lochner and Moretti 2004; Machin et al. 2011; Åslund et al. 2015; Hjalmarsson et al. 2015), which generally find that education has a negative and causal impact on crime. There are many reasons education is expected to lower crime, for example by increasing the opportunity cost of crime or by impacting individual factors such as instilling social values or increasing the patience of individuals. The motivation behind the first chapter is to recognize that while, overall, education reduces crime, that this may not be true for everyone and important heterogeneity in the crime reducing capabilities of education may exist below this overall effect. Similarly, the second chapter is motivated by the fact that unemployment and crime are related at an aggregate level (Raphael and Winter-Ebmer 2001; Gould et al. 2002; Öster and Agell 2007; Lin 2008; Fougère et al. 2009), but practically nothing is known about the individual level relationship between becoming unemployed and engaging in crime. Job loss can impact crime by lowering the opportunity cost of crime but also by having a psychological impact on the individual, and there are many reasons to believe the individual level and aggregate impacts may differ, in terms of both types of crime and the longevity of the effects. Finally, the third chapter combines and expands on two strands of existing literature: one which analyzes the impacts of discrimination on employment and wages (Mailath et al. 2000; Lang et al. 2005; Charles and Guryan 2008; Waisman and Larsen 2015) but without the additional analysis of an individual's educational decision and another which analyzes the importance of networking in securing employment (Calvó-Armengol and Jackson 2004; Andersson et al. 2009; Kramarz and Skans 2014). Negative attitudes may lead immigrants to obtain lower education, if immigrants are equally affected by discrimination, or may lead immigrant's to obtain more education, if only low-skilled immigrants are affected by education, while networking will increases education.

Within all of these strands of existing literature, a common underlying point of emphasis is the identification of a causal relationship between the explanatory variable and the outcome of interest. However, the methods used to estimate causal impacts differ across these fields. To examine the crime reducing effects of education, many studies have used changes in compulsory schooling laws; the first chapter of this thesis takes a different approach—controlling for characteristics which are common between twins. The previous literature on unemployment and crime has relied on exogenous changes at the state or county levels which impact unemployment but not crime; the second chapter relies on sudden and unexpected mass-layoffs at the individual level to generate exogenous changes in unemployment. The third chapter focuses on a sample of young immigrants for whom household location is plausibly exogenous to their educational decision and, additionally, provides evidence on the role of unobservable factors in explaining the findings.

All three chapters make use of detailed administrative population data from Denmark. Danish Register Data, which is maintained and provided by Statistics Denmark, is a panel dataset compiled from various administrative sources which completely covers the entire Danish population. The detail and depth of this data is useful for assessing the heterogeneous impacts of education on crime, enables the linking of matched employer-employee data to police data to analyze the impacts of job displacement on crime, and the institutional structure of the educational system of Denmark enables the analysis of the impact of negative attitudes on a non-compulsory decision to attend further education. All three chapters would not be possible using other non-Register based data sources where either information would not be detailed enough, data would simply not be available, or the timing of a crucial education decision would be either too early or too late.

- Andersson, F., S. Burgess, and J. Lane (2009). Do as the neighbors do: The impact of social networks on immigrant employment. IZA Discussion Papers 4423, Institute for the Study of Labor (IZA).
- Åslund, O., H. Grönqvist, C. Hall, and J. Vlachos (2015). Education and criminal behavior: Insights from an expansion of upper secondary school. IZA Discussion Papers 9374, Institute for the Study of Labor (IZA).
- Calvó-Armengol, A. and M. O. Jackson (2004). The effects of social networks on employment and inequality. American Economic Review 94(3), 426–454.
- Charles, K. K. and J. Guryan (2008). Prejudice and wages: An empirical assessment of becker's The Economics of Discrimination. Journal of Political Economy 116(5), 773–809.
- Fougère, D., F. Kramarz, and J. Pouget (2009). Youth unemployment and crime in france. *Journal* of the European Economic Association 7(5), 909–938.
- Gould, E. D., B. A. Weinberg, and D. B. Mustard (2002). Crime rates and local labor market opportunities in the united states: 1979–1997. *Review of Economics and Statistics* 84(1), 45–61.
- Hjalmarsson, R., H. Holmlund, and M. J. Lindquist (2015). The effect of education on criminal convictions and incarceration: Causal evidence from micro-data. *The Economic Journal* 125(587), 1290–1326.
- Kramarz, F. and O. N. Skans (2014). When strong ties are strong: Networks and youth labour market entry. *The Review of Economic Studies*.
- Lang, K., M. Manove, and W. T. Dickens (2005). Racial discrimination in labor markets with posted wage offers. *American Economic Review* 95(4), 1327–1340.
- Lin, M.-J. (2008). Does unemployment increase crime?: Evidence from u.s. data 1974-2000. Journal of Human Resources 43(2), 413–436.
- Lochner, L. and E. Moretti (2004). The effect of education on crime: Evidence from prison inmates, arrests, and self-reports. *American Economic Review* 94(1), 155–189.
- Machin, S., O. Marie, and S. Vujić (2011). The crime reducing effect of education. The Economic Journal 121 (552), 463–484.
- Mailath, G. J., L. Samuelson, and A. Shaked (2000). Endogenous inequality in integrated labor markets with two-sided search. *American Economic Review* 90(1), 46–72.
- Oster, A. and J. Agell (2007). Crime and unemployment in turbulent times. *Journal of the European Economic Association* 5(4), 752–775.
- Raphael, S. and R. Winter-Ebmer (2001). Identifying the effect of unemployment on crime. *Journal* of Law and Economics 44(1), 259–283.
- Waisman, G. and B. Larsen (2015). Wages, amenities and negative attitudes. Mimeo.

Chapter 1 - The Heterogeneous Effects of Education on Crime: Evidence from Danish Administrative Twin Data

The Heterogeneous Effects of Education on Crime: Evidence from Danish Administrative Twin Data

Patrick Bennett^{*}

March 2016

Abstract

Using administrative Danish Register Data to identify all twins aged 18-35 in 2000, this paper identifies heterogeneous effects of education on crime. Controlling for genetic and environmental factors, an additional year of education significantly lowers the probability of conviction for total, property, and violent crimes by 14%, 12%, and 19% for males. Estimation by parental education reveals family factors matter—education overwhelmingly lowers crime for children of low educated parents—while estimation by exposure to crime during childhood reveals environmental factors have less impact on the relationship between education and crime. Examining different educational programs reveals the completion of high school matters for crime reduction of males, while the completion of vocational training or university do not. These results are robust to correcting for the presence of dizygotic twins, directly estimating reverse causality between education and crime, and using data on incarcerations instead of convictions.

JEL classification: 12, K42

Acknowledgments: The author wishes to thank Birthe Larsen, Lisbeth la Cour, Paolo Buonanno, Fane Groes, Moira Daly, Matthew Lindquist, Kevin Schnepel, Marcus Asplund, Dario Pozzoli, Anna Piil Damm as well as seminar participants at the Copenhagen Business School and the Swedish Institute for Social Research (SOFI) and participants of the Economics of Crime workshop at the University of Copenhagen, the 2014 SMYE, the 2014 CEN Workshop, the 2014 DGPE, and the 2015 RES Conference and Symposium of Junior Researchers.

[°]Copenhagen Business School, Department of Economics, Porcelænshaven 16A, 2000 Frederiksberg, Denmark. pabe.eco@cbs.dk

1 Introduction

Education reduces crime—this fact has been established in a number of studies. This is true for both years of education as well as the completion of high school. This is also true for multiple crime types—total, property, and, in most instances, violent crimes. But is education equally effective in reducing crime for everyone? Given the vast differences in not only types of crime but also the motivating factors behind these crimes, there is good reason to believe education impacts the criminal behavior of specific types of individuals very differently.

This paper analyzes heterogeneity in the crime reducing capabilities of education using Danish twin data, and makes three important contributions to the literature. First, heterogeneous effects are examined across family factors such as parental education as well as environmental factors such as growing up in neighborhoods with high and low levels of crime. Second, it evaluates the importance of specific educational qualifications and programs for the crime reducing effects of education. Third, it expands upon Webbink et al. (2013) by providing more generalizable results which use administrative twin panel data rather than self reported survey data, analyzing both male and female twin pairs, and examining detailed crime types in addition to investigating heterogeneity along other dimensions.

Previous studies (Lochner and Moretti 2004; Machin et al. 2011; Meghir et al. 2012; Hjalmarsson et al. 2015) typically exploit changes in compulsory schooling laws to provide causal interpretations of the effects of education on crime. This paper takes a different approach—controlling for characteristics which are common between twins. As twins are genetically similar and are usually raised in the same environment as children, many unobservable factors which affect both education and crime are controlled for by comparing the outcomes of one twin to the other. Additionally, twin estimation is capable of estimating effects across the entire educational distribution of the population and not only for those at the lower end of the distribution.

Using within twin fixed effects estimation, this paper confirms the existence of significant negative effects of years of education on the probability of conviction for male twins of total and violent crimes, with marginally significant effects seen for property crimes. An additional year of education lowers the probability a male twin was convicted of any crime committed from 2001-2006 by 14%, of a property crime by 12%, and of a violent crime by 19%. Involvement in juvenile crime significantly increases an individual's probability of being convicted as an adult, but to less of an

extent than has been found previously (Webbink et al. 2013).

Having confirmed education reduces an individual's probability of conviction as an adult, the paper examines the presence of heterogeneous effects of education on crime. Family factors are found to be important—the effects of education on crime are large in magnitude for those from a low educated household and, with the exception of violent crimes, virtually non-existent for those with two highly educated parents. Environmental factors are found to be less important—education lowers crime irrespective of the levels of crime in the neighborhood twins are exposed to during childhood. Examining different educational qualifications reveals that the completion of high school is important in terms of crime reduction for males. However, contrary to expectations, the completion of vocational training geared directly towards professional employment has no effects on a male's propensity to engage in crime. When analyzing every available crime type for male twins, education significantly decreases participation in firearms and alcohol related traffic offenses, with marginally significant effects found for sexual and other crimes.

Isolating heterogeneous effects of education on crime which are causal in nature represents an empirical challenge. This relationship is complicated by the fact that education decisions are endogenous—more crime prone individuals are both less likely to pursue education and more likely to commit crime—and causality flows in both directions—participation in crime as a juvenile can directly affect the level of schooling an individual obtains. In addition to identifying causal effects of education on crime, there are many other advantages to using within twin estimation which enable the examination of heterogeneity in the effects of education on crime. Firstly, effects are identified over the entire population, not from those who comply with compulsory schooling reforms at the lower end of the educational distribution. Secondly, the analysis is not constrained to one specific educational change, giving the freedom to analyze the impact of multiple educational qualifications and time periods. Thirdly, estimation within twins enables the analysis of reverse causality between education and crime.

Despite the prominent use of twin studies within Economics,¹ there are also limitations to using twin data. Specifically, while monozygotic (MZ) twins are virtually genetically identical, differences in unobservable factors which affect both education and, in this paper, crime participation are

¹Twin studies have long been used in Economics, and particularly in the education literature, to estimate the returns to education (Ashenfelter and Krueger 1994; Ashenfelter and Rouse 1998; Isacsson 1999), the intergenerational transmission of education (Behrman and Rosenzweig 2002; Holmlund et al. 2011; Pronzato 2012; Lundborg et al. 2014), the impact of spousal education on earnings (Huang et al. 2009), and even in the portfolio choice literature (Calvet and Sodini 2014).

determined by more than just genetic factors, and these differences in unobservables could be what drive twins to obtain different education levels (Griliches 1979; Bound and Solon 1999; Sandewall et al. 2014).

While the results presented in this paper are subject to these criticisms, particularly as data on zygosity is unavailable, multiple steps are taken to ensure that unobservable differences between twins are accounted for. Importantly, twins raised in different households during childhood are excluded from the sample, as these twins are not exposed to similar environmental factors. Correcting for the potential bias introduced by the presence of dizygotic (DZ) twins as in Conley et al. (2006) and Holmlund et al. (2008) indicates that if zygosity were observable, the estimated effects of education on crime identified would be of a similar magnitude for MZ twins and, in most instances, remain statistically significant. Examining GPA differences for a subset of twins where GPA data is available reveals no major differences in the academic achievement of twins. Additionally, results are robust to using data on incarcerations instead of convictions, excluding twins with large differences in education, and directly estimating the reverse causality between education and crime.

The next section briefly outlines the reasons why education can affect crime. Section 3 details the existing literature on education and crime. Section 4 describes Danish Register Data, while Section 5 provides summary statistics. Section 6 outlines within twin fixed effects estimation and potential threats to this methodology. The baseline results of the effects of education on crime are reported in Section 7, while Section 8 examines the heterogeneous effects of education on crime. Section 9 details the robustness of the results and Section 10 concludes and provides a brief discussion of the results.

2 Reasons Education Can Affect Crime

2.1 Effects on Employment

Education can lower criminal activity by affecting an individual's labor market prospects, predominantly through increasing wages and increasing an individual's probability of employment.² Firstly, education builds human capital which leads to higher wages. Increased wages increases the opportunity cost of crime (foregone wages while incarcerated), thus reducing an individual's

²For a theoretical model, see Lochner (2011).

propensity to engage in criminal behavior. Secondly, if employers see educational qualifications as an indicator of potential productivity, education can increase the probability that an individual will be employed. Having a legal job reduces the financial need for illegal wages through crime, also lowering an individual's propensity to engage in crime. Lochner (2011) identifies that effects on employment and wages are the most prominent reasons why education can reduce crime.

2.2 Effects on the Individual

Education can also have direct effects on the individual, shifting individual preferences away from crime. Firstly, Lochner and Moretti (2004) argue that education alters an individual's preferences, leading to increased risk aversion and patience. Risk averse individuals will engage in less criminal activity due to their desire to avoid possible incarceration, while more patient individuals are more willing to invest in the time required to obtain education for higher future wages.

Secondly, Lochner (2011) identifies two additional effects of education on individuals: peer and incapacitation effects. By attending school, students interact with other students. This interaction reinforces the crime reducing effects that education has on the individual, reducing students' future participation in criminal activity. Lochner also explains that peer effects could have a positive effect on crime, as students are all released from school simultaneously, and that their propensity for "group-based delinquency" (Lochner (2011), 9) is increased. Additionally, by attending school, students' opportunities to engage in criminal activity are reduced, as students are unable to commit crime "on the streets" while in school. Also, when students leave school, they are assigned homework and required to study, further reducing free time available for criminal activity. Limiting the criminal opportunities of youth can theoretically have a major impact on crime, as a significant amount of criminals are repeat offenders, and by limiting initial involvement in crime, potential future crimes may be reduced as well.³

2.3 Possibility of Positive Effects

While it seems that, theoretically, education will have a negative effect on crime, it is also possible that more education will lead to increases in crime. The motivation, knowledge, and skills required for specific crimes are very different, and no two crimes are alike. Education will be most likely to increase crimes where it leads to increased returns from criminal behavior, either by decreasing an individual's probability of being caught or by increasing the monetary payoff of these crimes.

 $^{^{3}15.5\%}$ of the US parolee population returned to incarceration in 2007 (Bonczar and Glaze 2009).

Lochner (2011) concludes that this is most probable for white collar crimes, such as embezzlement and fraud, and there is some evidence that this may be the case (Lochner 2004).

3 Existing Literature

Education and crime is a topic increasingly analyzed, with studies building upon Lochner and Moretti's seminal publication. Lochner and Moretti (2004) exploit changes in compulsory schooling laws over time and states in the US to estimate causal effects of education on crime. Combining FBI and Census data, the authors find graduating high school has a significant negative effect on an individual's probability of incarceration and arrest for total, violent, and property crimes. They also find a significant negative effect on these types of crimes for years of education obtained.

Buonanno and Leonida (2009) analyze the effect of education on crime in Italy, using panel data from 1980-1995. Lacking a suitable instrument for education, the authors control for as many factors as possible. Using fixed effects estimation, the authors conclude graduating high school has a significant negative effect on a person's probability to commit property crimes, and an insignificant negative effect on someone's probability to commit total crimes. To account for the inertia of crime, the authors also use GMM system estimation, including a lag of crime rates. When included, both the effect of high school graduation and schooling years significantly reduce property and total crimes. Due to its significance across specifications, the authors conclude accounting for lagged crime rates is quantitatively important.

Machin et al. (2011) study the effects of education on criminal activity in the UK. Following Lochner and Moretti (2004), the authors use changes in compulsory schooling laws to identify causal effects of education on crime, using both IV and also RD estimation. The authors find a significant negative effect of years of education on property and total crimes, and an insignificant negative effect on violent crimes. They also find a significant negative effect of holding an educational qualification on property, violent, and total crimes. Their results are similar to Lochner and Moretti (2004), with the exception that they find a larger crime reducing effect of education for property crimes. Machin et al. (2012) analyze the introduction of the GCSE exam system in the UK,⁴ and also find significant negative effects of staying in school after 16 on property, violent, and total crimes.

 $^{{}^{4}\}mathrm{GCSE}$ (General Certificate of Secondary Education) is a national wide secondary education qualification in England, Northern Ireland, and Wales.

Hjalmarsson et al. (2015) analyze the impact of education on criminal activity in Sweden, making use of Swedish Register Data. Exploiting differences in the timing of changes in compulsory schooling, the authors use difference-in-differences and IV estimation to estimate causal effects of education on crime. The authors find significant negative effects of education on property and total crimes, and insignificant negative effects on violent crimes. The authors also analyze education and crime across age groups, finding greater negative effects of education on crime in younger years.

Table 1 summarizes the findings of the existing literature, and unless indicated otherwise, reports effects for males only. It is worth noting that the findings reported in Table 1 measure crime over a longer duration of an individual's lifespan than used in this paper.

Authors	Property Crimes	Violent Crimes	Total Crimes			
Additional year of education reduces crime by:						
Lochner and Moretti (2004)	11%	11-12%	16-18%			
Hjalmarsson et al. (2015)	14%	10%	7%			
Machin et al. (2011)	26-30%	$8 ext{-} 15\%$ †	19-21%			
10% Increase in High School Graduation/Those with Qualification reduces crime by:						
Lochner and Moretti (2004)	6%	8%	9-10%			
Buonanno and Leonida $(2009)^{\diamond}$	4-5%	N/A	3% †			
Machin et al. (2011)	9-10%	3-5%	7-8%			
10% Increase in proportion staying after 16 causes:						
Machin et al. (2012)	18%	13%	11-17% (women-men)			

Table 1: Existing Results

 \dagger indicates an effect is insignificant at standard significance levels. \diamond indicates effects reported are for both men and women.

While the relationship between an individual's educational attainment and their criminal propensity as an adult has not been examined in Denmark, Landersø et al. (2015) examine how the age at which children start school affects their criminal propensity, exploiting a discontinuity in the age that children typically first begin primary education. The authors find that children old for their grade are significantly less likely to be charged with a crime before the age of 18, and that this relationship appears to be driven by incapacitation effects, as those who start later also graduate later and are enrolled in school longer than those who start school at a young age.

3.1 Education and Crime Using Twin Data

Most similar in terms of methodology to this paper, Webbink et al. (2013) analyze education and crime using survey data on Australian twins. The authors investigate the reverse causality of education and crime, finding that the effects of early arrests on crime dominate the effects of education on crime; completing high school is associated with a 1.8 percentage point reduction in the probability of arrest,⁵ while an early arrest increases the probability of arrest as an adult by around 11 percentage points. When correcting for possible measurement error in self-reported levels of education, the estimated effects of education on crime are larger in magnitude, giving strength to the argument that measurement error could be an issue in the self-reported education data. Due to the strong effects of juvenile crime on education and the dominance of early arrest in explaining adult crime, the authors conclude that while education can reduce crime, limiting involvement in juvenile crime is a more dominant mechanism than education in terms of crime reduction. For the same reasons, the authors also conclude that reverse causality plays a large role in estimating the impact of education on crime. However, one shortcoming of Webbink et al. (2013) is that the concerns over measurement error in the twin survey data as well as the limited variation in the juvenile crime measure,⁶ limit the extent to which these strong results on juvenile crime are generalizable.

4 Data

In order to analyze the heterogeneous impacts of education on criminal behavior within Denmark, this paper makes use of detailed Danish Register Data provided by Statistics Denmark. A comprehensive statistical database of Danish residents from 1980-present, Danish Register Data is compiled through a variety of administrative sources to create an individual level panel dataset of every Danish resident. Each individual has a unique identification number, which is used to match individuals across various data sources. The detail and richness of the dataset make it possible to control for a variety of individual level factors which could be unobservable in other datasets. For example, by linking children to parents using their identification numbers, it is possible to include factors such as household characteristics during childhood and parental education. The dataset provides information on income, employment, personal characteristics (factors such as sex, age,

⁵This effect is 6 percentage points when considering only identical twins, and insignificant.

⁶Only 14 identical twin pairs have variation in their juvenile crime measure.

marital status, nationality, etc), education, and detailed criminal history including the exact date and crime type corresponding to specific offenses, convictions, and incarcerations.

4.1 Sample Definition and Restrictions

Twins are defined as individuals who are born on the same day and have both the same mother and father. An issue for twin analysis, which is especially pertinent for analyzing the effects of education on crime as the majority of individuals are not criminals, is having a sufficient number of twin observations. The potential usable twin sample is composed of 30,560 twins (from 15,280 pairs) all of whom are aged 18 or above in 2000, are not enrolled in education in 2000, contain educational information in 2000, and whose twin also contains educational information in 2000.

To ensure that twins are as similar as possible, two additional sample restrictions are imposed. Firstly, twins who were raised, at any point below the age of 16, in different childhood households are excluded from the estimation sample. While there are not many twin pairs raised in different households,⁷ focusing only on twins who were raised in the same household excludes twins who experience vastly different environmental factors. It is also worth noting a similar check was performed for the school twins attended, and revealed that all twin pairs attended the same schools from grade 7 and onwards.⁸

Secondly, 2,645 different gendered twin pairs, composing approximately 31% of the number of twin pairs, are excluded. Differences in gender within twins are important for two reasons. Firstly, men are much more likely to commit crimes than women, so accounting for gender differences between twins is quantitatively important. Secondly, gender different twins are non-identical (DZ),⁹ and are less genetically similar than identical twins. As data on twin type (MZ or DZ) is unavailable, distinguishing between different gendered twins is the only way to exclude twins who can confidently be classified as non-identical. Estimation throughout Sections 7 & 8 is always split by gender of the twin pair, where both twins are male or both twins are female.

While excluding different gendered twins will reduce the number of non-identical twins in the sample, non-identical twins remain in the sample. Skytthe et al. (2011) show that among twin pairs born from 1968-1982, 2,788 are MZ, 2,887 are same gender DZ, 2,921 are opposite gender DZ, and 1,616 have unknown zygosity. These measures give a very precise idea of the zygosity of the sample of twins analyzed in this paper, as this cohort of twins is 18-32 in 2000, nearly the exact

⁷There are 505 twin pairs in 2000 among the final usable sample of twins.

⁸This data is only available from grade 7 and onwards.

⁹Biologically, they cannot be MZ twins.

age ranges used in the final sample of twins described below. Taking these numbers as correct, eliminating different gendered twins from the estimation sample eliminates approximately half of the DZ twins from the entire twin sample.

In addition to this, it is also necessary to eliminate twins for whom data is unobservable at the ages of 15 and 16 in order to observe juvenile criminality. This imposes two additional restrictions: (1) limiting the sample to twins who are, at a maximum, aged 15 in 1980 when the data begins and (2) eliminating those who are not in the data at the ages of 15 or 16. Imposing these restrictions eliminates 12,314 twins who are either older than 15 in 1980, who are not present in the data as a juvenile, or whose twin is not present in the data as a juvenile from the potential estimation sample that could be used.

While not ideal, this restriction is necessary in order to control for, as in Webbink et al. (2013), whether a twin was convicted of any crime as a juvenile. It is worth noting that this restriction excludes any twin who is 36 or older in 2000, imposing an upper bound on the age of twins by construction. While this may be of concern, individuals aged 18-35 can be thought of as an age range where a large portion of offending takes place, and results which include all twins, produce similar (in terms of sign and significance), but unequivocally smaller effects of education on crime.¹⁰ This smaller magnitude of the effects of education on crime in the unrestricted sample is likely attributable to the use of an older sample, as young individuals are more likely to offend.

As juvenile criminality is a strong predictor of adult criminality, controlling for juvenile criminality is likely to be quantitatively important. Specifically, controlling for juvenile crime history can be thought to capture unobservable characteristics that vary within twin pairs and determine adult criminality.¹¹ Throughout this paper, juvenile crime is captured through the inclusion of two control variables: whether an individual was convicted of a crime while aged 15 and whether an individual was convicted of a crime while aged 16.¹² Juvenile crime is analyzed separately by age to examine if the timing of juvenile crime matters for the effects of education on crime.

Limiting the variable to crimes committed below the age of 17 increases the plausible exogeneity of this proxy variable, as younger persons are still restricted by compulsory schooling laws and are

¹⁰Results available upon request.

¹¹In addition to being a proxy for unobservables, the inclusion of a juvenile crime dummy could also be interpreted as including a lag dependent variable, which in the use of within twin fixed effects estimation, would produce biased results.

¹²The age of criminal responsibility in Denmark is 15. Due to this, any crimes committed by individuals aged 14 and younger are not observable in the data. Age is recorded as the exact age when an individual is charged by the police with a crime.

legally unable to leave school. Because of this, juvenile crime committed within the compulsory schooling window are less likely to directly affect an individual's educational attainment. While 15 year olds will still be restricted by these laws, some 16 year olds will have completed compulsory schooling during the age of 16. While using only 15 year olds would ensure students are still bound by compulsory schooling laws, including 16 year olds as well provides a more complete picture of juvenile delinquency. It should be noted that specifications excluding the juvenile crime at age 16 control and specifications including the control produce similar results. Because the extent to which a juvenile delinquency dummy can be considered truly exogenous in a regression of crime on education is perhaps uncertain, all baseline results are reported both with and without controls for juvenile crime history.

4.2 Crime and Education Definitions

Given the panel nature of Danish Register Data, a very complete picture of an individual's criminal history is observable. However, as with any administrative individual-level crime data, individuals are only classified as criminals if they are apprehended for the crime committed. Due to this, there could be measurement error in the crime data. In particular, if more skilled or clever criminals are both more educated and also better able to avoid detection, then the estimates of the effects of education on crime will be biased. As alternative measures of criminal activity, such as self-reported crime are unavailable, there is little that can be done to investigate this potential issue. However, as Danish Register Data is linked directly to police records, any individual either charged, convicted, or incarcerated in Denmark can be classified as a criminal.

Using detailed crime codes, it is possible to identify types of offenses (property, violent, etc) as well as specific offenses (assault, motor vehicle theft, etc). The obvious concern with using detailed offenses is that to begin with, each offense does not contain many individual observations, a problem which is only compounded when estimating within twins. As such, broad offense categories, such as total, property, and violent crimes, are used. Total crimes correspond to the Danish classification of offenses are comprised of: sexual, violent, property, alcohol related traffic, narcotics, firearms, tax, unknown, and other crimes, as well as crimes against special legislation.¹³ A detailed discussion of crime types, as well as the offenses that make up property and violent crimes, can be found in Appendix A.

Throughout this paper, crime is defined as a binary variable indicating if a twin has been ¹³This excludes traffic violations and citations, accidents, etc which are also recorded in police data.

convicted of a crime which was committed between 2001-2006,¹⁴ while education is measured as the highest education an individual has obtained in 2000. Using only the year 2000 enables the use of within twin fixed effects estimation, while using whether a twin was convicted in the 6 years following 2000 provides a more complete picture of offending than if a single year was used.¹⁵ A summary of Denmark during the time period of the analysis is provided in Appendix B.

A major advantage of using Danish Register Data for twin estimation is that, unlike most twin datasets, data is not obtained through surveys of twins, but through administrative sources. One problem in using twin survey data is that it is subject to measurement error, caused primarily by twin recall errors. In addition, selective response amongst surveyed twins can introduce selection bias if twins who do not respond to the survey differ systematically from twins who do respond. As Danish Register Data is linked directly to administrative sources, within twin estimation conducted in this paper is free from measurement error caused by twin recall errors and selection problems which remain in twin survey data. However, as years of education is calculated based on achieved qualifications, this could introduce measurement error in education length if a twin takes either more or less years to achieve a given qualification. Due to this, education defined in terms of the qualification achieved is also analyzed in addition to years of education.

5 Summary Statistics

Mean values and standard deviations for relevant determinants of an individual's criminal propensity included in the analysis for the twin sample are summarized in Table 2. These values are separated by whether or not an individual was convicted of a crime committed from 2001-2006 in columns (1) and (2) respectively. Two striking differences appear when comparing these two columns, the differences in years of education and whether an individual was convicted of a crime as a juvenile. Those convicted of some crime, on the whole, receive 1.5 years less of education, and are 9 and 10 percentage points more likely to have been convicted of a crime at age 15 and 16 respectively.

The gender difference in criminals, which has long been documented by economists and crimi-

 $^{^{14}}$ Specifically, whether an individual was convicted of a crime which was committed between January 1, 2001 to December 31, 2006. For a small fraction of crimes, the date of the offense is unobservable. For these crimes, the date of conviction is used instead. Estimation using the date of conviction or only those offenses with a date of offense produce similar results.

¹⁵While the results do fluctuate slightly between the chosen year, results are relatively stable across years, and are available upon request.

X 7 · 11			(3)	(4)
Variable	Not Convicted of Crime	Convicted of Crime	Total	(1)- (2)
	Committed 2001-2006	Committed 2001-2006		
Years of Education	12.46	10.97	12.38	1.49**
	(2.18)	(2.11)	(2.20)	[17.3]
Male	0.48	0.86	0.51	-0.38**
	(0.50)	(0.35)	(0.50)	[-19.2]
Age	27.82	27.06	27.77	0.75^{**}
	(4.99)	(5.38)	(5.02)	[3.8]
Juvenile Crime 15	0.01	0.10	0.02	-0.09**
	(0.11)	(0.31)	(0.13)	[-17.9]
Juvenile Crime 16	0.01	0.11	0.02	-0.10**
	(0.12)	(0.32)	(0.14)	[-18.1]
Parents Highly Educated [†]	0.43	0.35	0.42	0.08**
	(0.49)	(0.48)	(0.49)	[3.6]
High Crime Municipality at 15	0.66	0.71	0.66	-0.05*
	(0.47)	(0.46)	(0.47)	[-2.6]
Number of Twins	11276	670	11946	
Fraction of Twin Sample	94.39%	5.61%	100%	

Table 2: Summary Statistics of Sample of Twins by Whether Charged with Crime or Not During 2001-2005

Mean values for 2000 values. Standard deviations in parentheses, t statistics reported in brackets. **, *, and + correspond to significance at the 1% 5% and 10% levels respectively. \dagger : sample sizes for Both Parents Highly Educated are 9,888, 548, and 10,436 respectively due to missing parental education information.

nologists, is also visible in Table 2, with 86% of twins convicted during 2001-2006 being male. For twins whose parental education information is available, those convicted as an adult also have a lower fraction of parents who are both highly educated, a point which is investigated further in Section 8. The fraction of twins convicted of some crime in 2001-2006, 5.6% of the sample twin population, is also reported in the bottom of Table 2.

Consistent with the descriptive evidence on the links between education and crime, Figure 1 displays the fraction of twins with certain years of education separated by whether a twin was convicted of a crime committed from 2001-2006 or not.¹⁶ Twins who are convicted of any crime are, on the whole, less educated than twins who are not convicted, with more than 60% of those convicted with a crime receiving 10 years or less of education. Conversely, there are less twins who are not convicted of a crime educated only 10 years or less, with a large fraction of twin pairs receiving 12 years or more of education.

¹⁶In order to comply with Statistics Denmark's data confidentiality criteria, individuals with 17 or more years of education are not presented in Figure 1.



Figure 1: Percentage of Twins with Years of Education Separated by Criminality
Table 3 displays the distribution of crimes by type for twins convicted from 2001-2006 for both males and females combined. The majority of convictions are alcohol related traffic offenses followed by property crimes. The total number of convictions in Table 3 does not perfectly match the total number of twins convicted with a crime in Table 2 as there are some twins who are convicted of multiple types of crime during this time period.

	(1)	(2)
	Both Male and	d Female Twin Pairs
Crime Type	# Twins	%
	Convicted	
Number Twins Convicted Of Any Crime	670	
Sexual	17	1.9%
Violent	129	14.3%
Property	221	24.6%
Other	65	7.2%
Alcohol Related Traffic	225	25.0%
Narcotics	103	11.4%
Firearms	39	4.3%
Special Legislation	101	11.2%
Sum of All Types	900	100%
Number of Twins	11946	

Table 3: Distribution of Crimes by Type

Sum of all convictions is greater than the number of individuals convicted of any crime due to individuals being convicted of multiple crime types from 2001-2006. Percents in column (2) correspond to percentage of sum of all crime types (900) for a given crime. Twins are not separated by gender in order to comply with Statistics

Table 4 displays within the confidentiality iff the as of education. In 2000, 55% of twin pairs have different education lengths, with the vast majority of twins having differences of 4 years or less in educational attainment.

6 Empirical Framework

An advantage of using within twin estimation is it is possible to control for factors (both observable and unobservable) which both affect an individual's probability to engage in crime and are constant between twins, including genetic, environmental, and familial factors. By estimating fixed effects

Differences in Years	Frequency	Percent	Cumulative Percent
0	2675	44.78	44.78
1	1230	20.59	65.38
2	750	12.56	77.93
3	671	11.23	89.17
4	383	6.41	95.58
5	187	3.13	98.71
6	46	0.77	99.48
7+ years	31	0.52	100.00
Number of Twin Pairs	5973	100.00	

Table 4: Distribution of Education Differences by Twin Pair

within twin pair, variation in twin education levels enables the estimation of causal effects of education on crime. To obtain causal estimates, the following linear probability¹⁷ fixed effects regression is estimated:

$$C_{ij} = \beta(S_{ij}) + \phi_1(J_{ij}^{15}) + \phi_2(J_{ij}^{16}) + \gamma(A_{ij}) + \alpha_i + \varepsilon_{ij}$$
(1)

 C_{ij} represents whether twin j in twin pair i was convicted from 2001-2006 and is explained by: S_{ij} , the years of schooling of twin j in twin pair i; J_{ij}^{15} and J_{ij}^{16} , control variables for whether twin j in twin pair i was convicted of a crime as a juvenile at age 15 or 16; A_{ij} , unobservable factors which vary across both twin pairs and twins; α_i , unobservable factors which are identical to twins but vary across twin pairs; and ε_{ij} , the (criminal) error term.

To further explore the assumptions necessary for within twin fixed effects estimation to identify causal effects, the (criminal) error term, as in Bound and Solon (1999), can be expanded into two separate components: a component which is constant between twins and a component which is random, but only to twin j in twin pair i.

$$\varepsilon_{ij} = f_i + u_{ij} \tag{2}$$

¹⁷A linear probability model is used in order to facilitate the comparison of coefficients across specifications to see how the estimated effects change when examining the heterogeneity and robustness of the baseline results. Estimation using logit provide similar results, and are available upon request.

Additionally, an individual's schooling can be explained by:

$$S_{ij} = \delta_1(A_{ij}) + g_i + w_{ij} \tag{3}$$

Within twin fixed effects estimation is able to account for f_i , the portion of ε_{ij} which is constant within twin pair. Thus, as with any twin study, the identifying assumption is that differences in education are driven by factors which do not also impact the outcome of interest, here crime participation. In other words, it is assumed that differences in schooling levels are caused by random factors (w_{ij}) which are unrelated to an individual's propensity to engage in crime, $cov(u_{ij}, w_{ij}) = 0$, and also $cov(A_{ij}, w_{ij}) = 0$, so that schooling and crime are not correlated through A_{ij} when estimating within twins. Related to this, it is required that $A_{i1} = A_{i2}$, otherwise differences in schooling are driven by differences in unobservable factors which vary between twins. In order for equation (1) to produce unbiased estimates, these assumptions must hold. If differences in factors that affect both education and crime participation drive variation in twin education levels,¹⁸ then within twin fixed effects estimates will be biased.

An additional concern is the critique of Bound and Solon (1999) who, building upon the work of Griliches (1979), state that while MZ twins are nearly genetically identical, differences in unobservable factors¹⁹ which affect both education and, in this paper, crime decisions are determined by more than just genetic factors as twins experience different factors during their development. Twins are exposed to a variety of different environmental factors, particularly in school, and while genetically they may be similar, these environmental differences can contribute to unobservable differences which may be driving both crime and education decisions.

This criticism is related to the assumption of the equality of unobservable factors between twins, $A_{i1} = A_{i2}$. If unobservable factors between twins are different, then within twin estimation will produce biased estimates. To further investigate this criticism, A_{ij} can be broken down into two separate components:

$$A_{ij} = G_i + E_{ij} + \mu_{ij} \tag{4}$$

where G_i are unobservable factors which are the same between twins (genetic factors) and E_{ij} are unobservable factors which can vary both within and across twin pairs (environmental factors).

¹⁸For example time discounting or risk aversion.

¹⁹The classic example is ability

Twins experiencing different environmental factors is equivalent to the case where $E_{i1} \neq E_{i2}$. In this case, unobservable factors (A_{ij}) , which affect both schooling and criminality are different and within twin estimation will produce biased results. While it is not straightforward to test the validity of assuming $A_{i1} = A_{i2}$, one important factor that can be exploited, as previously mentioned, is to exclude twins who were raised in separate households as children. In this case, the environmental factors between twins are vastly different, and this identifying assumption for within twin estimation in all likelihood does not hold, as $E_{i1} \neq E_{i2}$, and $A_{i1} \neq A_{i2}$.

One additional concern is that early differences in health, potentially from the time of birth, could drive differences in twin education levels. In particular, competition for nutrients in utero could cause poor health outcomes of one twin and not another leading to differences in, for example, birth weight and these differences could potentially determine the within twin variation in education levels. Using Swedish data, Sandewall et al. (2014) empirically investigate these concerns over the validity of the identifying assumptions of twin studies, finding that the estimated effects of education on wages fall when including measures such as twin birth weight and IQ scores into standard twin earnings regressions. As such, the author's question the validity of the identifying assumptions in twin studies.

There is also a large literature, commonly using twins as an identification strategy, which analyzes the impact of infant health and birth weight on numerous adult outcomes such as employment, education, and mortality. This literature has produced mixed findings in that some papers find that differences in early health outcomes such as birth weight and Apgar infant health scores²⁰ have no effects on the probability of twin's attending high school at the age of 17 (Oreopoulos et al. 2008), estimate an effect of birth weight on education which is very small in magnitude (Royer 2009), or estimate an impact of twin birth weight on high school completion which is larger in magnitude (Black et al. 2007). While these studies tend to report some variation in the early health outcomes of twins, for example Oreopoulos et al. (2008) report differences in Apgar scores and birth weight, these differences are often not important enough to explain a large portion the differences in education between two twins. While the similarity of twins remains a concern here, Section 9.3 presents differences in GPA for a subset of twins for whom GPA information is available, as birth weight, IQ, or other early health measures are unavailable for the sample of twins used in this paper.

²⁰Apgar scores measure newborn health over five criteria, where the scores range from 0-10.

A related concern could be that one twin is born with a congenital birth defect, which could also drive differences in twin educational levels. However, previous studies have reported these defects affecting either twin are very rare,²¹ and are unlikely to be of major concern. One final possibility is that behavioral disorders affecting one twin could drive differences in education levels. In the working paper version of Webbink et al. (2013) (Webbink et al. 2008), the authors find that controlling for juvenile crime has a much larger impact on the effects of education on crime than when controlling for conduct disorders. While concerns over behavioral disorders may remain, it seems likely that controlling for juvenile crime will be more quantitatively important in estimating the impact of education on crime.

7 Results

7.1 OLS Estimation on Population Compared with Existing Results

Prior to examining the heterogeneous effects of education on crime, it is first worthwhile to establish whether the Danish population provides a good basis for comparing how the effects of education on crime depend on specific factors and individuals by comparing results for the general population to existing findings in the literature. If population results across countries are similar, it is suggestive that the heterogeneous effects found in Denmark are not due to something particular about Danish data. A natural point of comparison is Hjalmarsson et al. (2015), who use Swedish population data which is Register based and collected in a similar fashion. However, lacking a similar instrument, OLS results are the only results that can be compared.

Wherever possible, this separate population data is constructed to ensure the comparability of the Danish data with the dataset used in Hjalmarsson et al. (2015). For example, the exact same birth cohorts (1943-1954) are used, and the definitions of crime and education are identical. The main difference results from the fact that Danish Register Data only begins in 1980, while the Swedish data can be linked back to 1960. As such, factors such as municipality of residence and educational attainment are recorded at the start of 1980 and not earlier, and crime data is only available from 1980-2007 instead of from 1973-2007 as in the Swedish data.

OLS results on Denmark are presented in columns (1) and (2) of Table 5 and for Sweden in columns (3) and (4). Despite these minor differences in the years available, OLS results using

 $^{^{21}}$ For example, Black et al. (2007) report excluding only 2.1% of twin pairs from their sample where either twin has a birth defect.

	(1)	(2)	(3)	(4)
	Der	nmark	Sweden - from H	jalmarsson et al. (2015)
	Total - Male	Total - Female	Total - Male	Total - Female
Years of Education	-0.026**	-0.008**	-0.026**	-0.006**
	(0.001)	(0.001)	(0.001)	(0.000)
Municipal FEs?	Yes	Yes	Yes	Yes
Birth Cohort FEs?	Yes	Yes	Yes	Yes
	Duinen Mala	Duinen Franzia	Duinen Mala	Duinen Franch
	Prison - Male	Prison - Female	Prison - Male	Prison - Female
Years of Education	-0.010**	-0.001**	-0.011**	-0.001**
	(0.001)	(0.000)	(0.001)	(0.000)
Municipal FEs?	Yes	Yes	Yes	Yes
Birth Cohort FEs?	Yes	Yes	Yes	Yes
N	473942	453425	208626	203734

Table 5: OLS Results of Danish Data Compared to Hjalmarsson et al. (2015)

Standard errors clustered at municipal level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. OLS results from Sweden are taken from Hjalmarsson et al. (2015), Table 5. Total corresponds to a conviction for any crime from 1980-2007 in Denmark and from 1973-2007 in Hjalmarsson et al. (2015). Prison corresponds to a prison sentence from 1980-2007 in Denmark and from 1973-2007 in Hjalmarsson et al (2015).

comparable Danish data are nearly identical to the OLS results in Hjalmarsson et al. (2015).²² While Denmark and Sweden are institutionally very similar, the consistency of the OLS results of the entire population with findings of previous literature, albeit during a different time period used in the main results, suggests that Danish data provides a good basis for investigating the heterogeneous effects of education on crime. The results that follow examine the sample of Danish twins in order to identify exactly for what types of individuals we can expect education to reduce crime.

7.2 OLS Estimation on Sample of Twins

Having established the comparability of the Danish population as a whole to existing studies in the literature, Tables 6 and 7 report results using OLS estimation on the sample of male and female twins in the year 2000 respectively. Throughout the results section, columns (1) & (2), (3) & (4), and (5) & (6) show results using convictions for total crimes, property crimes, and violent crimes as a dependent variable. Even columns present results controlling for juvenile crime. Across all specifications, standard errors are clustered at the twin level.

 $^{^{22}}$ The sample in Denmark uses the full population data, where Hjalmarsson et al. (2015) use a representative sample of the Swedish population.

	(1) Total	(2) Total	(3) Property	(4) Property	(5) Violent	(6) Violent
Years of Education	-0.0255^{**} (0.0018)	-0.0221^{**} (0.0017)	-0.0089^{**} (0.0011)	-0.0071^{**} (0.0010)	-0.0067^{**} (0.0009)	-0.0055^{**} (0.0008)
Juvenile Crime 15		0.2089^{**} (0.0348)		0.1143^{**} (0.0273)		0.1269^{**} (0.0268)
Juvenile Crime 16		$\begin{array}{c} 0.2138^{**} \\ (0.0332) \end{array}$		$\begin{array}{c} 0.1151^{**} \\ (0.0258) \end{array}$		$\begin{array}{c} 0.0307^+ \ (0.0181) \end{array}$
R ² Number Individuals	$\begin{array}{c} 0.039 \\ 6038 \end{array}$	$\begin{array}{c} 0.076 \\ 6038 \end{array}$	$\begin{array}{c} 0.015\\ 6038 \end{array}$	$\begin{array}{c} 0.050 \\ 6038 \end{array}$	$\begin{array}{c} 0.012\\ 6038 \end{array}$	$\begin{array}{c} 0.041 \\ 6038 \end{array}$

Table 6: OLS Estimates of Sample of Male Twins In Same Childhood Environments

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

For both male and female twins, years of education significantly and negatively affects the probability of conviction across crime types. This effect is most dominant for total crimes, where an additional year of education, holding all other things equal, reduces the probability of conviction for an offense between 2001-2006 by 2.6 percentage points for males and 0.5 percentage points for females. When including juvenile crime controls, the effect of education on the probability of conviction is drastically reduced for males, while only marginally reduced for females. Having committed crime as a juvenile significantly increases the probability of conviction across crime types for males. The results in Tables 6 and 7 fail to deal with the endogenous relationship between education and crime and within twin fixed effects estimation, which identifies causal effects of education on crime, is reported below in Section 7.3.

	(1) Total	(2) Total	(3) Property	(4) Property	(5) Violent	(6) Violent
Years of Education	-0.0052^{**} (0.0009)	-0.0051^{**} (0.0009)	-0.0033^{**} (0.0007)	-0.0032^{**} (0.0007)	-0.0010^{**} (0.0004)	-0.0010^{**} (0.0004)
Juvenile Crime 15		$0.0762 \\ (0.0545)$		$0.0185 \\ (0.0328)$		-0.0037^{**} (0.0012)
Juvenile Crime 16		0.0021 (0.0314)		$0.0158 \\ (0.0310)$		-0.0038^{**} (0.0012)
R ² Number Individuals	$0.008 \\ 5908$	$0.010 \\ 5908$	$0.006 \\ 5908$	$0.006 \\ 5908$	$0.002 \\ 5908$	$0.002 \\ 5908$

Table 7: OLS Estimates of sample of Female Twins In Same Childhood Environments

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

7.3 Within Twin Estimation on Sample of Twins

Tables 8 and 9 analyze separately same gendered male and female twin pairs who were also raised in the same household. As criminals are overwhelmingly male, it is perhaps unsurprising that a greater negative effect of education on the probability of conviction is seen for males compared to females. For male twins who were raised in the same childhood household, a ceteris paribus additional year of education reduces an individual's probability of conviction for an offense committed between 2001-2006 for any crime by 1.3 percentage point (14% of the sample mean). For property crimes, marginally significant effects of education are found, where an additional year of education lowers the probability of conviction by 0.3 percentage points (12%) while for violent crimes, the estimated effect of an additional year of education is 0.4 percentage points (19%). When controlling for participation in juvenile crime, these numbers fall to 14%, 12%, and 18% respectively. On the whole, the estimated effects of education on crime are large for males. For female twins, the estimated effect of years of education is imprecisely estimated, with virtually no effects either positive or negative. Estimating the effects of education on the probability of conviction for female twins leads to the conclusion that education can predominantly reduce crime for male twin pairs.

	(1) Total	(2) Total	(3) Property	(4) Property	(5) Violent	(6) Violent
Years of Education	-0.0132^{**} (0.0031)	-0.0129^{**} (0.0031)	-0.0034^+ (0.0019)	-0.0033^+ (0.0019)	-0.0036^{*} (0.0016)	-0.0035^{*} (0.0016)
Juvenile Crime 15	()	0.0999^+ (0.0517)	()	0.0421 (0.0361)	· · ·	0.0965^{**} (0.0331)
Juvenile Crime 16		0.1071^{*} (0.0447)		0.0431 (0.0353)		0.0041 (0.0227)
\mathbb{R}^2	0.039	0.076	0.015	0.050	0.012	0.040
Number Individuals Twin Pairs	$6038 \\ 3019$	$6038 \\ 3019$		$6038 \\ 3019$	$\begin{array}{c} 6038\\ 3019 \end{array}$	$\begin{array}{c} 6038\\ 3019 \end{array}$

Table 8: Within Twin Fixed Effect Estimates of Sample of Same Gender Male Twins In SameChildhood Environments

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

Comparing even and odd columns reveals the importance of accounting for juvenile crime history. Again, participation in juvenile crime increases the probability of conviction as an adult for males, although the estimated effects are lower in magnitude and less precisely estimated

(1) Total	(2) Total	(3) Property	(4) Property	(5) Violent	(6) Violent
-0.0008 (0.0014)	-0.0009 (0.0014)	-0.0001 (0.0010)	-0.0001 (0.0010)	-0.0005 (0.0006)	-0.0005 (0.0006)
	0.1281^{+} (0.0674)	, , ,	0.0432 (0.0409)	`	0.0015 (0.0020)
	-0.0766 (0.0488)		-0.0373 (0.0352)		-0.0361 (0.0352)
$0.008 \\ 5908 \\ 2954$	$0.003 \\ 5908 \\ 2954$	$0.006 \\ 5908 \\ 2954$	$0.000 \\ 5908 \\ 2954$	$0.002 \\ 5908 \\ 2954$	$0.000 \\ 5908 \\ 2954$
	(1) Total -0.0008 (0.0014) 0.008 5908 2954	$\begin{array}{ccc} (1) & (2) \\ Total & Total \\ -0.0008 & -0.0009 \\ (0.0014) & (0.0014) \\ 0.1281^+ \\ (0.0674) \\ -0.0766 \\ (0.0488) \\ \hline 0.008 & 0.003 \\ 5908 & 5908 \\ 2954 & 2954 \\ \end{array}$	$\begin{array}{ccccc} (1) & (2) & (3) \\ Total & Total & Property \\ \hline \begin{tabular}{lllllllllllllllllllllllllllllllllll$	$\begin{array}{c ccccc} (1) & (2) & (3) & (4) \\ Total & Total & Property & Property \\ \hline 0.0008 & -0.0009 & -0.0001 & -0.0001 \\ (0.0014) & (0.0014) & (0.0010) & (0.0010) \\ 0.1281^+ & & 0.0432 \\ (0.0674) & & (0.0409) \\ -0.0766 & & -0.0373 \\ (0.0488) & & (0.0352) \\ \hline 0.008 & 0.003 & 0.006 & 0.000 \\ 5908 & 5908 & 5908 \\ 2954 & 2954 & 2954 & 2954 \\ \end{array}$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$

Table 9: Within Twin Fixed Effect Estimates of Sample of Same Gender Female Twins In Same Childhood Environments

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

than in Table 6. Having been convicted of a crime committed at 15 increases the probability of conviction by 10 percentage points for total crimes, 4.2 percentage points for property crime, and 9.7 percentage points for violent crime, where the effect on property crimes is insignificant. Having been convicted at the age of 16 positively impacts the probability of conviction as an adult, an effect which is significant for total crimes.

While the results of this paper indicate the underlying importance of involvement in juvenile crime on adult crime participation, an additional year of education leads to a 1.3 percentage point reduction in the probability of conviction for males. While this number is indeed smaller than the 10 percentage point effect estimated for juvenile crime at age 15 or 16, there is considerably less gap between the two figures than in Webbink et al. (2013), where the authors find a large effect of juvenile crime on adult crime participation but a much smaller (and sometimes statistically insignificant) effect of completion of high school on adult crime. As such, this paper finds evidence both that accounting for juvenile crime history when estimating the impact of education on adult crime is important and once controlling for this, that education significantly reduces adult participation in total, property, and violent crimes for males by a sizable factor. However, it should be noted that the results presented above are potentially biased as it is only possible to exclude fraternal twins, who only share 50% of their genes, of different genders and not fraternal twins of the same gender. Lacking zygosity information, Section 9 investigates the robustness of these results taking into account the potential bias of DZ twins in the estimation sample, while the next section highlights the heterogeneous effects of education on crime.

8 Heterogeneous Effects

Having confirmed a link between education and crime in the baseline results, the next section outlines how the crime reducing effects of education differ across four aspects: parental education, growing up in a high or low crime neighborhood, specific educational programs, and detailed crime types. In the interest of brevity and given that estimation including juvenile crime controls produces very similar results to estimation not controlling for juvenile crime, only estimation controlling for juvenile crime is reported in the following sections.

8.1 Parental Education

A variety of studies using various identification methods have established the existence of intergenerational transmission of education (see Holmlund et al. (2011) for a summary) as well as the intergenerational transmission of crime (Hjalmarsson and Lindquist 2012, 2013), so it is natural to investigate how the crime reducing effects of education vary across children with differently educated parents. Similar to this, Meghir et al. (2012) investigate the intergenerational links of parental education and criminal participation of children, finding a national education reform in Sweden not only decreased parental criminality through increased education, but also reduced son's criminality as well. Investigating the exact causes of the intergenerational relationship, the authors find that increases in parental resources and increases in the quality of parenting through better role modeling are the most plausible mechanisms driving this relationship. While the analysis conducted here is similar, the former investigates the impact of an exogenous change in parental education on criminality of children while the analysis below investigates how the effects of education on crime for children vary across a given level of parental education. Given the importance of parental background in both the education and criminality of children, larger effects of education on crime are expected for children of low educated families, with smaller effects expected for high educated families.

Throughout this subsection, parental education is defined as high if a parent has been educated to the high school level or beyond and low if a parent is educated to a level less than high school. Table 10 presents results restricting the estimation sample to two different subsamples of twin pairs: those with one or both parents with low education and those with both parents with high

	(1)	(2)	(3)	(4)	(5)	(6)
		Male Twins	ł	F	Female Twir	ns
At Least One Low	Educated	l Parent				
	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0199**	-0.0041^{+}	-0.0041*	-0.0022	-0.0020	-0.0014
	(0.0040)	(0.0024)	(0.0020)	(0.0022)	(0.0018)	(0.0010)
Number Individuals	3080	3080	3080	2924	2924	2924
Twin Pairs	1540	1540	1540	1462	1462	1462
Both Parents High	1 Educate	d				
	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0043	-0.0042	-0.0046^{+}	-0.0008	0.0012	0.0005
	(0.0051)	(0.0034)	(0.0024)	(0.0025)	(0.0014)	(0.0011)
Number Individuals	2200	2200	2200	2232	2232	2232
Twin Pairs	1100	1100	1100	1116	1116	1116

Table 10: Within Twin Fixed Effect Estimates of Sample of Same Gender Twins In Same Childhood Environments By Educational Attainment of Parents

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. Low educated parent is either a low educated father, low educated mother, or both low educated mother and father. Not reported are controls for juvenile crime.

education.²³ As before males and females are analyzed, as while little effect of education on crime was seen previously for women, it is possible that estimation on the overall sample of women masks substantial heterogeneous effects by parental education. Columns (1)-(3) present results for male twins, while columns (4)-(6) for female twins. It should be noted that the parental educational information of 755 same gender, same household twin pairs is missing, and these twin pairs are excluded from this analysis.²⁴

The top panel of Table 10 suggests that for male twins who have at least one low educated parent, the effects of education on crime are larger in magnitude than comparable estimated effects using an unrestricted sample. In contrast to male twins with at least one low educated parent, when twins' parents are both highly educated, there are virtually zero effects of education on crime, with the exception of violent crimes. For males of highly educated families, the estimated effects

 $^{^{23}35\%}$ of twins with at least one low educated parent also have low education, while 23% of twins with both parents highly educated have low education.

 $^{^{24}}$ Baseline estimation on the subsample of twins who contain parental information produces similar, but larger estimated effects of education on crime. This is attributable to a younger sample, as older twins, who are less crime prone than their younger counterparts, are more likely to have parents whose educational information is unobservable, making them more likely to be excluded from the parental education sample. Assuming that missing parental education corresponds to low parental education produces results similar to those in the top panel of Table 10.

of years of education on total and property crimes are both small in magnitude and insignificant, where these differences between low and highly educated families are significant for total crimes. For females, there appear to be no significant effects of education on adult crime for either low or highly educated families, although the estimated effect of education on adult crime is negative for low educated families, but far from significant.

Appendix C analyzes the role of parental education in further detail, presenting results of the three parental education subgroups which comprise at least one parent with low education: those with both parents low educated, those with a low educated mother but high educated father, and those with a high educated mother but low educated father. However, due to the small number of twins in each of these groups, the results should be interpreted with some degree of caution. Taking this into account, Table A1 indicates some non-linearity in the effects of education on crime by parental education: male twins with two low educated parents experience the largest effects of education on total crime while those with one low educated parent, either mother or father, and one high educated parent experience less crime reducing effects of education. For female twins, as before, there appear to be no significant effects of education on crime.

The results presented in Tables 10 and A1 lead to the conclusion that parental education matters for the crime reducing effects of education of their children, where for males with at least one low educated parent, there are large crime reducing capabilities of education, but for males of highly educated parents, there are little crime reducing benefits of education. The fact that no effects are seen for females of low educated families reinforces the previous finding that education can reduce crime predominantly only for males. Three separate, but interrelated, explanations of these findings seem plausible. Firstly, Table 10 could simply be capturing the intergenerational transmission of education. If individuals with highly educated parents obtain more education and there is less crime reduction caused by higher education than lower education, then the lack of any negative effect seen for highly educated parents is due to the higher educational attainment of their children. Secondly, education could act as an adjusting mechanism, in that an additional year of education in terms of crime reduction, is very important for children of a low educated family and not very important for children of a high educated family. If, for whatever reason, children of low educated parents are more disposed to commit crime, then an additional year of education would have greater crime reducing benefits for these children. A third explanation could be that parental education captures "good parenting", and good parenting simultaneously leads to both more education and less crime. Lacking a way to disentangle these potential explanations, identifying the exact cause of these differential effects of education on crime across parental education is beyond the scope of the paper. However, given the supportive evidence of the last two mechanisms found in Meghir et al. (2012), it seems likely that parenting or social mobility are fundamental reasons for the heterogeneous effects across parental education.

8.2 Growing Up In High or Low Crime Neighborhood

In addition to family factors, environmental factors could also have differential impacts on the effects of education on crime. Many studies examine the Moving to Opportunity (MTO) experiments, which randomly allocate moving vouchers to low income families, to analyze the effects of moving to a better neighborhood on both education and criminal outcomes. Sanbonmatsu et al. (2006) find that children of families who were offered moving vouchers performed no better in school, while Chetty et al. (2015) find that, in the longer run, children moving to a neighborhood with less poverty at young ages are more likely to attend college and have higher wages. Kling et al. (2005) look at the effects of moving to a lower poverty/crime area on crime, finding that both males and females are less likely to engage in crime in the short run, while for males, participation in property crime actually increases in the long run. Damm and Dustmann (2014) use an exogenous placement scheme of refugees to examine in the impact of living in a high crime neighborhood, finding the share of young individuals convicted in neighborhoods increases the likelihood of conviction for male refugees later in life.

Growing up in a high crime area could also have a strong impact on the effects of education on crime. For instance, those residing in a high crime area could be more likely to engage in juvenile crime and/or less likely to pursue education. At the same time, those residing in higher crime areas could, all else equal, have larger returns to education than those residing in lower crime areas. As the sample of twins constructed all reside in the same household during childhood, and hence same municipality, both twins are exposed similarly to these environmental factors.

Table 11 examines the impact of growing up in a high or low crime area on the effects of education on adult crime for a predetermined neighborhood.²⁵ High crime neighborhoods are defined as a municipality where the youth conviction rate is higher than the youth conviction rate of the median municipality in that year. Youth conviction rate is defined, as in Damm and

²⁵Across all twins, municipality of residence is measured in the year twins are age 15. Results are robust to alternative definitions of municipality. There are 275 municipalities in the time period analyzed in Denmark.

Dustmann (2014), as the fraction of individuals aged 15-25 convicted of a crime in a given year. As before, the estimation controls for juvenile crime, which is especially relevant in this subsection given the amplifying effects that growing up in a high crime neighborhood could have on an individual's propensity to engage in juvenile crime.

Table 11: Within Twin Fixed Effect Estimates of Sample of Same Gender Twins In Same Childhood Environments By Municipal Youth Conviction Rates

	(1)	(2)	(3)	(4)	(5)	(6)
		Male Twins	5	F	emale Twir	IS
Raised in a Munic	ipality Wi	th Low Yo	outh Conv	iction Rate		
	Total	Property	Violent	Total	Property	Violent
Years of Education	$\begin{array}{c} -0.0144^{**} \\ (0.0049) \end{array}$	-0.0046 (0.0030)	$\begin{array}{c} -0.0082^{**} \\ (0.0029) \end{array}$	-0.0025 (0.0029)	-0.0006 (0.0021)	-0.0013 (0.0018)
Number Individuals Twin Pairs	$2090 \\ 1045$	$2090 \\ 1045$	$2090 \\ 1045$	$\frac{1966}{983}$	$\frac{1966}{983}$	$\frac{1966}{983}$

Raised in a Municipality With High Youth Conviction Rate

	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0122^{**} (0.0039)	-0.0026 (0.0024)	-0.0010 (0.0020)	-0.0002 (0.0016)	0.0001 (0.0011)	-0.0002 (0.0005)
Number Individuals Twin Pairs	$\begin{array}{c} 3948 \\ 1974 \end{array}$	$\begin{array}{c} 3948 \\ 1974 \end{array}$	$\begin{array}{c} 3948 \\ 1974 \end{array}$	$3942 \\ 1971$	$\begin{array}{c} 3942 \\ 1971 \end{array}$	$\begin{array}{c} 3942 \\ 1971 \end{array}$

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1% 5% and 10% levels respectively. High juvenile conviction: residing in a municipality at age 15 where the youth conviction rate is higher than the rate of the median municipality in that year. Low juvenile conviction: residing in a municipality at age 15 where the young conviction rate is equal to or lower than the rate of the median municipality in that year.

For males, the estimated effects of education on total crimes presented in Table 11 are relatively

similar irrespective of growing up in a high or low crime municipality. While the effects, in percentage point terms, are greater for those raised in a low crime municipality, they are very similar to those raised in a high crime municipality. Education appears to only reduce violent crime for individual's raised in low crime areas, with negative, but insignificant effects seen for those in a high crime area. For property crimes, negative but insignificant effects of education are seen for both groups. For females, the effects of education on crime are greater for those raised in a low crime municipality, but are never significant at conventional levels for either low or high crime neighborhoods. With the exception of violent crimes, environmental factors, measured as being raised in a high or low crime neighborhood, appear to lead to minimal differences in terms of the crime reducing capabilities of education.

8.3 Differential Margins of Education

As discussed previously, one advantage of using within twin estimation to identify causal effects of education on crime is that, unlike a change in compulsory schooling laws, multiple margins of education can be investigated. Despite the many studies on the effects of education, little emphasis has been placed on how these effects differ across an individual's specific program of education. One exception is Åslund et al. (2015), who find that an expansion of vocational upper secondary education from two to three years lowered property crimes in Sweden. Their findings support the role of incapacitation effects, with the crime reduction largely attributable to the introduction of the additional third year in vocational education. This supports the idea that both vocational and non-vocational education can be expected to reduce criminal propensity.

In this section, qualifications are divided by vocational training and non-vocational education to correspond to the Danish education system. In Denmark during the time period examined, once students complete compulsory schooling of 9 years, they then can either progress directly to high school or attend an optional 10th grade and then proceed to high school. There are three different types of high school students can attend in Denmark: regular, business (Højere Handelseksamen), and technical (Højere Teknisk Eksamen) high school. In addition to this, students can attend vocational training directly after compulsory schooling, which is similar to apprenticeship programs and geared not towards attending higher education, but rather professional employment. For simplicity, education of 9 or 10 years is referred to as lower secondary education, such that individuals can either attend high school or vocational training following lower secondary education.

Table 12 reports the effect of completion of three different types of education on crime: (i) completing any high school from lower secondary education, (ii) completing any university education from any high school, (iii) completing vocational training from lower secondary education. Results reported in Table 12 are constructed in such a way that the sample is restricted to twin pairs where both twins are in one of the two educational groups analyzed. For example, the results in the top row compare the criminal participation of one twin who has only completed lower secondary education to a twin who has completed any high school. While in principle the three types of high school could also be separately analyzed, it is necessary to combine all types of high school into one qualification in order to ensure there are an adequate number of twins in each educational group. It is worth noting that for some qualifications, the effects are being driven by a small number of twins who make up a small percentage of the population.

	(1)	(2)	(3)	(4)	(5)	(6)
		Male Twins	3	F	emale Twi	ns
Regular (Non-Vocationa	l) Educati	on				
\mathbf{C}	ompleting H	S from Lowe	er Secondary	Education		
	Total	Property	Violent	Total	Property	Violent
Completing HS	-0.0993**	-0.0473^{*}	-0.0253^{+}	0.0273^{+}	0.0266^{+}	0.0072
	(0.0331)	(0.0225)	(0.0142)	(0.0144)	(0.0144)	(0.0072)
Number Individuals	1748	1748	1748	1820	1820	1820
Twin Pairs	874	874	874	910	910	910
	Comple	ting Any Ur	niversity from	h HS		
	Total	Property	Violent	Total	Property	Violent
Completing Any Univ.	-0.0107	-0.0089	-0.0178	0.0056	0.0056	0.0056
	(0.0181)	(0.0089)	(0.0125)	(0.0097)	(0.0056)	(0.0056)
Number Individuals	1144	1144	1144	1608	1608	1608
Twin Pairs	572	572	572	804	804	804
Vocational Training						
Completin	g Vocationa	l Training fr	om Lower Se	condary Educ	ation	
	Total	Property	Violent	Total	Property	Violent
Completing Voc. Training	-0.0263	0.0002	-0.0117	-0.0104	-0.0036	-0.0095^+
	(0.0190)	(0.0104)	(0.0106)	(0.0084)	(0.0032)	(0.0055)
Number Individuals	3462	3462	3462	2940	2940	2940

Table 12: Differential Margins of Education by Type of Education Program

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. Not reported are controls for juvenile crime.

Twin Pairs

Previous studies on education and crime have found that while the completion of high school (or an equivalent level) produces large crime reducing effects for males, attending university has little to no impact on crime participation (Lochner and Moretti 2004). Consistent with these expectations, for male twins, results on regular (non-vocational) eduction in Table 12 reveal the completion of high school leads to a 9.9 percentage point reduction in the probability of conviction for any crime, while the completion of any university has no significant effects either positive or negative. For female twins, completion of high school actually leads to a 2.7 percentage point increase in total and property crimes, an effect which is significant at the 10% level.

Conversely, results using vocational training, reported in the bottom panel of Table 12, display a different trend. For male twins, additional vocational training past the compulsory schooling level appears to have no crime reducing benefits. For female twins, vocational education lowers the probability of conviction for total and violent crimes, where the effect on total crimes is insignificant. The lack of any crime reducing effect of vocational training for males is striking, especially when compared to the strong crime reducing effects of completing high school seen for males. As the lengths of normal high school and vocational training programs are very similar, the differences seen in Table 12 are not due to incapacitation effects.²⁶

These differential effects by type of education could be driven by underlying differences between the students of both types of education, underlying differences between the two types of education, or by underlying differences in employment prospects between the two education groups. If, for example, students who attend vocational training are more prone to commit crime than students who attend regular education, then it could take longer for education to negatively impact an individual's propensity to engage in criminal behavior. If this were the case, this could also be reinforced through peer effects while in school. An alternative explanation could be that the teaching and educational methods differ across the two types of education. If this were the case, student's criminal propensity could develop differently across the two educational programs. Another possible explanation is that the completion of high school increases an individual's probability of employment, but not for vocational training. If this were the case, then the crime reducing effects of employment caused by greater education would affect the two groups of students differently. However, as many vocational training programs are directly geared towards professional

 $^{^{26}}$ If anything, some vocational training programs can take longer than normal high school. In this case, the incapacitation effects of being involved in education for longer would lead to greater crime reducing effects of vocational training compared to normal high school, and the effects shown in Table 12 would actually underestimate the effects of completing high school compared to vocational training.

employment, this seems a less plausible explanation.

Whatever is driving the underlying differences in reduced criminality between these two types of education, the results displayed in Table 12 demonstrate the importance to crime reduction not only of an individual's educational qualifications, but also of an individual's educational program.

8.4 Detailed Crime Types

Estimation using all available crime categories that comprise total crime for male same household twin pairs is reported in Table A2 in Appendix D. Results are directly comparable to even columns of Table 8, which include juvenile crime controls. For more uncommon crimes, the effects are being driven by a small amount of male twins who make up a small percentage of the population. Taking this into account, estimated effects of education are generally negative across crime types, but with fluctuating significance.

The results of Table A2 reveal large crime reducing effects of education on traffic related alcohol crimes, where a ceteris paribus year increase in education lowers the probability of conviction by 0.7 percentage points (20%). Large negative effects are also seen for firearms and sexual crimes, where a ceteris paribus year increase in education leads to a 0.2 percentage point (32%) and 0.1 percentage point (44%) reduction in the probability of conviction respectively, where the effect for sexual crimes is significant at the 10% level. Education also negatively affects participation in other crimes for male twins, an effect which is significant at the 10% level.²⁷ Table A2 reveals that, for males, education can not only reduce participation in violent and total crimes, but in most other crime types as well.

9 Robustness Checks

Having detailed heterogeneous effects by parental education and across educational program, the results below not only investigate the robustness of the main specification but also investigate the validity of the identifying assumptions required for within twin fixed effects.

9.1 Using Incarceration as a Dependent Variable

Table 13 uses, as a dependent variable, whether an individual has spent time in jail resulting from an offense committed from 2001-2006. Results in the columns are divided by gender and not crime

²⁷The three most common crimes in the definition of other crimes are false accusations, the sale of drugs (those that are not covered by the law on narcotics), and crimes against public authority.

type, where column (1) shows results for only male twin pairs and column (2) shows results for only female twin pairs. While being convicted of a crime corresponds to criminal activity, individuals who are ultimately incarcerated usually commit more severe crimes, and as incarceration is costly to society, it is worthwhile examining both convictions and incarcerations. As such, incarceration data is used to not only determine whether the negative effects seen using conviction data are robust to an alternative measure of criminal activity but also whether education can effectively reduce incarcerations. If the effects of education on crime are robust to changes in the crime measure used, this supports the argument that education truly reduces an individual's propensity to engage in crime.

	(1)	(2)
	Male Twins	Female Twins
	Incarcerated	Incarcerated
Years of Education	-0.0035*	-0.0001
	(0.0014)	(0.0006)
Juvenile Crime 15	0.0517^{+}	0.0000
	(0.0272)	(0.0001)
Juvenile Crime 16	0.0203	-0.0001
	(0.0195)	(0.0004)
Number Individuals	6038	5908
Twin Pairs	3019	2954

Table 13: Within Twin Fixed Effect Estimates of Sample of Same Gender Twins In Same Childhood Environments using Incarceration

Standard errors reported in parentheses. ** , * , and + correspond to significance at the 1%, 5%, and 10% levels respectively.

As before, years of education significantly lowers the probability of being incarcerated for male twin pairs, but not for female twin pairs. For male twin pairs, a ceteris paribus year of education lowers the probability of incarceration by 0.4 percentage points (22%). For female twin pairs, education appears to have no significant negative effects on the probability of incarceration.

As with estimation using convictions, having been convicted of a crime committed as a juvenile increases the probability of incarceration, by roughly 2-5 percentage points depending on the age an individual engaged in juvenile crime, but there still remain significant negative effects of education on incarceration. The results in Table 13 lead to the conclusion that education has a larger, in terms of percentage of mean reduction, impact on an individual's probability of incarceration than on their probability of conviction, a finding which is also seen in other papers using Scandinavian data (Hjalmarsson et al. 2015). This is indicative that education not only decreases an individual's probability of committing crime, but also has additional effects on reducing the severity of crimes committed, as these crimes usually correspond to jail sentences.

9.2 Accounting for role of DZ Twins in Estimation

Following Holmlund et al. (2008), under certain assumptions and exploiting knowledge of the fraction of same gender DZ twins out of all same gender twins in the population (θ), the potential bias introduced by the presence of DZ twins in the estimation sample can be corrected for, even if zygosity of the individual twins is unknown. Making use of non-twin siblings (with both the same mother and father) to approximate the effect for DZ twins and under certain assumptions, the same gendered twin estimate, a composite of both MZ and DZ twins, can be decomposed into an average of the effect for MZ twins and the effect of DZ twins. This requires assuming that the education and crime equation, equation (1), is the same for same gender DZ twins and non-twin siblings, that DZ twins and non-twin siblings are treated the same, and there is no measurement error in schooling.²⁸

In order to make these similarity assumptions more believable and, again, following Holmlund et al. (2008), only same gendered siblings who are born close to each other, defined here as within 2.5 years, are used.²⁹ Some suggestive evidence of the extent to which these sibling populations are similar is presented in Björklund and Jäntti (2012), who examine sibling correlations for years of education, earnings, IQ, and non-cognitive skills. The authors explore a wide variety of siblings, including twin siblings, and show that DZ twins and closely spaced siblings (which they define as those with less than 4 years of age difference) have similar sibling correlations in education. While the DZ correlations are more imprecise, they also show that the sibling correlations for closely spaced siblings are more similar to DZ twins compared to non-closely spaced siblings (those with 4 or more years of age differences). As siblings with age differences of exactly 2.5 years are used in this paper, the similarity of same gender DZ twins and closely spaced siblings is, presumably, even more believable.

 $^{^{28}}$ As the quality of Register data is regarded as high, particularly with education, it seems realistic to assume measurement error in schooling is unproblematic.

²⁹Closely spaced siblings are those born within exactly 912 days of each other. Results are robust to altering this somewhat arbitrary cutoff.

Under the above assumptions, the MZ twin estimate is calculated as:

$$\hat{\beta}_{MZ} = \frac{1}{\theta} \hat{\beta}_{MZDZ} - \hat{\beta}_{CS}$$

where $\hat{\beta}_{MZDZ}$ is the estimate for both same gendered MZ and DZ twins which has been estimated previously and $\hat{\beta}_{CS}$ is the estimate for close siblings which represents the estimate for same gendered DZ twins had zygosity been observable. θ is taken as 0.509 from Skytthe et al. (2011), who report among twins born from 1968-1982, 2,788 are MZ and 2,887 are same gender DZ. Standard errors are estimated as in Conley et al. (2006); Conley and Strully (2012), and require the additional assumption that the covariance between the same gender twin estimates and the close sibling estimates is zero, $cov(\beta_{MZDZ}, \beta_{CS}) = 0$. This is a fairly strong assumption, which is equivalent to assuming that the error terms for same gendered twins and closely spaced siblings are unrelated.³⁰ While this assumption is necessary in order to calculate any standard errors, the standard errors resulting from this method should be interpreted with some degree of caution.

Results presented in Table 14 show that if it were possible to identify zygosity, the estimated effects of education on crime would be of a similar magnitude for MZ twins alone than in the same gender twin sample. For males, the effects of education on total convictions for all same gendered, same household males and for those with low parental education remain negative, sizable, and statistically significant. For estimation using incarceration as a dependent variable, while the estimated effects of education on crime become insignificant, the imputed MZ coefficient shows an effect which is similar in terms of magnitude, only less precisely estimated. Table 14 suggests that if zygosity were observable, a MZ twin sample would produce estimated effects of education on crime in a similar range as the effects from the same gendered twin sample used throughout the paper.

9.3 Examining Differences in GPA

As discussed in Section 6, underlying differences in the ability levels of twins will violate the identifying assumptions required for within twin estimation. Sandewall et al. (2014) investigate the possibility of this using IQ scores and birth weight. While both IQ scores and birth weight are unavailable for the sample of twins used in this paper, for a subsample of twins, average GPA

 $^{^{30}}$ To the extent there could be positive covariance between same gendered twins and closely spaced siblings estimates, if these are biased in the same direction, this would lead to standard errors which are too large. For a further discussion of this method and its shortcomings, see Conley et al. (2006).

	(1)	(2)	(3)
	Twins - $\hat{\beta}_{MZDZ}$	Close Siblings - $\hat{\beta}_{CS}$	Imputed MZ Twins - $\hat{\beta}_{MZ}$
Same Gendered, S	ame Household	Male Twins	
	Total	Total	Total
Years of Education	-0.0129**	-0.0137**	-0.0116+
	(0.0031)	(0.0017)	(0.0064)
	Incarcerated	Incarcerated	Incarcerated
Years of Education	-0.0035*	-0.0035**	-0.0035
	(0.0014)	(0.0008)	(0.0029)
Number Individuals	6124	17794	
Twin/Sibling Pairs	3062	8897	

Table 14: Within Twin Fixed Effect Estimates of Sample of Same Gender Male Twins In Same Childhood Environments Correcting for Dizygotic Twins

At Least One Low Educated Parent - Male Twins

	Total	Total	Total
Years of Education	-0.0199** (0.0040)	-0.0171^{**} (0.0024)	-0.0220* (0.0085)
Number Individuals Twin/Sibling Pairs	$3118 \\ 1559$	$8970 \\ 4485$	

Standard errors clustered at twin/sibling level reported in parentheses. Standard errors in column (3) are calculated as in Conley et al. (2006), as the square root of the variance of the $\hat{\beta}_{MZ}$ where $Var(\hat{\beta}_{MZ}) = Var(\hat{\beta}_{MZDZ})(\frac{1}{(1-\theta)})^2 + Var(\hat{\beta}_{CS})(\frac{\theta}{(1-\theta)})^2$. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. Column (1) presents results from same gendered twins (MZ & DZ). Column (2) presents results from closely spaced siblings.

during high school is observable. At the time the twins examined in this study attended high school, grades in Denmark were on a 13 point scale, where an individual could receive either a 13, 11, 10, 9, 8, 7, 6, 5, 03, or 00, where grades 5 or below were considered failing. At the time, 11 and above corresponded to an excellent grade, 10 corresponded to very good, 9 and 8 corresponded to good, while 7 was satisfactory and 6 was passing.³¹

Figure 2 shows the differences in average high school GPA, for all same gendered, same household twin pairs where data is available. It is expected that the closer twins' academic achievement is, the more similar they are, and the more likely the identifying assumptions required for within twin estimation will hold. More than 50% of the twins have average GPA differences of 0.4 or less, and on the whole, the majority of twins have very similar GPA. Consistent with this, 86% of all twins have GPA differences of 1 point or less. While 1 point may seem like a large difference, on the 13 point scale, a change of one point is often not enough to change the quality associated with the GPA.



Figure 2: Distribution of High School GPA Differences

³¹Denmark now uses a 7 point scale, which is a similar concept to the 13 point scale.

Ideally, GPA would be included as a control variable throughout the estimation, as this would provide more direct insight into how the estimated effects of education on crime differ when accounting for some measure of ability differences between twins. However, since only high school GPA is available for a subsample of twins, there is insufficient variation in either education defined as a qualification or the length of education for twin pairs where GPA is observable. While the lack of major differences in academic achievement between twins who both completed high school shown in Figure 2 is supportive of the underlying similarity of twins, this is as much as the role of underlying differences between twins can be analyzed in this paper.

9.4 Excluding Twins with Large Differences in Education

Previous twin studies (Amin 2011; Lundborg 2013) have emphasized the importance of outliers, where outlying twin observations have been shown to affect results, sometimes significantly. While a binary dependent variable may limit the potential influence of outlying twins as there can only be two outcomes, the results presented could potentially be driven by twins with large differences in education. This could be true especially given the rarity of criminal convictions, where despite a large sample of twins, a few individuals could have a large impact on the estimation. Additionally, twins who have large educational differences may also be less similar than twins with closer education. Excluding twins with large differences in education serves to not only limit the potential influence of outlying twins but also imposes a sample of twins who are, arguably, most similar.

Table 15 excludes twin pairs who have differences in education of 4 years or more. The top panel presents results on the baseline sample of twins, while the bottom panel presents results for those with at least one low educated parent. Results for total and violent crime remain largely unaffected for the baseline male sample, while the point estimate on property crime remains unchanged but loses the marginal significance seen in Table 8. Results for males from low educated families are very similar to results presented in Table 10. For females, as before, no significant effects of education on crime are seen. Overall, the results presented in Table 15 reveal that twins with large differences in education are not driving the results presented previously.

9.5 Investigating Reverse Causality

Given the importance of juvenile crime in the previous results it is natural, in addition to simply including juvenile crime as a proxy for unobservables, to analyze the reverse causality between education and crime in further detail by directly estimating the effects of participation in juvenile Table 15: Within Twin Fixed Effect Estimates of Sample of Same Gender Twins In Same Childhood Environments Excluding Twins with Large Education Differences

	(1)	(2)	(3)	(4)	(5)	(6)	
]	Male Twins	ns Female Twins				
Entire Sample							
_	Total	Property	Violent	Total	Property	Violent	
Years of Education	-0.0138**	-0.0040	-0.0052*	0.0018	0.0011	-0.0002	
	(0.0049)	(0.0030)	(0.0021)	(0.0019)	(0.0012)	(0.0010)	
Number Individuals	5236	5236	5236	5416	5416	5416	
Twin Pairs	2618	2618	2618	2708	2708	2708	
At Least One Low	At Least One Low Educated Parent Sample						
	Total	Property	Violent	Total	Property	Violent	

	Total	Toperty	VIOLEIII	Total	Toperty	VIOlent
Years of Education	-0.0192**	-0.0076^+	-0.0059*	0.0016	0.0005	-0.0011
	(0.0072)	(0.0044)	(0.0028)	(0.0023)	(0.0015)	(0.0011)
Number Individuals	2636	2636	2636	2700	2700	2700
Twin Pairs	1318	1318	1318	1350	1350	1350

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. Low educated parent is either a low educated father, low educated mother, or both low educated mother and father. Not reported are controls for juvenile crime. Excludes twins with 4 or more years differences in education.

crime on educational attainment. In addition to the findings of Webbink et al. (2013) already mentioned, Hjalmarsson (2008) analyzes the effects of juvenile crime on education in the US, finding significant negative effects of arrest and incarceration before age 17 on high school graduation, but virtually no effect of being charged with a crime as a juvenile on high school graduation. Additionally, Aizer and Doyle (2015) find, using the randomization of case assignment to judges, that juvenile incarceration significantly lowers the probability of graduating high school and that these effects are strongest at the ages of 15 and 16 due to the fact that incarceration will likely end education for these individuals. While Hjalmarsson (2008) uses criminal activity that occurred when aged 16 or younger, Webbink et al. (2013) examine how the age an individual was first arrested impacts educational attainment, finding that arrests at younger ages (13, 14, and 15) are more detrimental to education than arrests at later ages (16 and 17). As criminal behavior prior to the age of 15 is unobservable in Danish data, and criminal behavior after the age of 16 is outside the compulsory schooling window, the estimation below analyzes the effects of a conviction for a crime committed at the age of 15 or 16 on educational attainment.

Three separate measures of educational attainment are used in Tables A3 & A4 in Appendix E

to estimate the reverse causality between education and crime: whether an individual was educated past compulsory schooling, whether an individual completed high school, and how many years of education an individual received. These measures are reported in columns (1), (2), and (3) respectively. Using only male twin pairs, the estimated effects of juvenile crime are generally negative, but mostly insignificant, while for females, large and negative estimated effects of juvenile crime at the age of 16 on education are seen for all education measures.

While the effects of juvenile crime on education are large, they are much smaller than what has previously been found. Webbink et al. (2013) find approximately a 20 percentage point reduction in the probability of being a senior in high school due to juvenile crime,³² Hjalmarsson (2008) finds an 11 percentage point reduction in the probability of graduating high school, and Aizer and Doyle (2015) find juvenile incarceration leads to a 13 percentage point reduction in the probability of high school graduation. For the purposes of comparison with these existing results, Table A4 in Appendix E defines juvenile crime as being convicted of any crime as a juvenile committed at any point during the ages of 15 and 16, ignoring any effects that juvenile crime age have on educational attainment. As before, the effects of juvenile crime on education are not so large that they dominate the estimated effects of education on crime. For example, Webbink et al. (2013) find that an arrest before the age of 18 reduces education for all twin pairs by 0.7 years,³³ much larger than the results found here.

The results presented in Tables A3 and A4 reveal that crime committed as a juvenile reduces educational attainment and is an important component in the relationship between education and crime. However, the impact of the juvenile crime on education is less than has previously been found, and once accounting for the reverse causality between education and crime, educational attainment still significantly reduces participation in crime as an adult for males. Interestingly, there appear to be larger effects of juvenile crime on education for women. Given that it is less common for women to be involved in crime compared to men, it is possible that engaging in such a rare event as juvenile crime is very detrimental to the educational attainment of women.

 $^{^{32}}$ For identical twins, this effect is 12 percentage points and insignificant.

 $^{^{33}}$ For identical twins, this effect is 0.02 years, but when correcting for missing values, this number climbs to 0.32. Both are insignificant.

10 Conclusions

This paper provides new evidence on the heterogeneous effects of education on crime, overcoming the endogeneity problem that exists between education and crime by making use of twin data contained in detailed Danish Register Data. The paper first confirms a causal link between education and crime in Denmark, finding significant negative effects of education on an individual's probability of conviction for total and violent crimes, with marginally significant effects for property crimes. In the preferred baseline specification, an additional year of education significantly reduces the probability of conviction as an adult for any crime by 14%, for a property crime by 12%, and for a violent crime by 19% for males. The effects of education on adult crime are largely unchanged controlling for whether an individual was convicted of a crime as a juvenile.

Heterogeneous effects are then examined. Family factors, measured by parental education, are found to be important for the effects of education on crime for children, with large crime reduction seen for children of low educated families and virtually no effects on crime seen for children with both parents highly educated. Environmental factors, measured as growing up in a high or low crime neighborhood, have much less impact on the effects of education on crime. Examining how the effects of education on crime differ across crime types reveals education can also reduce traffic related alcohol, sexual, and firearms crimes. Consistent with past literature, the completion of high school is found to reduce criminal propensity for all crime types for male twins, while for female twins, the completion of high school is found to significantly increase the probability of conviction for total and property crimes. Contrary to expectations, the completion of vocational training after lower secondary education is found to have no crime reducing effects for males, while, consistent with expectations, additional university education past high school is found to have no significant crime reducing effects both for males and females.

Due to the potential bias introduced by the presence of DZ twins, and following Conley et al. (2006) and Holmlund et al. (2008), closely spaced siblings are used to account for the presence of these DZ twins. Imputed effects for identical twins remain similar, in terms of magnitude, to the effects of education on crime found and, in most cases, statistically significant. Additionally, the environment twins were raised in is explored in great detail, ensuring the environmental component of twins' development is the same. Education is found to also reduce an individual's probability of incarceration, eliminating concerns that the effects of education on crime seen are not due to

reduced criminality but perhaps something particular in the use of conviction data. Similar effects of education on crime are found when excluding twins with large differences in education, limiting concerns that the findings are driven by a few potentially outlying twins. Direct estimation of the reverse causality between education and crime reveals that while juvenile crime does lead to lower educational attainment as a youth, the effects are much smaller than has previously been found, and are not large enough to explain the entire relationship seen between education and adult crime.

While on the whole, the estimated effects of education on crime using Danish twin data are in line with previous research, this paper improves on existing studies, which predominantly examine changes in compulsory schooling laws, by estimating heterogeneous effects of education on crime which are representative of the entire population while, at the same time, netting out inherent common factors between twins which is crucial to identifying causal effects of education on crime. One obvious concern is that the twins analyzed are unique and not representative of the general population. However, OLS results for twins are comparable not only to the non-twin sibling population, but also to the entire non-twin Danish population as well.³⁴ This is indicative that the effects of education on crime found are representative at least of the Danish population.

From a policy perspective, the findings of this paper reveal not only the overall importance of education in terms of crime reduction, particularly for the completion of high school, but also the importance of family background in the crime reducing capabilities of education. For children of low educated parents, it appears to be extremely beneficial to motivate additional schooling as for these individuals, levels of education obtained can be less than is socially optimal. The high crime reduction from children of low educated families also highlights the importance of the intergenerational transmission of education, as while individuals from low educated families experience large crime reduction from additional education, they may be more likely follow in the educational footsteps of their parents and fail to benefit from the crime reducing capabilities of education. This may be of concern even in Nordic countries, where most intergenerational estimates find that an additional year of parental education increases the education of their children by around 0.1 years (Holmlund et al. 2011). Encouraging individuals from low educated families to remain in education, as well as to avoid engaging in juvenile crime, could offer an effective way to not only reduce longer term educational inequality but also reduce criminality on the whole. The

³⁴Results available on request.

heterogeneous effects outlined reveal the importance of accounting for differences across individuals when estimating the crime reducing capabilities of education, and further research is required to identify precise mechanisms behind these differences.

References

- Aizer, A. and J. J. Doyle (2015). Juvenile incarceration, human capital, and future crime: Evidence from randomly assigned judges. *The Quarterly Journal of Economics* 130(2), 759–803.
- Amin, V. (2011). Returns to education: Evidence from uk twins: Comment. American Economic Review 101(4), 1629–35.
- Ashenfelter, O. and A. Krueger (1994). Estimates of the economic return to schooling from a new sample of twins. *The American Economic Review* 84(5), 1157–1173.
- Ashenfelter, O. and C. Rouse (1998). Income, schooling, and ability: Evidence from a new sample of identical twins. The Quarterly Journal of Economics 113(1), 253–284.
- Åslund, O., H. Grönqvist, C. Hall, and J. Vlachos (2015). Education and criminal behavior: Insights from an expansion of upper secondary school. IZA Discussion Papers 9374, Institute for the Study of Labor (IZA).
- Behrman, J. R. and M. R. Rosenzweig (2002). Does increasing women's schooling raise the schooling of the next generation? *American Economic Review* 92(1), 323–334.
- Björklund, A. and M. Jäntti (2012). How important is family background for labor-economic outcomes? *Labour Economics* 19(4), 465 474.
- Black, S. E., P. J. Devereux, and K. G. Salvanes (2007). From the cradle to the labor market? the effect of birth weight on adult outcomes. *The Quarterly Journal of Economics* 122(1), 409–439.
- Bonczar, T. P. and L. E. Glaze (2009). Probation and parole in the united states, 2007 statistical tables. *Bureau of Justice Statistics, US Department of Justice*, 1–32.
- Bound, J. and G. Solon (1999). Double trouble: on the value of twins-based estimation of the return to schooling. *Economics of Education Review* 18(2), 169 182.
- Buonanno, P. and L. Leonida (2009). Non-market effects of education on crime: Evidence from italian regions. *Economics of Education Review* 28(1), 11–17.
- Calvet, L. E. and P. Sodini (2014). Twin picks: Disentangling the determinants of risk-taking in household portfolios. *The Journal of Finance* 69(2), 867–906.
- Chetty, R., N. Hendren, and L. F. Katz (2015). The effects of exposure to better neighborhoods on children: New evidence from the moving to opportunity experiment. Working Paper 21156, National Bureau of Economic Research.
- Conley, D. and K. W. Strully (2012). Birth weight, infant mortality, and race: Twin comparisons and genetic/environmental inputs. Social Science & Medicine 75(12), 2446 – 2454.
- Conley, D., K. W. Strully, and N. G. Bennett (2006). Twin differences in birth weight: The effects of genotype and prenatal environment on neonatal and post-neonatal mortality. *Economics & Human Biology* 4(2), 151 183.

- Damm, A. P. and C. Dustmann (2014). Does growing up in a high crime neighborhood affect youth criminal behavior? *American Economic Review* 104(6), 1806–32.
- Griliches, Z. (1979). Sibling models and data in economics: Beginnings of a survey. Journal of Political Economy 87(5), S37–S64.
- Hjalmarsson, R. (2008). Criminal justice involvement and high school completion. Journal of Urban Economics 63(2), 613–630.
- Hjalmarsson, R., H. Holmlund, and M. J. Lindquist (2015). The effect of education on criminal convictions and incarceration: Causal evidence from micro-data. *The Economic Journal* 125(587), 1290–1326.
- Hjalmarsson, R. and M. J. Lindquist (2012). Like godfather, like son: Exploring the intergenerational nature of crime. Journal of Human Resources 47(2), 550–582.
- Hjalmarsson, R. and M. J. Lindquist (2013). The origins of intergenerational associations in crime: Lessons from swedish adoption data. *Labour Economics* 20(0), 68–81.
- Holmlund, H., M. Lindahl, and E. Plug (2008). The causal effect of parent's schooling on children's schooling: A comparison of estimation methods. IZA Discussion Papers 3630, Institute for the Study of Labor (IZA).
- Holmlund, H., M. Lindahl, and E. Plug (2011). The causal effect of parents' schooling on children's schooling: A comparison of estimation methods. *Journal of Economic Literature* 49(3), 615–51.
- Huang, C., H. Li, P. W. Liu, and J. Zhang (2009). Why does spousal education matter for earnings? assortative mating and cross-productivity. *Journal of Labor Economics* 27(4), 633–652.
- Isacsson, G. (1999). Estimates of the return to schooling in sweden from a large sample of twins. Labour Economics 6(4), 471–489.
- Kling, J. R., J. Ludwig, and L. F. Katz (2005). Neighborhood effects on crime for female and male youth: Evidence from a randomized housing voucher experiment. *The Quarterly Journal of Economics* 120(1), 87–130.
- Landersø, R., H. S. Nielsen, and M. Simonsen (2015). School starting age and the crime-age profile. *The Economic Journal*. Forthcoming.
- Lochner, L. (2004). Education, work, and crime: A human capital approach. International Economic Review 45(3), 811–843.
- Lochner, L. (2011). Non-production benefits of education: Crime, health, and good citizenship. National Bureau of Economic Research Working Paper Series No. 16722, -.
- Lochner, L. and E. Moretti (2004). The effect of education on crime: Evidence from prison inmates, arrests, and self-reports. *American Economic Review* 94(1), 155–189.
- Lundborg, P. (2013). The health returns to schooling–what can we learn from twins? Journal of Population Economics 26(2), 673–701.
- Lundborg, P., A. Nilsson, and D.-O. Rooth (2014). Parental education and offspring outcomes: Evidence from the swedish compulsory school reform. *American Economic Journal: Applied Economics* 6(1), 253–78.

- Machin, S., O. Marie, and S. Vujić (2011). The crime reducing effect of education. The Economic Journal 121 (552), 463–484.
- Machin, S., O. Marie, and S. Vujić (2012). Youth crime and education expansion. German Economic Review 13(4), 366–384.
- Meghir, C., M. Palme, and M. Schnabel (2012). The effect of education policy on crime: An intergenerational perspective. Working Paper 18145, National Bureau of Economic Research.
- Oreopoulos, P., M. Stabile, R. Walld, and L. L. Roos (2008). Short-, medium-, and long-term consequences of poor infant health: An analysis using siblings and twins. *Journal of Human Resources* 43(1), 88–138.
- Pronzato, C. (2012). An examination of paternal and maternal intergenerational transmission of schooling. Journal of Population Economics 25(2), 591–608.
- Royer, H. (2009). Separated at girth: Us twin estimates of the effects of birth weight. American Economic Journal: Applied Economics 1(1), 49–85.
- Sanbonmatsu, L., J. R. Kling, G. J. Duncan, and J. Brooks-Gunn (2006). Neighborhoods and academic achievement: Results from the moving to opportunity experiment. *Journal of Human Resources XLI*(4), 649–691.
- Sandewall, Ö., D. Cesarini, and M. Johannesson (2014). The co-twin methodology and returns to schooling testing a critical assumption. *Labour Economics* 26(0), 1–10.
- Skytthe, A., K. Ohm Kyvik, N. Vilstrup Holm, and K. Christensen (2011). The danish twin registry. *Scandinavian Journal of Public Health* 39(7 suppl), 75–78.
- Webbink, D., P. Koning, S. Vujic, and N. Martin (2008). Why are criminals less educated than non-criminals? evidence from a cohort of young australian twins. CPB Discussion Paper 114, CPB Netherlands Bureau for Economic Policy Analysis.
- Webbink, D., P. Koning, S. Vujić, and N. G. Martin (2013). Why are criminals less educated than non-criminals? evidence from a cohort of young australian twins. *Journal of Law, Economics,* and Organization 29(1), 115–144.

Appendices

A Data Sources and Explanation

Danish Register Data is composed of various databases through administrative sources, which are interlinked by a consistent and unique personal identifier number. This enables Danish residents to be tracked over time and the various separate databases. There are separate databases for employment data, individual data, family data, education data, income data, and crime data. The crime data used throughout the paper is conviction data which is combined with data on charges to make use of the exact date the offense was committed on.

In Denmark, crimes against penal code are: sexual crimes (comprised of incest, rape, heterosexual offenses, homosexual offenses, public indecency, and prostitution), property crimes (comprised of forgery, check forgery, arson, various degrees of burglary, theft from vehicles, theft of various vehicles, shoplifting, larceny, embezzlement, fraud, blackmail, robbery, handling stolen goods, tax evasion, malicious damage to property, and offenses against property), violent crimes (comprised of violent against public authorities, disturbance of public order, homicide, attempted homicide, varying degrees of assault, intentional bodily harm, offenses against life, and threats), as well as some offenses which are categorized as unknown and other. There are special laws against offenses such as narcotics, firearms, and tax crimes.

B Denmark from 2000-2006

While the time period from 2000-2006 is not a perfectly stable, it is relatively stable both in terms of unemployment and crime. There are no large or sudden increases or decreases during the period for the crime data, and while unemployment does fluctuate from 2002-2004, by 2006, the unemployment rate returns to its 2001 level. Results using other years, while not perfectly identical, give similar coefficients. First two figures are author's own calculation, aggregating individual level data to the national level, while last figure uses labor force survey data available from http://statistikbanken.dk.



Figure A1: Total Reported Crimes from 1997-2008

Source: Author's calculation using data provided by Statistics Denmark.



Figure A2: Total Charged Crimes from 1997-2008

Source: Author's calculation using data provided by Statistics Denmark.



Figure A3: Quarterly Unemployment Rate from Labour Force Survey from 1997-2008

Source: http://statistikbanken.dk, "AKU123: Seasonaly adjusted labour force status in percentage by employment status".

C Within Twin Estimates by Detailed Parental Education

Table A1: Within Twin Fixed Effect Estimates of Sample of Same Gender Twins In Same Childhood Environments By Educational Attainment of Parents

	(1)	(2)	(3)	(4)	(5)	(6)
		Male Twins		F	emale Twir	ns
Both Parents Low	Educated					
	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0243**	-0.0022	-0.0041	-0.0039	-0.0021	-0.0030
	(0.0063)	(0.0041)	(0.0032)	(0.0034)	(0.0021)	(0.0022)
Number Individuals	1322	1322	1322	1304	1304	1304
Twin Pairs	661	661	661	652	652	652

Low Educated Mother, High Educated Father

	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0149^{*} (0.0065)	-0.0049 (0.0034)	-0.0005 (0.0025)	-0.0015 (0.0040)	-0.0031 (0.0038)	0.0000(.)
Number Individuals Twin Pairs	$1188 \\ 594$	$\begin{array}{c} 1188 \\ 594 \end{array}$	$1188 \\ 594$	$\begin{array}{c} 1072 \\ 536 \end{array}$	$\begin{array}{c} 1072 \\ 536 \end{array}$	$\begin{array}{c} 1072 \\ 536 \end{array}$

High Educated Mother, Low Educated Father

	Total	Property	Violent	Total	Property	Violent
Years of Education	-0.0216^{*} (0.0083)	-0.0090 (0.0056)	-0.0118^{*} (0.0058)	0.0000 (.)	-0.0000 (0.0000)	-0.0000 (0.0000)
Number Individuals Twin Pairs	$570 \\ 285$	$570 \\ 285$	$570 \\ 285$	$548\\274$	548 274	548 274

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively. Not reported are controls for juvenile crime.
D Detailed Crime Types

Results for all available crime categories are reported in Table A2.

ronments for De	etailed Crim	ıe Types		F					
	(1) Total	(2) Property	(3) Violent	(4) Sexual	(5) Other	(6) Firearms	(7) Traffic Alcohol	(8) Narcotics	(9) Special Legislation
Years of Education	-0.0129**	-0.0033+	-0.0035*	-0.0012^{+}	-0.0020^{+}	-0.0020*	-0.0066**	-0.0007	0.0019
	(0.0031)	(0.0019)	(0.0016)	(0.0006)	(0.0012)	(0.0010)	(0.0023)	(0.0012)	(0.0014)
Juvenile Crime 15	0.0999^+	0.0421	0.0965^{**}	0.0178	-0.0275	0.0168	0.0415	-0.0194	0.0186
	(0.0517)	(0.0361)	(0.0331)	(0.0123)	(0.0195)	(0.0125)	(0.0339)	(0.0306)	(0.0215)
Juvenile Crime 16	0.1071*	0.0431	0.0041	-0.0082	0.0155	0.0141	0.0426	0.0453	-0.0152
	(0.0447)	(0.0353)	(0.0227)	(0.0075)	(0.0149)	(0.0211)	(0.0312)	(0.0276)	(0.0235)
Number Individuals Twin Pairs	$6038 \\ 3019$	$6038 \\ 3019$	$\begin{array}{c} 6038\\ 3019 \end{array}$	$6038 \\ 3019$	$6038 \\ 3019$	$6038 \\ 3019$	$6038 \\ 3019$	6038 3019	$\begin{array}{c} 6038\\ 3019 \end{array}$
Standard errors clustered respectively.	l at twin level	l reported in I	parentheses.	**, *, and +	correspond to) significance a	at the 1%, 5%	, and 10% leve	sle

ronments for Detailed Crime Types	Table A2: Within Twin Fixed Effect Estimates of Sample of Same Gender Male Twins In Same Childhood
	100d Envi-

E Investigating Reverse Causality

Results showing the estimated effect of juvenile crime on educational attainment are reported in Table A3.

Table A3: Within Twin Fixed Effect Estimates of the Effect of Juvenile Crime on EducationalAttainment of Sample of Same Gender Twins In Same Childhood Environments

	(1)	(2)	(3)
Estimates of Sa	ample of Same Gendered, S	Same Household M	fale Twins
	Educ. Past Compulsory	Completion HS	Years of Educ.
Juvenile Crime 15	-0.0510	-0.0253	-0.1899
	(0.0448)	(0.0458)	(0.2027)
Juvenile Crime 16	-0.0582	-0.0366	-0.1361
	(0.0445)	(0.0454)	(0.1968)
R^2	0.019	0.017	0.018
Number Individuals	6038	6038	6038
Twin Pairs	3019	3019	3019

Estimates of Sample of Same Gendered, Same Household Female Twins Educ. Past Compulsory Completion HS Vers of Educ

	Educ. 1 ast Compulsory	Completion IIS	Tears of Educ.
Juvenile Crime 15	-0.1639	0.0104	-0.0581
	(0.1120)	(0.1028)	(0.4374)
Juvenile Crime 16	-0.0656	-0.2504^{**}	-0.6051^{*}
	(0.0842)	(0.0963)	(0.2875)
\mathbb{R}^2	0.005	0.005	0.004
Number Individuals	5908	5908	5908
Twin Pairs	2954	2954	2954

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

Table A4: Within Twin Fixed Effect Estimates of the Effect of Juvenile Crime on EducationalAttainment of Sample of Same Gender Twins In Same Childhood Environments

	(1)	(2)	(3)
Estimates of Sample	e of Same Gendered, Same	e Household Male	Twins
	Educ. Past Compulsory	Completion HS	Years of Educ.
Any Juvenile Crime 15 or 16	-0.0558^{+}	-0.0305	-0.1523
	(0.0331)	(0.0358)	(0.1509)
\mathbb{R}^2	0.016	0.016	0.017
Number Individuals	6038	6038	6038
Twin Pairs	3019	3019	3019
Estimates of Sample	of Same Gendered, Same	Household Female	Twins
	Educ. Past Compulsory	Completion HS	Years of Educ.
Any Juvenile Crime 15 or 16	-0.1200^{+}	-0.1400^{+}	-0.3600
v	(0.0672)	(0.0749)	(0.2650)
\mathbb{R}^2	0.006	0.008	0.006
Number Individuals	5908	5908	5908
Twin Pairs	2954	2954	2954

Standard errors clustered at twin level reported in parentheses. **, *, and + correspond to significance at the 1%, 5%, and 10% levels respectively.

Job Displacement and Crime^{*}

Patrick Bennett^{\dagger} Amine Ouazad^{\ddagger}

March 2016

Abstract

We use a detailed employer-employee data set matched with detailed crime information (timing of crime, convictions, crime type) to estimate the impact of job loss on an individual's probability to commit crime. We focus on job losses due to displacement, i.e. job losses in firms losing a substantial share of their workers, for workers with at least three years of tenure. Displaced workers are more likely to commit offenses leading to conviction for total crimes and property crimes in the years following displacement. We find no evidence that displaced workers' propensity to commit crime is higher than non-displaced workers before the displacement event, but it is significantly higher afterwards. We find that the impacts of displacement on crime depend on the education, household factors, and post-displacement employment outcomes of displaced individuals.

^{*}We would like to thank Maria Guadalupe, Birthe Larsen, Jay Shambaugh, Dario Pozzoli, Anna Piil Damm as well as participants of the INSEAD Symposium 2014, the Fourth SOLE/EALE World Conference, the 17thIZA/CEPR ESSLE, the 7thTransatlantic Workshop on the Economics of Crime, and seminar participants at the Rockwool Foundation for fruitful comments on preliminary versions of this paper. The authors acknowledge financial and computing support from Copenhagen Business School, INSEAD, and New York University. The usual disclaimers apply.

[†]Copenhagen Business School.

[‡]INSEAD, Boulevard de Constance, Fontainebleau.

1 Introduction

In the last decade, Europe has experienced a "reversal of misfortunes": as crime rates have reached historic lows in the U.S., Europe on the contrary currently experiences historically high crime rates (Buonanno et al. 2011). Crime, arrests, and convictions generate large social costs, and the determinants of crime have been the focus of existing literature (Benson & Zimmerman 2010, Freeman 1999).

Descriptive statistics suggest that in the United States, the peak of crime of the early 1990s approximately matches to the peak of the U.S. unemployment rate in 1994, and a positive relationship between unemployment and crime also exists for Denmark. Gould et al. (2002) uses trade instruments to estimate that wage trends explain more than 50% of the variation in property crime in the U.S. over their sample period (1979–1997), and that the decline in the unemployment rate of non-college-educated men after 1993 contributed to the decline in crime rates. Lin (2008) uses union membership rates and a state's industrial structure to estimate that a one percentage point increase in the unemployment rate leads to a 4 percent increase in property crime.

Prior literature has indeed uncovered convincing causal estimates of the impact of unemployment on crime in a number of countries including Sweden (Öster & Agell 2007) and France (Fougère et al. 2009). However, prior literature relies on state or municipality level data. In particular, it is hard to pinpoint exactly what individual-level mechanism generates the state- or municipality-level relationship between unemployment and crime rates.

This paper estimates the impact of mass layoffs on the individual probability of committing a criminal offense in Denmark. Focusing on Denmark allows us to use a detailed employer-employeeunemployment matched individual-level data set with crime information taken from police records. The data set includes information on convictions, broken down by crime type—property crime, violent crime, etc—as well as individual earnings, weeks of unemployment, age, marital status, family information, and the area of residence of the individual. We focus on displaced workers, i.e. male individuals that have been in employment for at least 3 years in the same firm and move into unemployment when the firm experiences a mass layoff event, i.e. loses a substantial fraction of its employees compared to peak employment in a five year window prior to the time period of analysis. If events that drive the firm's business cycle are arguably independent of the individual dynamics of the employee's criminal offenses, focusing on displaced workers during mass layoff events is likely leading to more causal estimates of the impact of job loss on the probability to commit crime.

This paper finds statistically and economically significant impacts of displacement on the probability of an offense leading to a conviction for total and property crimes. We find that although displaced workers are no more likely to commit crime at any point prior to displacement, displaced workers are substantially more likely to commit crime after displacement. Results are robust to the inclusion of individual fixed effects, controls for family factors, and municipality fixed effects. Results are also robust to alternative definitions for mass layoffs: either (i) changing the threshold (30 or 40%) decline in firm size below which a firm is labeled as experiencing a mass layoff, (ii) using mean firm employment as the reference point for the firm-size decline instead of peak employment (Jacobson et al. 1993), or (iii) identifying mass layoffs as large deviations from a firm-specific trend in employment, estimated using prior firm size changes in 1985-1989. Results are also robust to focusing on larger-sized firms, for which a given percentage decline in size is less likely to be driven by temporary changes in firm size.

We examine a variety of individual and family factors to assess the potential mechanisms behind why displacement leads to increases in crime. We assess whether there exists an intergenerational impact of father's job displacement on the criminality of their children. We see a small impact of father's displacement on son's criminality in the short-run which is, at best, marginally different from the criminality of sons of non-displaced fathers. We find effects of displacement on crime are concentrated for individuals with low education—those with less than high school education and, to a lesser extent, those with vocational training education. Those displaced with education to the university level or higher are no more likely to be convicted than non-displaced individuals following displacement. We show that those living without another adult, either those unmarried or living in a single adult household, are more likely to engage in crime than displaced workers who are living with another adult, while displaced individuals are likely to engage in crime irrespective of age or whether they have children. Displaced workers experience substantial short-run and permanent earnings losses and spend longer in unemployment after the displacement event. While earnings losses and unemployment spells may explain part of the impact of displacement on crime, results suggest that our estimates are an effect of displacement on crime over and above what is explained by earnings losses but that time spent in unemployment can explain substantially more of the impact of displacement on crime. Given the generous unemployment system in Denmark which may in part mitigate the role of earnings losses, this points to the fact that idleness may, at the very least, explain a portion of the impact of job displacement on crime.

This paper makes contributions to two different literatures. First, the paper provides individuallevel estimates of the impact of job losses on crime using detailed employer-employee data. Previous literature (Gould et al. 2002, Öster & Agell 2007, Fougère et al. 2009) used regional-level data (such as county-level or state-level data) to estimate such impacts. Although the literature uses credible instrumental variable estimates, individual-level evidence of a mechanism relating unemployment and crime remains to be established. In particular, no U.S. data set matches individuals with their employers and includes crime data. Focusing on Denmark allows such analysis.

Focusing on individual level data for Denmark provides estimates of a different relevance as compared to U.S. aggregate estimates. Individual-level estimates document the individual-level mechanism that may explain the aggregate level results: in particular a discrepancy between individual-level estimates and area-level estimates suggests either that regional level estimates are confounded or that there are social interactions in crime within states or municipalities: as unemployment rises, both individual incentives and social incentives to commit crime increase (Glaeser et al. 1996), and area-level estimates may be larger than individual level estimates. On the other hand, Denmark differs from the U.S. in significant respects. First, unemployment benefits and social benefits are significantly more generous in Denmark than in the U.S. Second, this paper focuses on the impact of job displacement on offenses leading to a conviction. Per capita incarceration rates are significantly lower in Denmark than in the U.S., and given the vast institutional differences in crime between the two countries, results presented in this paper are arguably a lower bound compared to what would be expected if similar data were available in the U.S.

The paper also makes a contribution to the literature on the wider impacts of job displacement, which has documented the impact of job displacement on earnings (Jacobson et al. 1993, Couch & Placzek 2010), health (Eliason & Storrie 2009, Sullivan & von Wachter 2009, Browning & Heinesen 2012, Black et al. 2012), and mobility (Huttunen et al. 2015). Jacobson et al. (1993) documented the short-run and long-run earnings losses of displaced workers using U.S. Social Security data. Sullivan & von Wachter (2009) present evidence that job displacement leads to higher mortality rates. In this paper, we present results defining displacement in a similar way as in Jacobson et al. (1993) and Sullivan & von Wachter (2009), but we also use declines relative to a firm-specific trend in employment to identify large and sudden changes in firm size. Results should be useful to policymakers: by establishing a link between individual-level displacement and crime, a job separation is likely to impact other parties than the firm and the employee. Job displacement may thus lead to increased policing costs, and overall negative welfare externalities for the municipality. In Blanchard & Tirole (2008) framework, neither employers nor workers may fully internalize the social cost of the job separations, which justifies either additional taxation of employers and/or active labor market policies that incentivize or help unemployed individuals to go back to formal employment.

The paper proceeds as follows. Section 2 presents the rich Danish employer-employee data set. Section 3 presents the identification challenges and the paper's identification strategy. Section 4 describes our empirical findings for convictions by crime types. Section 5 examines the possibility of intergenerational impacts of displacement on children's probability of engaging in crime. Section 6 provides estimates of the impact of displacement by education, while Section 7 examines heterogeneity within our baseline results and identifies which individuals are more likely to be affected by displacement in terms of increased criminality. Section 8 presents results controlling for potential confounding characteristics and estimates what share of the effect of displacement on crime is mediated by the effect of displacement on earnings and unemployment, while Section 9 tests whether results are robust to alternative definitions for mass layoff and displacement. Section 10 concludes.

2 Data Set

To analyze the impact of job displacement on crime, we utilize detailed employer and employee data contained in Danish Register Data made available by Statistics Denmark. Danish Register Data is a database of every individual residing in Denmark from 1980-present which is collected from various governmental and administrative sources. We follow individuals over time and across different data sources via an anonymous personal identification number derived from the central personal register (CPR, Det Centrale Personregister), and are able to match individuals to their employer using a unique firm identification number. We focus solely on males as males are overwhelmingly those who commit crime. We construct an individual level panel of every male residing in Denmark from 1985-2000 by combining five different data registers.

First, the Population Registers include demographic factors such as age, gender, municipality

of residence, and immigrant and marital status. Second, the Danish Student Register contains education data such as an individual's exact educational qualification and educational institution as well as information of any ongoing schooling. Third, the employer-employee data comes from the Integrated Database for Labor Market Research (IDA, Integrerede Database for Arbejdsmarkedsforskning) and contains information on an individual's employment as well as the universe of firms in Denmark in a given year. The employee data set provides information such as an individual's employment status (recorded at the end of November), the number of weeks in the year an individual was unemployed, information on part time or full time status, salary earned in the job , and a workplace (plant) identification number as well as a firm identification number. The employer data contains variables such as the number of employees in a workplace and the number of workplaces in a firm. We use this information to construct a firm level dataset. We consider only an individual's primary job, according to a criteria set by Statistics Denmark which follows the definitions of the International Labour Organization (ILO).¹ All of the employee and employer data contained in IDA is observed annually, as in the French (Abowd et al. 1999) and Pennsylvania (Sullivan & von Wachter 2009) employer-employee data sets.

Fourth, welfare and unemployment insurance payments received are observed at weekly frequency in the Central Register of Labour Market Statistics (CRAM, Det Centrale Register for Arbejdsmarkedsstatistik). This paper's data set thus includes annual measures of an individual's unemployment status by pointing out in which week of the year the individual goes from receiving no benefits to receiving some form of benefits.² As we match individuals to their firm, we know exactly when employees transition in and out of employment with that specific firm.

Finally, crime data comes directly from the Central Police Register, and is available for charges (individuals charged by the police with a crime), convictions, as well as incarcerations. After a crime is reported, if the police suspect someone of committing this crime, this individual is formally charged with the crime. The criminal record then includes the date of the offense and the date charges were filed. After this, we observe whether and when the individual was tried for the crime, and the trial's conviction outcome. The outcome can be either incarceration, a suspended sentence, a fine, a settlement, no charge/warning, or another, less frequent decision

 $^{^1 \}mathrm{See}$ www.dst.dk/kvalitets deklaration/848 for an explanation.

 $^{^{2}}$ In what follows, we use the terms receiving unemployment benefits and weeks of unemployment interchangeably. This could be problematic in that we would misclassify someone as non-displaced if they were displaced, but did not claim any benefits. Past studies have found this to be an unimportant factor, and this is particularly the case for us, as the high tenure individuals we identify are all eligible to receive social assistance or unemployment insurance (if they are a member) following job loss.

(for example a youth program or military punishment). While all of these are possible conviction outcomes, the overwhelming majority of convictions in Denmark result in either probation, a fine, or incarceration. In the estimation that follows we examine all types of convictions.

A unique police case number links the criminal offense across the crime registers and the personal identification number links crimes to the specific offender's in all other registers. The crime register also records the precise day charges were filed, convictions recorded, and incarceration started, which can then be compared to the week unemployment benefits started and the individual's employment spell ended. Across all three crime databases, we also observe a detailed crime code, corresponding to the Danish classification system of offenses. Crimes are comprised of: sexual, violent, property, alcohol related traffic, narcotics, firearms, tax, unknown, and other crimes, as well as crimes against special legislation.³ Within these large categories, the specific kind of offense is also recorded, i.e. burglary within property crimes, assault within violent crimes.

Table 1 provides summary statistics for the individual level panel of males born from 1945-1960 residing in Denmark from 1985-2000 for our five different data sources.

3 Empirical Strategy

A number of identification challenges make the identification of the effect of the loss of employment on crime difficult. In particular, individuals typically do not leave their firm for exogenous reasons. Individuals may choose to leave the labor force altogether, or choose to leave their current job and start an unemployment spell to look for a better match. Employers may also choose to separate from employees who are more likely to commit crime. If individuals leave their firm because of opportunities in the informal sector, OLS results will suggest a positive correlation between changes in employment status and the probability of crime. Such correlation will not likely reflect a causal relationship between employment loss and crime; indeed, current unobservables would drive both the probability of job loss and the probability of committing crime.

The literature has identified a firm-level cause of job losses that should be arguably independent of the worker's individual dynamics. In this paper, we focus on high-tenure workers whose firm experiences a mass layoff event, as in Jacobson et al. (1993).

We focus on individuals born from 1945-1960 that are continuously employed in full-time work (30 hours a week or more) in the same firm who have at least 3 years of tenure in 1989. We also

³Excluding offenses such as traffic fines and accidents, which are also recorded in the police data.

Variable	(1) Mean	(2) SD	(3) Observations
Annual Wage (2000 DKK)	238170	169906	8830448
Weeks Fully Unemployed	2.88	9.06	8830448
Firm size	4124.46	9860.5	7494777
Weeks on social assistance	27.1	17.05	150083
Weeks on UI benefits	16.77	15.02	1271574
A mo	20.02	6 56	8830448
Age Dirth Veen	1052.23	0.50	8830448
Lag than high school	1952.27	4.07	8830448
Less than high school	21.23%	0.4452	8830448
High School	4.20%	0.2006	8830448
Vocational	44.33%	0.4968	8830448
University or beyond	22.75%	0.4192	8830448
Missing education	1.49%	0.121	8830448
Household income (2000 DKK)	484396	451135	8830448
Wage as fraction of HH Income	50.47%	29.97	8830448
Household size	2.89	1.35	8830448
Adults in Household	1.89	0.62	8830448
Number of children	1.05	1.14	8830448
Probability of charge	9 97%	14 80%	8830448
Number of charges	1.66	2 24	200301
Probability of conviction	1.00	13.04	8830448
Probability of conviction Property	1.9170	2 06 ⁰⁷	8830448
Probability of conviction - Property	0.03%	8.00%	8830448
Probability of conviction - Violent	0.13%	3.67%	8830448
Probability of conviction - DUI	0.67%	8.14%	8830448
Number of convictions	2.26	5.89	168517
Probability of conviction to Prison	26.29%	44.02%	168517
Length of prison sentence (days)	2341.89	5844.60	44304

Table 1: Impact of Displacement on Crime

Sample: Danish males born from 1945-1960 who are continuously in Denmark from 1985-2000. Means and standard deviations from relevant variables from five different data sources are reported.

exclude individuals whose firm has fewer than 10 employees immediately prior to the displacement period in 1989 in order to exclude the possibility that small changes in the number of employees constitute a mass layoff event. In addition, we exclude any individual who is enrolled in education during the pre-displacement period, a first or second generation immigrant as residency may be tied to employment, or employed in a firm in 1989 which is publicly owned by the state or municipality as the definition of what a firm is for these employees is less obvious. Finally, we restrict our sample to those who are observable throughout our sample period from 1985-2000.⁴

Individuals identified by our sample construction with high-tenure are more likely to have accumulated firm-specific human capital, more likely to be enjoying a more favorable employeremployee specific match, and are thus more likely to face relatively worse outside options as compared to low-tenure workers. Such workers are also less likely to leave a firm during a mass layoff event.⁵ The restrictions we put on the sample of workers ensure that those individuals we classify as displaced are high tenured workers with strong ties to their firms, for whom displacement is likely to be sudden and unexpected. In our detailed employer-employee dataset, an individual loses employment between year t and year t - 1 if the individual was in employment with their 1989 firm in year t - 1 and has at least one week of unemployment in year t.⁶ In the data weeks of unemployment are identified such that a positive number of weeks of unemployment indicates that the worker either lost his employment or has transitioned back into employment with an unemployment spell in between positions. This paper's data set excludes individuals that leave the workforce as (i) a majority of employees receive either unemployment benefits or social assistance and are counted as unemployed and thus (ii) individuals not in employment in year t are individuals that most likely separated from their firm voluntarily.

A firm experiences a mass layoff event in year t if the firm experiences a decline in employment greater than 30% from that firm's peak of employment from 1985-1989 (before the displacement period).⁷ According to our definition of displacement, an individual is displaced only if they transition from employment into unemployment in a firm who experiences a mass layoff event, where displacement occurs somewhere between year t - 1 and t.

⁴Individuals can exist in the data in one year but not the following year either by emigrating from Denmark or through death. This sample restriction excludes a very small portion of individuals and relaxing this restriction does not alter our results.

⁵By focusing on high-tenure individuals, we are likely to get an underestimate of the impact of job losses on crime given the negative correlation between the probability of committing crime and job tenure.

⁶Appendix A presents a detailed discussion of the unemployment system in Denmark as well as the level and type of benefits available during unemployment.

⁷We also consider two alternative definitions of mass-layoff events in Section 9.

An individual is then displaced if he loses employment (following the above definition) from their 1989 employing firm during a firm-level mass layoff event, and we write $Displaced_{it} = 1$. The sample focuses on mass layoffs that occur in the five years after our pre-displacement period such that an individual can be first displaced in 1990 and last displaced in 1994. We follow workers, both displaced and non-displaced, unconditionally in the post-displacement period from 1990-2000, such that our non-displaced sample is composed of individuals who remain in employment (either with the same firm or another firm), individuals who transition into unemployment in a non-mass layoff firm, individuals who transition into non-employment, and individuals who transition from full-time work into part-time work. This ensures we compare the criminal outcomes of high tenure displaced workers to the high tenure workers identified in our sample construction who are not displaced not only in the short run but also in the long run. Table A1 in Appendix B shows the structure of our final sample as well as the number of displaced individuals in each year of our displacement period, and the number of crimes committed in our final estimation sample.

Within a firm that experiences a 30% or greater reduction in employment, individuals who lose employment may be specific individuals in observable and unobservable dimensions. Specifically, a firm and a set of employees may agree on voluntary layoffs. If such voluntary or selective layoffs affect workers that are more likely to commit crime, results correlating displacement events with criminal outcomes will be upward biased. Workers who are less productive, or whose nominal wage is high compared to the firm's outside options, may be more likely to experience job separations. This is where the availability of a longitudinal dataset of individuals with wage and crime in every time period, with individual identifiers, allows us to control for an individual-specific fixed effect. For instance, childhood experiences, dimensions of educational achievement that are not controlled for, will be absorbed by the worker fixed effect.

Individuals may also experience negative productivity shocks right before the firm's mass layoff event, and thus be more likely to lose employment during that firm's mass layoff event. For instance, the loss of a relative (Bennedsen et al. 2006), changes in marital status (Korenman & Neumark 1991), and other time-varying shocks have been shown to affect either worker pay or worker productivity. Such unobservable time-varying life events in the period from 1990-1994, that are correlated with worker productivity or pay and also with the propensity to commit crime, may confound the estimates of the impact of displacement on crime. To test for such possibility, we observe the criminal outcomes of individuals in all years prior to displacement and all years after displacement. If displacement is truly exogenous to the individual's prior unobservables – including propensity to commit crime – we should not observe that displaced individuals display a more crime-prone history before displacement than other workers.

Another identification issue is that causality could flow from crime to job displacement if, for example, an individual is convicted of or incarcerated for a crime, an employer may separate from the worker as a result. Indeed, the timing of both the displacement event and criminal activity are crucial for our results to be interpreted as the effects of job displacement on crime. A criminal event – left-hand side of the regressions – is an offense that leads to a conviction. Because the dataset includes the date of the offense and a unique identifier for the charges that is linked to the judicial outcome and the sentencing, we trace each conviction back to the date of the corresponding offense.⁸ In that way the data focuses only on severe offenses, i.e. with convictions, but avoids the problem of reverse causality that would occur if we defined a criminal event using the conviction date.⁹ For instance, an individual could commit a crime prior to displacement, but, due to lags between when the offense is committed and being charged and convicted, may not ultimately be convicted until after displacement. Table A2 in Appendix C presents the typical time between when an offense is committed, is charged, is convicted, and is incarcerated for the overall population from 1985-2000 as well as our displaced sample. The lag between when the crime is committed and when an individual is finally convicted can be quite substantial. In this paper, $Conviction_{it} = 1$ when an offense committed in year t results in a conviction. Focusing on the offense date alleviates concerns that the conviction itself may be a driver of employment loss, as this would reveal itself in the dummies prior to displacement.

Noting $Conviction_{it}$ such a criminal event (offense leading to a conviction), we estimate the full dynamics of criminal events pre- and post-displacement event. As such the main specification of this paper considers the regression of criminal events $Crime_{i,t}$ on the full set of dummy variables. The propensity to commit crime is modeled by the following equation:

$$Conviction_{it} = \alpha_i + \underbrace{\sum_{k=0}^{k=\overline{K}} \delta_k(Disp.\ in\ t-k)}_{\text{post-displacement}} + \underbrace{\sum_{k=\overline{K}}^{k=1} \delta_{-k}(Disp.\ in\ t+k)}_{\text{pre-displacement}} + \tau_t + x_{it}\beta + \varepsilon_{it} \qquad (1)$$

⁸For a small percentage of crimes, date of offense is unobservable. For these crimes, date of conviction is used. Results are robust to dropping these crimes.

⁹While convictions corresponds to established guilt of a crime, estimation using charges instead of convictions produces relatively similar results which, if anything, are slightly larger in magnitude.

where (Disp. in t - k) is a set of post-displacement dummy variables, with k ranging from 0 to $\bar{K} = 7$ years, equal to one if the individual was displaced in year t - k and 0 otherwise. Thus the coefficients δ_k measure the impact of displacement on crime for the 7 years following displacement. In particular, δ_k measures both the short-run and the long-run impact of displacement on crime. δ_0 is the coefficient of the impact of displacement on the probability of crime in the year of displacement.

For values k < 0, the coefficients δ_k measure pre-displacement propensity to commit crime. If the event of displacement is truly exogenous with respect to individual time-varying unobservables, the timing of criminal offenses leading to conviction will always be such that, for each individual, the probability of being convicted for an offense is significantly higher after displacement than before displacement. As such the propensity to commit crime k years prior to displacement should not be different from the propensity to commit crime in the year right before displacement. Formally, their estimates $\hat{\delta}_k$, k = -5, ..., -1 of pre-displacement or so-called placebo dummies δ_k should not be statistically significantly different from 0 if displacement is truly independent of the individual's unobservables prior to displacement. In particular if individuals who experience negative family events are more likely to commit crime in the years prior to displacement and are also more likely to be laid off during a mass layoff event, the placebo coefficients δ_k in specification (1) will be significantly above zero.

 τ_t are year dummies that captures national trends in the evolution of crime. Such a control is key, as the fraction of individuals committing offenses declines over the time period of analysis. Unemployment declines too, but the country-level correlation between the decline in crime in Denmark and the unemployment rate may be spurious. Hence the dummy-variables control for country-level trends in crime.

 X_{it} is a vector of individual characteristics that capture changes in, for example, the individual's marital status and the change in whether the individual has children. As such, as individuals that transition from unmarried to married are less likely to commit crime (in correlation) and are less likely to be displaced, not including the time-varying marital status X_{it} in the vector of covariates may lead to an overestimation of the impact of displacement on crime. Similarly if an increase in children reveals an individual's unobservable propensity to be displaced in future years, including the control for children in the regression should dampen the estimate δ_k of the impact of displacement on crime.

In specification (1), residual unobservables ε_{it} are assumed robust and clustered by individual, so that the unobservables in the propensity to commit crime can experience some degree of autocorrelation $Cov(\varepsilon_{it}, \varepsilon_{it'}) \neq 0$ for any two periods t, t'. For examining the impact of a shock on an individual's outcomes over time, controlling for autocorrelation typically leads to substantially wider standard errors.¹⁰

We estimate the effects of displacement on the probability of conviction for an offense committed in year t using fixed effect estimation where α_i absorbs individual-specific non-time-varying observables and unobservables that affect an individual's propensity to commit crime, as in specification (1).¹¹ It is advantageous to control for individual specific time-invariant factors, as these are relevant in determining an individual's criminal propensity both prior to and following displacement. Estimation using an individual fixed effect implies multiple cohorts of displacement are pooled, in the sense that the non-displaced control group will contain everyone who is not displaced at that time period: both non-displaced individuals but also individuals who have not yet, but will be in the future, displaced.

Specification 1 is estimated over the period 1985 to 2000. The first displacement events occur in 1990, but the inclusion of data from 1985 ensures the identification of the pre-displacement dummies δ_k for k = -5 to k = -1 starting with the 1990 displacement cohort. As the sample ends in 2000, those individuals who are displaced later in the 1994 cohort, are only followed for 6 years following displacement.

4 Baseline Empirical Findings

Table 2 presents the results of estimating the impact of displacement on offenses leading to a conviction using fixed effect estimation. Columns (1)-(3) show the impacts of displacement on offenses leading to a conviction using individual fixed effect estimation for total, property, and violent crimes respectively. Authors have suggested, using state- or county-level U.S. data, that unemployment may have a causal impact on property crime as individuals' opportunities in the formal labor market are affected by their job separation (Becker 1974, Raphael & Winter-Ebmer 2001). This paper's identification strategy focuses on job displacement as an arguably exogenous

¹⁰On the importance of controlling for autocorrelation or clustering of residuals in the estimation of treatment effects, see Bertrand et al. (2002).

¹¹The coefficient δ_{-1} is normalized by convention to 0 in individual fixed effect estimation, so that all effects are measured relative to the year just prior to displacement.

source of job separation, as individuals may separate endogenously when they have higher opportunities in the formal sector (Rogerson et al. 2005). Job displacement may also cause other types of crime such as violent crime that have been explained either by economic mechanisms (Fajnzylber et al. 2002) or social and psychological mechanisms (Fergusson et al. 2001). Effects are estimated up to 7 years after displacement and up to 5 years prior to displacement.

The coefficients are percentage point impacts: the coefficient of 0.0052 on the impact of displacement on total crime in the year of displacement indicates the probability of committing an offense leading to a conviction is 0.52 percentage points higher in the year of displacement than it was in the year prior to displacement. Using the mean values reported in the last row of Table 2, the impact of displacement on total crime corresponds to an increase in the probability of conviction of 81% sample mean. Effects for total crime peak immediately after displacement, and vanish almost entirely after 4 years. For property crimes, there are both short-run effects, seen in the first few years after displacement, and long-run effects, seen up to 7 years following displacement, of displacement on crime. This effect is strongest 4 years after displacement, where the probability of a conviction for a displaced worker is 0.56 percentage points higher than a non-displaced worker, relative to the year prior to displacement. In terms of percent of the sample mean, these effects for property crime range from a 100-400% increase in the probability of committing a property crime which leads to a conviction. For violent crimes, there are little effects of job displacement, while the probability of conviction for displaced workers is actually significantly lower 4 years after displacement. However, violent offenses are very rare in our sample of high tenured workers and this significant and negative impact 4 years after displacement is attributable to the fact there are zero displaced workers convicted in this particular year. As there is no strong prior reason to believe that displaced individuals are significantly less likely to commit violent crimes 4 years following displacement, particularly as no effects are seen in the years prior, this significant and negative impact is of little consequence.¹²

Pre-displacement coefficients are an important indicator of the magnitude of potential dynamic selection effects: if individuals that are more likely to be convicted are also more likely to be displaced, we should observe positive and significant effects for the $\hat{\delta}_k$ with k < 0. The coefficients are presented in the "Year -5" to "Year -2" rows. In all cases none of these placebo coefficients are statistically different from 0, suggesting that displaced individuals are not significantly different

¹²While it is meaningful that no one is convicted of an offense committed 4 years after displacement, it could be imagined that with a much larger sample of displaced workers this effect may be negative but insignificant.

	(1)	(2)	(3)
	Total	Property	Violent
Year +7	0.0026	0.0042***	-0.0003
	(0.0025)	(0.0015)	(0.0004)
Year $+6$	-0.0003	0.0025^{**}	0.0000
	(0.0022)	(0.0012)	(0.0005)
Year $+5$	0.0005	0.0007	-0.0003
	(0.0023)	(0.0010)	(0.0004)
Year $+4$	0.0046^{*}	0.0056^{***}	-0.0007***
	(0.0026)	(0.0016)	(0.0002)
Year $+3$	0.0015	0.0014	0.0010
	(0.0024)	(0.0011)	(0.0008)
Year $+2$	0.0028	0.0022^{*}	-0.0004
	(0.0025)	(0.0012)	(0.0004)
Year $+1$	0.0050^{*}	0.0036^{**}	0.0010
	(0.0027)	(0.0014)	(0.0008)
Displacement year	0.0052^{*}	0.0038^{***}	-0.0000
	(0.0027)	(0.0014)	(0.0004)
Year -1		Reference	
Year -2	-0.0000	-0.0001	-0.0003
	(0.0023)	(0.0008)	(0.0004)
Year -3	-0.0009	0.0001	-0.0003
	(0.0023)	(0.0008)	(0.0004)
Year -4	-0.0012	0.0002	-0.0003
	(0.0022)	(0.0008)	(0.0004)
Year -5	-0.0013	-0.0005	-0.0003
	(0.0022)	(0.0006)	(0.0004)
\mathbb{R}^2	0.097	0.089	0.074
Observations		1638016	
Individuals		102376	
Sample Means	0.64%	0.14%	0.03%
~ mpro mouno	0.01/0	0.11/0	0.0070

Table 2: Impact of Job Displacement on Crime

Columns (1)-(3) report the impact of job displacement on the probability of committing an offense leading to a conviction for total, property, and violent crimes respectively. Standard errors are clustered at the individual level. ***, **, and * correspond to significance at the 1%, 5%, and 10% levels respectively.

from non-displaced individuals in terms of their propensity to commit crime prior to displacement. The coefficients are graphically depicted in Figure A1a for offenses leading to a conviction for total crimes and in Figure A1b for offenses leading to a conviction for property crimes in Appendix D.

Results available on request also suggest that individual fixed effects α_i (see specification 1) are positively and significantly correlated both with the probability of displacement and with the probability of a conviction, indicating that they capture a significant share of the individual unobservables that both affect the likelihood of a job separation and of crime. Year dummies capture the national-level declining trend in crime, as aggregate statistics on crime indicate that all types of crime peak in 1994 before declining all throughout the 1990s.

To obtain a detailed sense of the specific types of property crime committed following displacement, Figure A2 in Appendix E breaks down the offenses making up property crimes committed by displaced individuals. Theft is overwhelmingly the most frequent offense, followed by fraud.

5 Household Dynamics

As our data allows us to link not only families but also parents to children, we are able to examine whether job displacement of a father has not only an impact on their criminal participation but also if any impact exists on the criminal participation of their children. Within the job displacement literature, there are numerous studies examining the intergenerational impact of job displacement on numerous outcomes of children, earnings, employment, education, and health (Oreopoulos et al. 2008, Lindo 2011, Schaller & Zerpa 2015). Most studies tend to find that parental job displacement during childhood adversely impacts the outcomes of their children, and emphasize the importance of displacement as a stress inducing event with long lasting repercussions. In addition, there is a wide literature investigating the intergenerational transmission of crime (Hjalmarsson & Lindquist 2012, Meghir et al. 2012, Hjalmarsson & Lindquist 2013), which generally find that more criminal parents cause, at least in part, more criminal children. The literature on the intergenerational transmission of crime emphasizes a number of potential mechanisms explaining the impact such as bad parenting, fostering a negative family culture, negative role model effects, the poor quality of parental peers, direct or indirect transmission of criminal skills, and abusive behaviors transmitting from one generation to the next. Studies generally emphasize that parenting and role modeling effects can explain a large portion of the intergenerational effects. In particular, as males are the dominant gender committing crime, there is the possibility that effects flowing from father to son may matter most. Previous studies on the impacts of unemployment on crime (Öster & Agell 2007) have findings consistent with the fact that instability in the lives of parents may spillover negatively to the lives of their children. Given the findings of these separate strands of literature, parental displacement could have an impact on the propensity to commit crime of their children.

Table 3 reports the effects of father's job displacement on the criminal behavior of sons and daughters (columns 1 and 2) and only sons (columns 3 and 4) for total and property crimes respectively.¹³ We examine these outcomes for children born after 1960, such that displaced individuals are excluded, and define the family as the family in the pre-displacement period in 1989 such that we follow the criminal outcomes of all individuals who were in a 1989 family whose father experienced displacement in the following years independent of whether the children remain living with the same family or not. This is due to the fact that a child may not commit crime while not at home, but commit crime later on, potentially after having moved out. While estimated effects are generally positive for total and property crimes, these are, at best, only marginally significant in the year following displacement for children. We see that the effect is driven entirely by an impact of father's displacement on son's criminal behavior, and there is little evidence that displacement of the father impacts the crime of the daughter. While we find that children of displaced fathers are marginally more likely to engage in crime compared to children of non-displaced fathers one year after displacement, these effects are short lasting and are suggestive of only a minor and short-run impact of father's displacement on son's criminal activity. In terms of potential mechanisms, we see some evidence (in regressions unreported) suggesting that the stress inducing impact of job displacement at the very least explains a portion of this short-run effect as we see adverse effects of job displacement on measures of family structure such as marital status and adult family size in the years after, but not prior to displacement.

6 Displacement Impacts by Education

For education, results are grouped into separate samples by educational attainment (measured prior to displacement in 1989) where we estimate the impact of displacement on property crime separately for individuals with less than high school education, vocational training, and university (or greater) education. In Denmark during the time period of this paper, education was compul-

¹³In the interest of brevity, only post-displacement dummies are reported.

	(1)	(2)	(3)	(4)
	Sons and	Daughters		Sons
	Total	Property	Total	Property
Year +7	0.0025	0.0032	0.0029	0.0040*
	(0.0026)	(0.0022)	(0.0024)	(0.0021)
Year $+6$	-0.0001	0.0004	-0.0010	0.0000
	(0.0024)	(0.0020)	(0.0021)	(0.0018)
Year $+5$	-0.0007	-0.0009	-0.0023	-0.0025
	(0.0024)	(0.0019)	(0.0021)	(0.0016)
Year +4	-0.0010	-0.0012	0.0001	-0.0005
	(0.0022)	(0.0018)	(0.0021)	(0.0016)
Year $+3$	0.0025	0.0019	0.0019	0.0020
	(0.0024)	(0.0020)	(0.0022)	(0.0018)
Year $+2$	0.0016	0.0003	0.0022	0.0009
	(0.0025)	(0.0020)	(0.0023)	(0.0018)
Year $+1$	0.0042*	0.0025	0.0046**	0.0032^{*}
	(0.0024)	(0.0020)	(0.0023)	(0.0019)
Displacement year	0.0020	0.0014	0.0014	0.0006
	(0.0023)	(0.0019)	(0.0021)	(0.0017)
Year -1	· · · · ·	, , , , , , , , , , , , , , , , , , ,	Reference	· · · ·
\mathbb{R}^2	0.191	0.168	0.200	0.178
Observations	163	8016	10	638016
Individuals	102	2376	1	.02376

Table 3: Impact of Job Displacement on Crime

Crime is defined an offense leading to a conviction for children born from 1961 onwards who in 1989 reside in the same household as their father. Columns (1)-(2) use this measure of crime for both sons and daughters, columns (3)-(4) use this measure only for sons. Not reported are pre-displacement variables. Standard errors are clustered at the individual level. ***, **, and * correspond to significance at the 1%, 5%, and 10% levels respectively.

sory to grade 9.¹⁴ After compulsory education, individuals may decide to continue to high school (either directly after compulsory education or via an optional 10th grade) which is geared towards attending university, vocational training which is geared towards professional employment, or terminate education. Given the importance of educational attainment in determining an individual's criminal propensity (Lochner & Moretti 2004, Machin et al. 2011, Hjalmarsson et al. 2015), all else equal, we would expect to see larger effects of displacement on crime for those with less education.

Table 4 reveals that individuals with less than high school education experience sizable, significant, and long lasting effects of displacement on the probability of conviction for property crimes. Those displaced with vocational training education experience significant increases in crime compared to non-displaced individuals, but not as strong as the effects seen for those displaced with less than high school education. For individuals with university education, the estimated effects fluctuate between positive and negative, but never significantly different from zero.

7 Which Individuals Are Most Affected?

In addition to education, we can examine what other factors (if any) predict whether displacement has an impact on crime. Table 5 investigates which types of individuals are most affected by the impact of job displacement on property crimes. While overall, we see large and long lasting effects of job displacement on property crimes, these effects, both in their magnitude and longevity, are likely to differ by family and socioeconomic factors. As with an individual's education, we separate our estimation sample for predetermined levels of four potentially relevant variables in their 1989 values: age, whether displaced individuals have children, marital status, and single adult family. Ages are examined across two birth cohorts within our displaced sample, those born from 1945-1950 and those born from 1951-1960. The effects of age on offending are established to peak at late teens and early 20s, and declining over time. Thus it could be expected that the younger birth cohort may have larger effects of displacement on crime. Those who have any children in 1989 could be more responsible and less likely to offend in order to avoid the possibility of being apprehended and losing time spent with their children. On the other hand, it could be that those with children have a greater need for income in order to maintain their levels of expenditure prior to displacement, increasing the probability of committing crime. Those married and living with at least one other adult in 1989 are, similar to those with children, potentially less likely to offend to avoid the

 $^{^{14}\}mathrm{Recently}$ a grade 0 became compulsory as well.

	(1)	(2)	(3)
	Less than High School	Vocational Education	University or Beyond
		Property	
Year $+7$	0.0051^{*}	0.0042^{**}	0.0026
	(0.0029)	(0.0020)	(0.0032)
Year $+6$	0.0057^{*}	0.0017	0.0020
	(0.0030)	(0.0014)	(0.0030)
Year $+5$	0.0018	-0.0003	0.0020
	(0.0022)	(0.0009)	(0.0030)
Year $+4$	0.0051^{*}	0.0044**	0.0075
	(0.0029)	(0.0019)	(0.0049)
Year $+3$	0.0036	0.0001	0.0047
	(0.0028)	(0.0011)	(0.0040)
Year $+2$	0.0057^{*}	-0.0000	0.0018
	(0.0032)	(0.0011)	(0.0021)
Year +1	0.0070^{**}	0.0035^{*}	-0.0009
	(0.0035)	(0.0018)	(0.0007)
Displacement year	0.0097^{**}	0.0019	-0.0008
	(0.0039)	(0.0015)	(0.0007)
Year -1		Reference	
\mathbb{R}^2	0.078	0.087	0.081
Observations	377024	896672	292256
Individuals	23564	56042	18266

Table 4: Impact of Job Displacement on Crime by Educational Attainment

Samples are defined taking an individual's educational qualification in the pre-displacement period in 1989. Property crime is defined as committing a property offense which leads to a conviction. Not reported are pre-displacement variables. ***, ***, and * correspond to significance at the 1%, 5%, and 10% levels respectively.

possibility of losing time spent with their spouse or fellow household members. Additionally, there is also the possibility both for those married and those living with another adult that income could be pooled over the household, and the impacts of job displacement on personal income could be less meaningful to financial stability. Estimation using whether a displaced individual lives with another adult takes into account the possibility that two individuals could be living together, but not be married.

The impact of job displacement on property crime is presented in Table 5 by displaced individual's age, whether they have children, whether they are married, and whether they live with other adults in columns (1)-(4) respectively. Within the separate birth cohorts, the estimated effects of job displacement on property crimes are positive and relatively similar irrespective of the displaced individual's birth cohort. This suggests that displacement is such a negative and sizable shock to an employee that it is sufficient to push someone on the margin of committing crime to engaging in crime irrespective of their age. Examining the importance of having children reveals that job displacement impacts crime for both those with and without children in 1989. While it could be argued that having children of a young age, as opposed to children of any age, is what matters, the post-displacement effects on property crime are unchanged if children are defined not of any age, but those below the age of 18. The lack of differences by whether displaced individuals have children suggests that family factors may not necessarily predict the existence of effects of job displacement on crime. Where the estimated effects of displacement on crime differ is whether a displaced individual is married or living with another adult in 1989. There is a strong impact of job displacement on the crime participation of unmarried and those living with no other adults and while some effects are seen for those married or those living with at least one other adult, both the short- and long-run impacts of displacement on crime are much smaller in magnitude and, in most instances, not significantly different from zero. Job displacement appears to have meaningful and large impacts on crime, both in the short- and long-run, for individuals who are residing in a single adult household, suggesting the earnings potential of the household plays a role in mitigating the effects of job displacement on crime.

(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Aε	;es	Chile	dren	Mar	ried	Adult	in HH
1945-50	1951-60	Children	No Children	Married	Unmarried	Single Adult	2+ Adults
Prop	oerty	Prop	erty	Prop	oerty	Prop	oerty
0.0024	0.0058**	0.0030^{*}	0.0064^{**}	0.0018	0.0075***	0.0105^{**}	0.0022
(0.0018)	(0.0023)	(0.0016)	(0.0030)	(0.0015)	(0.0029)	(0.0042)	(0.0014)
0.0032^{*}	0.0018	0.0024	0.0026	0.0014	0.0041*	0.0090**	0.0004
(0.0019)	(0.0016)	(0.0015)	(0.0021)	(0.0014)	(0.0022)	(0.0037)	(0.0011)
0.0004	0.0009	-0.0003	0.0025	-0.0005	0.0023	0.0031	-0.0001
(0.0013)	(0.0014)	(0.0010)	(0.0021)	(0.0009)	(0.0019)	(0.0025)	(0.0010)
0.0050**	0.0061^{***}	0.0042**	0.0083^{***}	0.0028*	0.0095^{***}	0.0100**	0.0042^{**}
(0.0022)	(0.0023)	(0.0017)	(0.0032)	(0.0016)	(0.0030)	(0.0040)	(0.0016)
0.0013	0.0014	0.0009	0.0023	-0.0002	0.0037	0.0027	0.0009
(0.0016)	(0.0016)	(0.0013)	(0.0021)	(0.0011)	(0.0022)	(0.0025)	(0.0012)
0.0005	0.0039^{*}	0.0018	0.0031	0.0008	0.0043^{*}	0.0038	0.0017
(0.0014)	(0.0021)	(0.0015)	(0.0023)	(0.0014)	(0.0023)	(0.0029)	(0.0014)
0.0046^{**}	0.0026	0.0033^{*}	0.0041*	0.0014	0.0067^{**}	0.0094^{**}	0.0017
(0.0022)	(0.0018)	(0.0017)	(0.0025)	(0.0014)	(0.0028)	(0.0040)	(0.0014)
0.0027	0.0048^{**}	0.0030*	0.0053^{**}	0.0010	0.0076^{***}	0.0097 **	0.0019
(0.0019)	(0.0022)	(0.0017)	(0.0027)	(0.0014)	(0.0029)	(0.0040)	(0.0014)
			Refer	ence			
0.091	0.087	0.091	0.084	0.083	0.094	0.093	0.087
720704	917312	1213872	424144	1105248	532768	269840	1368176
45044	57332	75867	26509	69078	33298	16865	85511
	$(1) \\ A_g \\ 1945-50 \\ Prop \\ 0.0024 \\ (0.0019) \\ 0.0002* \\ (0.0019) \\ 0.00013 \\ 0.00013 \\ (0.0013) \\ 0.0005 \\ (0.0013) \\ 0.0016 \\ 0.0005 \\ (0.0016) \\ 0.0016 \\ 0.0016 \\ 0.0005 \\ (0.0012) \\ 0.0046^{**} \\ (0.0022) \\ 0.0027 \\ (0.0019) \\ 0.0091 \\ 720704 \\ 45044 \\ (0.0010) \\ 0.0010 \\ 0.0010 \\ 0.0010 \\ 0.0010 \\ 0.0011 \\ 0.0012 \\ 0.0011 \\ 0.0012 \\ 0.0012 \\ 0.0012 \\ 0.0011 \\ 0$	$(1) \qquad (2) \\ Ages \\ 1945-50 \qquad 1951-60 \\ I945-50 \qquad 1951-60 \\ V \\ Property \\ 0.0024 \qquad 0.0058^{**} \\ (0.0018) \qquad (0.0018) \\ (0.0019) \qquad (0.0018) \\ (0.0013) \qquad (0.0016) \\ 0.00050^{**} \qquad 0.0061^{***} \\ (0.0022) \qquad (0.0023) \\ 0.0013 \qquad 0.0014) \\ (0.0013) \qquad (0.0023) \\ 0.0016) \qquad (0.0014) \\ (0.0016) \qquad (0.0014) \\ (0.0016) \qquad (0.0014) \\ (0.0016) \qquad (0.0023) \\ 0.0046^{**} \qquad 0.0061^{***} \\ (0.0014) \qquad (0.0021) \\ 0.0046^{**} \qquad 0.0026 \\ (0.0018) \qquad (0.0022) \\ (0.0018) \qquad (0.0022) \\ (0.0019) \qquad (0.0087 \\ 720704 \qquad 917312 \\ 45044 \qquad 57332 \\ \end{array}$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $

H
abl
ю Сл
Impact
\mathbf{f}
J
9
Di
g
lac
en
ler
H
on
Crime
for
Relevant
Samples

8 Controlling for Confounder Characteristics

Figure A3 replicates the findings of Jacobson et al. (1993), i.e. estimates the impact of job displacement on annual earnings and on the number of weeks of unemployment. Jacobson et al.'s (1993) finding is that displaced workers will experience permanent earnings losses and long-term impacts on weeks of unemployment. Figure A3 presents results of the estimation of their specification on our Danish data. In particular the specification includes an individual fixed effect, year dummies, and the full set of pre- and post-displacement year-level dummies. Figure A3a is for annual salary in Danish Kroner, while Figure A3b is for weeks of unemployment. Figure A3a is similar to JLS's finding that displaced workers experience permanent long-term earnings losses. In this paper's analysis, the permanent loss 7 years after displacement is about 50,000 DKK, or 7,500 USD per year in 2015 dollars. Given the magnitude of such impacts of displacement on long-term earnings in the formal sector, opportunities in the informal sector—burglary, thefts, pilfering—may be become relatively more attractive; and thus the impact of displacement on earnings may explain the impact of displacement on crime.

Tables 6 present estimates of the impact of displacement on offenses leading to a conviction for property crimes, controlling for a variety of family, geographic, and employment factors which were previously unaccounted for. The first column presents results including municipality fixed effects, while the next two columns present estimates conditional on whether a displaced individual has children and marital status in the current year. For example, as we do see an impact of job displacement on family dissolution, it is possible that job displacement impacts crime only, or partially, through its negative impacts on family structure. Likewise it could be that specific municipal factors could also affect an individual's probability of engaging in crime and accounting for these may be important. However, the results of the three columns are virtually unchanged for property crimes, leading to the conclusion that neither family factors nor time-invariant municipality factors have much impact on the increase in crime due to displacement.

The last two columns of Tables 6 present estimates of the impact of displacement on offenses leading to a conviction for property crimes, conditional on the individual's unemployment weeks in the current year (column 4) and conditional on the individual's annual salary in the current year (column 5). Given that lower earnings lead to increased probability of criminal offenses, and that there is a negative correlation between displacement and earnings, we should expect

	(1)	(2)	(3)	(4)	(5)
	Municipality	Children	Married	Weeks	Annual
	Effects	Dummy	Dummy	Unemployed	Salary
			Property Crime		
Year +7	0.0042***	0.0042***	0.0042***	0.0040***	0.0040***
	(0.0015)	(0.0015)	(0.0015)	(0.0015)	(0.0015)
Year $+6$	0.0025**	0.0025**	0.0025**	0.0022*	0.0022*
	(0.0012)	(0.0012)	(0.0012)	(0.0012)	(0.0012)
Year +5	0.0007	0.0006	0.0006	0.0002	0.0004
	(0.0010)	(0.0010)	(0.0010)	(0.0010)	(0.0010)
Year +4	0.0056***	0.0056***	0.0056***	0.0049***	0.0053***
	(0.0016)	(0.0016)	(0.0016)	(0.0016)	(0.0016)
Year $+3$	0.0014	0.0014	0.0013	0.0005	0.0010
	(0.0011)	(0.0011)	(0.0011)	(0.0011)	(0.0011)
Year $+2$	0.0022^{*}	0.0022^{*}	0.0022*	0.0008	0.0018
	(0.0012)	(0.0012)	(0.0012)	(0.0012)	(0.0012)
Year +1	0.0036**	0.0036**	0.0036**	0.0014	0.0030**
	(0.0014)	(0.0014)	(0.0014)	(0.0014)	(0.0014)
Displacement year	0.0038***	0.0038***	0.0038***	0.0021	0.0034**
- •	(0.0014)	(0.0014)	(0.0014)	(0.0015)	(0.0014)
Year -1	· · · ·	· · · ·	Reference		
Control		-0.0003*	-0.0004***	0.0001^{***}	-0.00056***
		(0.0002)	(0.0001)	(0.0000)	(0.0000)
Municipality FEs	Yes	No	No	No	No
\mathbb{R}^2	0.089	0.089	0.089	0.089	0.089
Observations	1638016	1638016	1638016	1638016	1638016
Individuals	102376	102376	102376	102376	102376
Year -1 Control Municipality FEs R ² Observations Individuals	(0.0014) Yes 0.089 1638016 102376	$(0.0014) \\ -0.0003^{*} \\ (0.0002) \\ No \\ \\ 0.089 \\ 1638016 \\ 102376 \\ \\ \end{tabular}$	(0.0014)Reference - 0.0004^{***} (0.0001) No 0.089 1638016 102376	(0.0015) 0.0001*** (0.0000) No 0.089 1638016 102376	(0.0014) -0.00056*** (0.0000) No 0.089 1638016 102376

Table 6: Controlling for Confounder Controls - Property

Control corresponds to the estimate of the variable listed in the column heading. Annual salary is reported per 100,000DKK. Property crime is defined as committing a property offense which leads to a conviction. Not reported are pre-displacement variables. Standard errors are clustered at the individual level. ***, **, and * correspond to significance at the 1%, 5%, and 10% levels respectively.

that the coefficient conditional on earnings will be smaller in magnitude. Contrary to this, the results controlling for earnings are relatively similar, although less precisely estimated, to the baseline results, despite the negative and significant effect of wages. Given that the generous unemployment benefits are available in Denmark, these benefits may diminish the role of lost salary in explaining the effects of job displacement on crime, as individuals still receive a base level of income. However, when controlling for weeks spent in unemployment, the effect of displacement on property crimes is reduced, where all post-displacement coefficients decline in magnitude. The positive and significant effect of weeks of unemployment as well as its impact on the effects of displacement on crime indicate that inactivity, more so than individual earnings losses, may play some role in the links between job displacement and crime.

9 Robustness of Baseline Findings

The paper's baseline results present estimates of the impact of job displacement on crime assuming that firms' mass layoff events occur when the firm's size is lower than 30% of its peak employment, measured between 1985 and 1989 in the pre-displacement period (Jacobson et al. 1993). We examine the robustness of our baseline findings re-estimating our displacement effects considering displaced workers from larger firms, an alternative threshold for a mass layoff event, and two alternative definitions for mass layoff events.

9.1 Changes in Firm-Size Threshold

The paper's baseline estimates of Table 2 present results considering firms with 10 or more employees. Firms in Denmark are of a typically smaller size than firms in the United States, and the threshold of 10 corresponds to firms in 72nd percentile of the distribution of firm size in 1989.¹⁵ However, one may argue that with only 10 workers, a mass layoff event may be caused by a temporary changes in firm size. We thus test the robustness of our results to restricting the data set to firms with 20 or more employees (87th percentile), 25 or more employees (90th), and 50 or more employees (95th) in 1989. Results are presented in columns (2)-(4) of Table 7 and overall show very similar effects of displacement on conviction.

However, it is worth noting that marginally larger effects are seen for the samples with larger firms. One potential shortcoming in using displacement as an identification strategy is that while an individual's job loss is unanticipated, their fellow co-workers in their potential network may also experience a similar employment shock. It could be, in the case of very large firms, that these mass layoff events will be sufficiently large to alter an individual's potential network, in which case some of the effects we attribute to job displacement on crime could also be due to the effects of a change in one's network on crime. However, given that these effects are only marginally larger in these robustness specifications, are well within the confidence intervals of the baseline findings, and that (in regressions unreported) excluding individuals in firms in the top 1 percentile of firm size in 1989 produces very similar results to the baseline estimation, this issue seems minor, but is worth taking into consideration.

 $^{^{15}\}mathrm{Excluding}$ those firms which are a self-employed individual.

	(1)	(2)	(3)	(4)	(5)
	Baseline	20+	25+	50+	40% JLS
	Estimation	Employees	Employees	Employees	Threshold
	Property Crime				
Year +7	0.0042***	0.0049***	0.0054***	0.0061***	0.0025*
	(0.0015)	(0.0017)	(0.0019)	(0.0022)	(0.0014)
Year +6	0.0025**	0.0025^{*}	0.0028**	0.0023	0.0023^{*}
	(0.0012)	(0.0013)	(0.0014)	(0.0015)	(0.0014)
Year +5	0.0007	-0.0001	0.0001	-0.0001	0.0012
	(0.0010)	(0.0009)	(0.0010)	(0.0011)	(0.0012)
Year +4	0.0056***	0.0067***	0.0068***	0.0070***	0.0048***
	(0.0016)	(0.0018)	(0.0019)	(0.0022)	(0.0017)
Year +3	0.0014	0.0017	0.0019	0.0023	0.0005
	(0.0011)	(0.0013)	(0.0014)	(0.0016)	(0.0011)
Year +2	0.0022*	0.0031**	0.0035**	0.0033*	0.0012
	(0.0012)	(0.0015)	(0.0016)	(0.0018)	(0.0013)
Year +1	0.0036**	0.0040**	0.0044**	0.0038**	0.0036**
	(0.0014)	(0.0016)	(0.0017)	(0.0019)	(0.0016)
Displacement year	0.0038***	0.0046***	0.0046***	0.0046**	0.0042**
	(0.0014)	(0.0017)	(0.0017)	(0.0020)	(0.0017)
Year -1	()	()	Reference	()	()
Year -2	-0.0001	0.0002	0.0003	0.0007	0.0001
	(0.0008)	(0.0009)	(0.0010)	(0.0012)	(0.0009)
Year -3	0.0001	0.0004	0.0006	0.0004	0.0004
	(0.0008)	(0.0009)	(0.0010)	(0.0011)	(0.0009)
Year -4	0.0002	0.0005	0.0006	0.0005	0.0005
	(0.0008)	(0.0009)	(0.0009)	(0.0011)	(0.0009)
Year -5	-0.0005	-0.0008	-0.0007	-0.0007	-0.0005
	(0.0006)	(0.0006)	(0.0006)	(0.0007)	(0.0006)
\mathbb{R}^2	0.089	0.090	0.089	0.087	0.089
Observations	1638016	1472016	1407120	1201344	1638016
Individuals	102376	92001	87945	75084	102376

Table 7: Robustness

Column (1) corresponds to baseline estimation for property crime. Property crime is defined as committing a property offense which leads to a conviction. Columns (2)-(4) alter the number of employees in 1989 restriction, column (5) increases the criteria for a mass-layoff event. Standard errors are clustered at the individual level. ***, **, and * correspond to significance at the 1%, 5%, and 10% levels respectively.

9.2 Increasing JLS Criteria

Column (5) of Table 7 examines defining mass layoff events as a year when firm size is 40% lower than its peak from 1985-1989 compared to the baseline of reduction of 30%. Using a threshold of 40% should lead to less frequent mass layoff events and thus to more conservative estimates of the impact of displacement on crime.¹⁶ The estimated effects of displacement on property crime using the 40% threshold are similar in magnitude to the baseline estimation, with slightly less statistical significance in the long-run effects of job displacement on crime.

9.3 Alternative Definitions for Firm-Level Mass Layoff

Additionally, Figure 1 examines two alternative definitions of the reference point used to define mass layoff: a year in which firm size is 30% below the firm's average employment in 1985-89 rather than its peak employment and a year in which firm size is 30% below a predicted firm-specific trend in employment. Such a firm-specific trend in employment is constructed as follows. First, for each firm, a linear regression of firm size on years since 1985 is run, for years between 1985 and 1989: $FirmSize_{jt} = Constant_j + Slope_j(Year - 1985)_t$, where j indexes firms. When the firm has increasing size, we set $Slope_j = 0$ and use its average employment to keep only genuine declines in employment, such that the firm-specific trend is only used for firms with declining employment. Such linear trend is then used to predict firm size after 1989. By using this firm-specific trend, we exclude the possibility that a firm whose employment is in slow decline throughout the entire sample period could eventually be classified as a firm with a mass layoff event, if gradual declines in employment persisted over multiple years. Within such a framework, the firm experiences a mass layoff event if its size in year t for t = 1990, ..., 1994 is 30% lower than its predicted size according to this linear extrapolation.

These two alternative definitions should lead to more conservative estimates of job displacement on crime as using average levels of employment to calculate a mass layoff will understate mass layoff events for firms with increasing employment trends who then experience a sudden decline in employment during the displacement period. Despite this, Figure 1 shows that for all definitions of firm-level mass layoff events, the effects of job displacement on property crimes are positive and long lasting. While these results are overall similar to the results of the baseline estimation, these alternative definitions produce effects of displacement on property crimes which are smaller

¹⁶Indeed the total number of displaced individuals decreases by slightly more than 800 individuals.

in magnitude.





Baseline corresponds to baseline estimation for property crimes. Blue and red lines define a mass layoff as a 30% decline from the average number of employees in a firm from 1985-1989 and a 30% decline from a firm-specific trend in employment respectively.

10 Conclusion

Using a data set of high-tenure male workers from 1985-2000 in Denmark, we examine the relationship between unemployment and crime at the individual level by estimating the impact of job displacement on an individual's probability to commit a criminal offense leading to a conviction. Following Jacobson et al. (1993), we define as a mass layoff event where firms lose 30% of their employment from their peak employment during a pre-displacement period. We find that displaced workers are significantly more likely to engage in total and property crimes than their non-displaced counterparts following displacement, but importantly, are not significantly different prior to displacement. We find that individuals with less than high school, and slightly less so, vocational education are proportionately more affected by displacement than individuals with university education. The effects are both long lasting and economically significant, lasting up to 7 years following displacement.

We show that there are little effects of father's displacement on the criminality of sons and daughters, where we estimate a marginally significant short-run increase in crime attributable to an impact of father's displacement on sons. We find that individuals residing in a single adult household are more likely to be affected by displacement than those either married or living with another adult, but see impacts of displacement on crime irrespective of an individual's age and whether a displaced individual has children. Estimating the effects of displacement on annual salary and weeks of unemployment, we conclude that these effects of displacement on crime are partly driven by the impact that displacement has on unemployment, and less so by the impact on earnings. We also examine the possibility that a relationship between marital status, number of children, or municipality factors drives our results, and find that results are robust to the inclusion of such controls—even though the probability of being married is lower after displacement as compared to after displacement. Our results are robust to a variety of specification adjustments, including changing our definition of a mass layoff event and increasing the number of employees required to be laid off to cause a mass layoff event. Our results are useful to policymakers and suggest that job separations are likely to have impacts on society which extend beyond employers and employees, which may provide scope for welfare increasing labor market policies.

References

- Abowd, J. M., Kramarz, F. & Margolis, D. N. (1999), 'High wage workers and high wage firms', *Econometrica* 67(2), 251–333.
- Becker, G. S. (1974), Crime and punishment: An economic approach, in 'Essays in the Economics of Crime and Punishment', pp. 1–54.
- Bennedsen, M., Perez-Gonzalez, F. & Wolfenzon, D. (2006), Do ceos matter?, Nyu working paper no. fin-06-032.
- Benson, B. L. & Zimmerman, P. R. (2010), Handbook on the Economics of Crime.
- Black, S. E., Devereux, P. J. & Salvanes, K. G. (2012), Losing heart? the effect of job displacement on health, Working Paper 18660, National Bureau of Economic Research.
- Blanchard, O. J. & Tirole, J. (2008), 'The joint design of unemployment insurance and employment protection: a first pass', *Journal of the European Economic Association* **6**(1), 45–77.
- Browning, M. & Heinesen, E. (2012), 'Effect of job loss due to plant closure on mortality and hospitalization', *Journal of Health Economics* **31**(4), 599 616.
- Buonanno, P., Drago, F., Galbiati, R. & Zanella, G. (2011), 'Crime in europe and the united states: dissecting the 'reversal of misfortunes'', *Economic Policy* **26**(67), 347–385.

- Couch, K. A. & Placzek, D. W. (2010), 'Earnings losses of displaced workers revisited', American Economic Review 100(1), 572–89.
- Eliason, M. & Storrie, D. (2009), 'Does job loss shorten life?', Journal of Human Resources 44(2), 277–302.
- Fajnzylber, P., Lederman, D. & Loayza, N. (2002), 'What causes violent crime?', European Economic Review 46(7), 1323–1357.
- Fergusson, D. M., John Horwood, L. & Woodward, L. J. (2001), 'Unemployment and psychosocial adjustment in young adults: causation or selection?', *Social science & medicine* 53(3), 305–320.
- Fougère, D., Kramarz, F. & Pouget, J. (2009), 'Youth unemployment and crime in france', *Journal* of the European Economic Association 7(5), 909–938.
- Freeman, R. (1999), The economics of crime, *in* 'Handbook of Labor Economics', chapter The economics of crime, pp. 3529–3571.
- Glaeser, E. L., Sacerdote, B. & Scheinkman, J. A. (1996), 'Crime and social interactions', The Quarterly Journal of Economics 111(2), 507–548.
- Gould, E. D., Weinberg, B. A. & Mustard, D. B. (2002), 'Crime rates and local labor market opportunities in the united states: 1979–1997', *Review of Economics and Statistics* 84(1), 45–61.
- Hjalmarsson, R., Holmlund, H. & Lindquist, M. J. (2015), 'The effect of education on criminal convictions and incarceration: Causal evidence from micro-data', *The Economic Journal* 125(587), 1290–1326.
- Hjalmarsson, R. & Lindquist, M. J. (2012), 'Like godfather, like son: Exploring the intergenerational nature of crime', *Journal of Human Resources* 47(2), 550–582.
- Hjalmarsson, R. & Lindquist, M. J. (2013), 'The origins of intergenerational associations in crime: Lessons from swedish adoption data', *Labour Economics* **20**, 68 – 81.
- Huttunen, K., Moen, J. & Salvanes, K. G. (2015), Job loss and regional mobility, IZA Discussion Papers 8780.
- Jacobson, L. S., LaLonde, R. J. & Sullivan, D. G. (1993), 'Earnings losses of displaced workers', The American Economic Review 83(4), 685–709.
- Korenman, S. & Neumark, D. (1991), 'Does marriage really make men more productive?', Journal of Human Resources pp. 282–307.
- Lin, M.-J. (2008), 'Does unemployment increase crime?: Evidence from u.s. data 1974-2000', Journal of Human Resources 43(2), 413–436.
- Lindo, J. M. (2011), 'Parental job loss and infant health', *Journal of Health Economics* **30**(5), 869–879.
- Lochner, L. & Moretti, E. (2004), 'The effect of education on crime: Evidence from prison inmates, arrests, and self-reports', *American Economic Review* **94**(1), 155–189.
- Machin, S., Marie, O. & Vujić, S. (2011), 'The crime reducing effect of education', The Economic Journal 121(552), 463–484.
- Meghir, C., Palme, M. & Schnabel, M. (2012), The effect of education policy on crime: An intergenerational perspective, Working Paper 18145, National Bureau of Economic Research.
- Oreopoulos, P., Page, M. & Stevens, A. H. (2008), 'The intergenerational effects of worker displacement', Journal of Labor Economics 26(3), 455–483.
- Öster, A. & Agell, J. (2007), 'Crime and unemployment in turbulent times', *Journal of the European Economic Association* 5(4), 752–775.
- Raphael, S. & Winter-Ebmer, R. (2001), 'Identifying the effect of unemployment on crime', *Journal* of Law and Economics 44(1), 259–283.
- Rogerson, R., Shimer, R. & Wright, R. (2005), 'Search-theoretic models of the labor market: A survey', *Journal of Economic Literature* **43**, 959–988.
- Schaller, J. & Zerpa, M. (2015), Short-run effects of parental job loss on child health, Working Paper 21745, National Bureau of Economic Research.
- Sullivan, D. & von Wachter, T. (2009), 'Job displacement and mortality: An analysis using administrative data', The Quarterly Journal of Economics 124(3), 1265–1306.

A Unemployment in Denmark

When an individual becomes unemployed in Denmark, they can receive either unemployment insurance benefits or social assistance. In order to be eligible for unemployment insurance, an individual must, voluntarily, pay monthly membership into a unemployment insurance fund. As a member of an unemployment insurance fund, an individual receives 90% of their original salary, subject to a maximum and eligibility rules. During the time period in this paper, these eligibility criteria were membership into an unemployment insurance fund for a minimum of one year, and employment for at least half a year out of the previous 3 years before unemployment. If an individual was not a member of an unemployment insurance fund, they still may be eligible for social assistance. Social assistance in Denmark is means-tested, and less clearly defined compared to unemployment benefits. The amount an individual would receive depends on a variety of factors such as household size, marital status, spousal earnings, and household wealth. For example, a single father with kids would receive fairly large welfare payments, while a married father whose spouse earns a high income would receive very little welfare payments. While all these factors go into the level of benefits one would receive, the precise amount received ultimately depends on an assessment determined by a welfare counselor. Membership in an unemployment insurance fund is relatively common in Denmark, where during the 1990s, membership among employed workers was around 85%. Unemployment insurance membership does vary across socio-economic factors and earnings, due to the fact that some individuals would receive similar payments whether they joined a fund or not, while some individuals would receive nearly no social assistance payments but high unemployment insurance payments.

B Structure of the Estimation Sample

Year	1985	1986	1987	1988	1989	1990	1991	1992	1993	1994	1995	1996	1997	1998	1999	2000
	Ι	Pre-Dis	placeme	ent Peri	od				P	ost-Dis	placeme	nt Peri	od			
							Displa	acement	t Years							
Observations											I					
Final panel			102,37	6							102,376	0.				
Displacement	0	0	0	0	0	647	710	623	660	321	0	0	0	0	0	0
Crime Property	122	95	78	98	89	114	155	216	226	216	189	136	157	154	135	86
Violent	39	26	26	30	24	36	28	25	40	43	34	30 25	30	27	27	23
Total	713	687	674	609	663	723	693	794	767	730	623	575	600	612	591	427
Reported are the crimes in each ye	e final nu ear for t]	ımber o he estin	f individ nation sa	luals in mple ov	each year rer the pre	of the dat - and pos	ta, the s t-displa	ums of c cement]	displace periods.	ment for	the dis	placeme	nt years	, and th	e sums o	of
crimes in each y€	ear for t	he estin	nation sa	mple ov	er the pre	- and pos	t-displa	cement]	periods.							

C Time Between Charges, Convictions, and Incarcerations

Time from Offense to Charges (days) Time from Charges to Conviction (days) Time from Convic	lays) Time from	Conviction to Prison (days)
Mean Median P25 P75 Charges Mean Median P25 P75 Convictions Mean Median P2	ictions Mean Media	un P25 P75 Prison Te
Sample: all of Denmark with at least one charge over sample period	ge over sample period	
Total Crimes 78.1 1 0 44 2759322 136 94 43 180 1172128 170.6 124 47	72128 170.6 124	47 229 213246
Sample: displaced individuals with at least one charge over sample period	harge over sample period	
Total Crimes 81 0 0 18 922 129.9 89.6 52 151.4 646 203.2 166 106	346 203.2 166	106 236.8 117
(70.1%)	1.1%)	(18.1%
Top panel reports the time between charges, convictions, and incarcerations for all individuals charged by the police from 1985-2000. reports the same times for individuals who are displaced at any point from 1990-1994 from 1985-2000. For confidentiality reasons, the the median, and the 75 th percentile are calculated using the average observations of the 5 individuals surrounding each statistic.	arged by the police from 198 2000. For confidentiality reas uals surrounding each statist	5-2000. Bottom panel sons, the 25 th percentile, .ic.

Table A
2: Timin
g of J
udicial
Process
\mathbf{for}
Overall
Sample
and
Displaced
Sample

D Graphical Representation of Baseline Estimates

Figure A1: Impact of Job Displacement on Probability of an Offense Leading to a Conviction

(a) Total Crimes 0.006 0.005 0.004 Displacement Effect 200.0 Displacement Effect 200.0 0 -5 -4 -3 -2 -1 0 1 2 3 4 5 7 -0.001 -0.002 Years Since Displacement (b) Property Crimes 0.006 0.005 0.004



Plots correspond to baseline estimates in Table 2 for total and property crimes.

E Detailed Property Crimes



Figure A2: Detailed Property Crimes Committed by Displaced Individuals Post-Displacement

Percentages correspond to the fraction of specific offenses within the property crimes committed by displaced individuals following displacement. "Other" corresponds to arson, burglary, extortion, and embezzlement and is aggregated in order to comply with Statistics Denmark's data confidentiality policies.

F The Impact of Job Displacement on Earnings Losses and Unemployment



Figure A3: Impact of Job Displacement on Employment Outcomes

(a) Impact of Displacement on Annual Salary

Plotted are coefficients from estimation of pre- and post-displacement dummies on weeks unemployed and annual salary respectively.

Chapter 3 - Negative Attitudes, Network and Education

Negative Attitudes, Network and Education

Patrick Bennett, Lisbeth la Cour, Birthe Larsen, Gisela Waisman[†]

March 2016

Abstract

This paper assesses, both theoretically and empirically, the potential explanations behind the educational gap between young natives and immigrants using two measures, negative attitudes towards immigrants and networking. The paper considers the impact of negative attitudes and networking and that these parameters may influence high and uneducated workers as well as immigrants and natives differently, creating different incentives to acquire education for the two ethnic groups. Theoretically, this paper concludes that if all immigrants are equally affected by discrimination, immigrants obtain less education than natives while if only low-educated immigrants are affected by negative attitudes, immigrants obtain more education than natives to improve their employment prospects. Using rich Danish administrative data, this paper finds evidence consistent with this second case, that greater negative attitudes have a positive impact on male immigrants decision to acquire education and that networking can also increase immigrant education.

We want to thank participants at the Search and Matching conference in Edinburgh 2014, the Workshop on Gender and Ethnic Differences in Market Outcomes, Aix-en-Provence 2014, the Copenhagen Education Network Seminar December 2014, School of Economics, Singapore Management University 2015. the RES conference 2015 in Manchester, the EEA conference 2015 in Mannheim, the WEAI conference in Singapore 2016, and Bochum University 2015, Kevin Lang, John Kennes, Pietro Garibaldi, Linas Tarasonis, Dario Pozzoli, and Anna Piil Damm. Finally, we want to thank Simon Backlund for excellent research assistance.

^{*}Department of Economics, Copenhagen Business School.

[†]Regeringskansliet, Stockholm.

1 Introduction

An OECD report from 2006 reveals that immigrant and immigrant offspring at a very young age express equal or sometimes even higher motivation to learn mathematics than their native counterparts and very positive attitudes towards school and education in general.¹ However, at the age of 15, they under perform compared to the natives. More than a third of the first and second generation immigrant children in Austria, Belgium, Denmark, Germany, Norway and the USA, who have spent all their entire schooling in the host country, perform below the baseline PISA benchmark for mathematics performance, a period at which students begin to demonstrate the kind of skills that enable them to actively use mathematics.² Furthermore, when taking their parental background into account, immigrants tend not to perform as well in school as their native peers.³ This fact may then, in turn, influence their choice of further education, and eventually their labour market outcome and performance.

When explaining the educational gap between immigrants and natives, measures which impact immigrants and natives differently are likely to be important. The aim of this paper is to discover the factors that shift the motivation and performance of immigrants when the decision about education beyond compulsory school is taken. For the educational decision, workers compare the value corresponding to acquiring education to the value of not acquiring education. These values depend on the expected incomes which are influenced by both the employment probability as well as wages. The novelty of this paper is to examine theoretically, as well as empirically, whether negative attitudes towards immigrants and networking could influence immigrant employment chances, as well as immigrant wages, differently for educated workers and uneducated workers compared to the same variables for natives. If this were the case, the value of acquiring education may be impacted differently for natives and immigrants and as such, may explain the educational gap between natives and immigrants.

In particular, we will examine the effect of negative attitudes towards immigrants in a region and potential impact of networking through individuals of the same ethnicity living in a region. Negative attitudes towards immigrants may cause discrimination, implying that workers are fired or decide to quit a job. This lowers the value of employment, through both shorter employment

 $^{^{1}}OECD 2006$

²ibid.

³Nielsen and Rangvid

periods and lower wages, as the bargaining power of immigrants falls which in turn affects the value of acquiring education.

There are some empirical papers on discrimination and employment and wages (see for example Waisman and Larsen 2015, Kofi Charles and Guryan 2008) but, to our knowledge, no papers on the additional impact through these channels on education. Concerning networking, immigrants from the same home country or region may increase the likelihood of getting a job and improve labour market performance. Hence, more well-educated immigrants from the same home country or region may increase the return of education, implying that more immigrants acquire education. This may work in different ways. Social networks may influence employment outcomes: the more employed contacts the individual has, the more likely it is that the individual will learn about new job openings (Calvo-Armengol and Jackson 2004, Hellerstein et al 2009) and networks may influence both wages and employment opportunities (Fontaine 2007, Galeanios 2014, Damm 2014). Similarly, empirical research confirms that (see for example Andersson et al 2009, Solignac and Tô 2015) more immigrants living in areas with a large number of employed neighbours are more likely to have jobs compared to immigrants living in areas with fewer employed neighbours. This could be due to networking and/or social norm effects. Furthermore, Kramarz and Skans (2014) show for Swedish data that family networks are important, in terms of obtaining the first job after graduation, and that this impact is stronger for youth of uneducated parents and immigrants in regions with high unemployment. Hence, networking may increase employment probability, and more networking among immigrants may, to some extent, offset the decrease in employment perspectives and wage modifications due to negative attitudes or discrimination.

We formulate a Becker-style taste discrimination model within a search and wage bargaining setting. Bowlus and Eckstein (2002), Flabbi (2010), Mailath et al. (2000), and Lang et al. (2005) study discrimination in the presence of search frictions but with no educational decision. We assume that potential negative tastes towards immigrants imply that their separation rate from the job is higher than the separation rate of a native worker. This may be due to both the worker deciding to quit and the employer firing the worker. This assumption allows us to assume that neither job searchers nor employers know whether discrimination will take place in a particular firm; all that is known is that immigrants face a higher separation rate than natives. We show that immigrants' potential higher separation rate, ceteris paribus, also implies that their employment chances fall as firms, in turn, supply fewer vacancies. Natives and immigrants decide whether to educate or not. They are aware of the existence of discrimination in the labour market and of the possibility of influencing their chances of getting employed through networking. In terms of negative attitudes towards immigrants, we consider two different cases. In the first case, all immigrant workers are affected by negative attitudes towards them and in the second, only low-educated workers are affected. The channel through which the educational level is affected by networking and negative attitudes in our model is through the impact on the expected employment perspectives. However, the possibility that negative attitudes also influence the value of being unemployed directly, that is, over and above the impact on wages and employment chances, could easily be included in the theoretical model and is consistent with the empirical analysis which we perform.

Empirically, we analyse the educational gap between immigrants and natives using Danish Register Data at both the municipality and individual level. Due to the excellent quality of the Danish Register Data, we have the whole population, can link to family members, and have information on employment, education, income, etc. More specifically, we analyse the impact of networking and negative attitudes on education by considering how young immigrants' high school decision, which is not obligatory in Denmark, depends on the the number of people of their own nationality and negative attitudes towards immigrants in the area where they live relative to the impact on young natives. We examine this decision to attend high school as it is made at a young age, around 16, and individuals will usually be living at home while attending high school. The advantage of this is that the parents decide where to live and young immigrants and natives then decide whether to continue in high school. We therefore have that the household placement is, plausibly, exogenous for the person making the educational decision; that is the young immigrants and natives are not both deciding where to live and deciding whether to attend high school.

Despite this, there are concerns that unobservable factors could drive parents, either in their emigration or subsequent relocation, to locate in order to give the young immigrant or native a better choice of high school prior to this high school decision. While we control for parental characteristics together with a variety of municipality controls, we address these potential omitted unobservable characteristics in two ways. Firstly, we allow for the possibility that individuals can relocate due to educational considerations by estimating, as a robustness check, the high school decision for only those who have not recently moved. Secondly, we directly examine the importance of unobservable characteristics compared to observables in explaining our estimated effects using a procedure developed in Oster (2015). While there are reasons to believe that concerns over the importance of unobservables may be mitigated due to the timing of the decision to attend high school, we are able to directly quantify how our estimates change depending on the degree of omitted variable bias due to unobservables.

In the macro-econometric level analysis, we exploit the panel nature of our data to control for unobserved time-invariant factors which affect the fraction of young individuals attending high school. We find positive, but imprecisely estimated, effects of negative attitudes on the fraction of immigrants attending high school. We see little impact of networking on high school attendance, and turn to analysis at the individual level to not only more precisely estimate an individual's potential network but also take into account important family and individual level factors.

At the micro-econometric level we see a positive and significant impact of negative attitudes on male immigrants' probability to attend high school and positive, but less precise effects for females. We see no effects, either positive or negative, for natives. We find that networking matters for immigrants, but indications that the quality of an immigrant's potential network matters for males while only the quantity matters for females. These results on negative attitudes, and to a lesser extent networking, are robust to the exclusion of households who have moved recently, within the past 3 or 6 years. Under reasonable assumptions about the importance of unobservables, we are able to bound the estimated effect for males away from zero; that is we can state with a good deal of confidence that even when accounting for potentially correlated unobservables, negative attitudes have a positive impact on high school attendance for male immigrants. Lastly, we see that the negative attitudes measure matters only for 1st generation immigrants and not for 2nd generation immigrants, who are likely more assimilated and less likely to be adversely impacted by negative attitudes. Overall, our empirical findings are consistent with the second case of the theoretical model, where negative attitudes are prevalent only in the low-skilled sector and more severe negative attitudes increase the incentives of immigrants to acquire education.

The paper is structured as follows. In section 2 the model is setup, then the following sections consider the impact of negative attitudes towards immigrants and the fraction of immigrants. In Section 6 we consider heterogenous networking effects. Sections 7 and 8 provide a macroeconometric and a micro-econometric analysis. Section 9 explores the robustness of the microeconometric results, and Section 10 concludes.

2 The Model

We consider a search and matching model with natives, N and immigrants, I, which may be educated with productivity y^h or non-educated with productivity, y^l where $y^h > y^l$. The workers search for jobs and firms search for workers and the labour force is normalised at one. For simplicity, we assume that firms may supply vacancies directed towards natives or immigrants. We then include the two features, which may differ for immigrants and natives, influencing their labour market performance differently and thereby their educational decision - namely negative attitudes towards immigrants and networking effects.⁴

Immigrants may be harmed by negative attitudes towards them at their workplace, resulting in separation from the job. The reason may be many-fold: negative attitudes against immigrants may imply that a firm needs to deal with unexpected issues in the firm or with clients, and/or the immigrant voluntary quits. Hence, immigrants face a random negative shock. We therefore assume that the separation rate, s_i^m , m = h, l, i = N, I, may be increasing in negative attitudes towards immigrants, a^m , m = h, l, giving a separation rate for immigrants of $s_I^m = s_N (1 + a^m)$ and a separation rate for natives of $s_N^h = s_N^l = s_N$. Negative attitudes may (among other things) themselves be influenced by the fraction of immigrants in an area, an issue we will return to below.

On the other hand, more immigrants may make it easier to obtain employment through networking. We here follow Fontaine (2007) by assuming that networking, $\lambda_i^m, i = N, I, m = h, l$ is increasing in the number of people of the same origin as the individual. We assume that networking for high productivity immigrants and natives is given by: $\lambda_I^h = t^h \frac{I(1-\hat{e_I})}{(N+I)(1-\hat{e_N})} = t^h I$ and $\lambda_N^h = t^h \frac{N(1-\hat{e_N})}{(N+I)(1-\hat{e_N})} = t^h N = t^h (1-I)$ as N + I = 1, and that networking for low productivity immigrants and natives is given by: $\lambda_I^l = t^l \frac{I\hat{e_I}}{(N+I)\hat{e_I}} = t^l I$ and $\lambda_N^l = t^l \frac{N\hat{e_N}}{(N+I)\hat{e_N}} = t^l N = t^l (1-I)$ as N + I = 1, where $0 < t^m < 1$, m = h, l, and $\hat{e_i}, i = N, I$ is the number of low-educated workers and $1 - \hat{e_i}, i = N, I$, is the number of educated workers. One may argue that a very large number of own ethnicity may not be as important as a relative smaller number, a potential network may grow so big that it is not really a usually network in terms of employment perspectives. This could be included in the analysis by changing the functional form of the network variable, so that it is increasing in the number the worker's own nationality but at a decreasing rate. We will return to

⁴In Larsen and Waisman 2012, it is assumed that it is not possible for firms to direct their search to either immigrants or natives. Therefore, any negative impact on immigrants, will through changed vacancy supply also affect natives. As the present paper also include educational choice and networking we, for simplicity, keep this additional channel out of the present set-up.

this issue below.

2.1 Matching

We assume that firms advertise V_i^m , i = N, I, m = h, l vacancies. Unemployment rates are given by u_i^m , i = N, I, m = h, l and there are L_i^m , i = N, I, m = h, l employees. Labour market tightness by the ethnic group is given by $\theta_i^m = (V_i^m + \lambda_i^m L_i^m)/u_i^m$, where the transition rate for an unemployed worker is given by $f(\theta_i^m)$ and for the firm it is $q(\theta_i^m)$. We assume that the worker transition rate is increasing in labour market tightness and at a decreasing rate, $\partial (f(\theta_i^m)) / \partial \theta_i^m > 0$, $\partial^2 (f(\theta_i^m)) / (\partial \theta_i^m)^2 < 0$ and the firm's transition rate is decreasing in labour market tightness at a decreasing rate, $\partial (q(\theta_i^m)) / \partial \theta_i^m < 0$ and $\partial^2 (q(\theta_i^m)) / (\partial \theta_i^m)^2 > 0$.

2.2 The Firm

The firm chooses the number of vacancies so as to maximise profits subject to negative attitudes towards immigrants and subject to networking effects. We assume, for simplicity, that firms can direct their search towards natives or immigrants and that each worker produces y^m , m = h, l and receives the bargained wage, $w_i^m, i = N, I, m = h, l$. We denote the discount rate by ρ and hiring costs are increasing in productivity, ky^m , m = h, l. A firm chooses the number of vacancies to advertise, $V_i^m, i = N, I, m = h, l$ and takes into account that its employees also produce applicants through networking. Each firm hiring natives or immigrants solves the following Bellman equation:

$$\rho \Pi_i(L_i^m) = max \left(y^m L_i^m - w_i^m - k y^m V_i^m + \Pi_i(L_i^m) \right), i = N, I, \ m = h, l, \qquad s.t.$$
(1)

$$\dot{L}_N^m = (\lambda_N^m L^m + V_N^m) q(\theta_N^m) - s_N L_N^m, \ m = h, l,$$
(2)

$$\dot{L}_{I}^{m} = (\lambda_{I}^{m} L^{m} + V_{I}^{m})q(\theta_{I}^{m}) - s_{I}^{m} L_{I}^{m}, \ m = h, l.$$
(3)

Firms choose their optimal number of employees, using two methods of search: advertising by the firm or networking, which happens at the rate $\lambda_i^m L_j^m f(\theta_i^m)$, i = N, I. Separation rates for immigrants, $s_I^m = s_N (1 + a^m) \ge s_N$, which depend on negative attitudes, a^m , m = h, l may differ for low productivity and high productivity workers. Hence, matches between immigrants and the firm may be dissolved more often than matches for natives and also may differ for highand low-educated workers. This implies that, for given networking, the expected profitability of a firm employing natives may be different than the expected profitability of employing a highand/or low-educated immigrant.

With identical firms, using equations (1)-(3) and Kuhn-Tucker conditions, we obtain the nontrivial solution in the steady state determining labour market tightness, θ_i^m , i = N, I, m = h, l:

$$\frac{ky^{m}}{q(\theta_{N}^{m})} = \frac{y^{m} - w_{N}^{m}}{\rho + s_{N} - \lambda_{N}^{m}q(\theta_{N}^{m})}, \frac{ky^{m}}{q(\theta_{I}^{m})} = \frac{y^{m} - w_{I}^{m}}{\rho + s_{N}(1 + a^{m}) - \lambda_{I}^{m}q(\theta_{I}^{m})}.$$
(4)

The partial equilibrium results are the following: more severe negative attitudes, a higher a^m , will tend to reduce labour market tightness and more networking, a higher λ_i^m , will raise labour market tightness for the firm hiring the specific type, for either immigrants or natives.

2.3 The Worker

Let U_i^m be the value of being an unemployed worker and $E_i^m, m = h, l, i = N, I$ be the value of being an employed worker. The values are determined by

$$\rho U_i^m = f(\theta_i^m) (E_i^m - U_i^m) - \Gamma(m) c(e_i), i = N, I, m = h, l,$$
(5)

$$\rho E_I^m = w_I^m + s_I^m (U_I^m - E_I^m) - \Gamma(m) c(e_i), m = h, l$$
(6)

$$\rho E_N^m = w_N^m + s_N (U_N^m - E_N^m) - \Gamma(m) c(e_i), m = h, l.$$
(7)

We assume that workers have different abilities, e_i , and therefore different costs of obtaining education, $c(e_i)$. The variable e_i is uniformly distributed, $e_i \in [0, 1]$ where educational costs are decreasing in ability at a decreasing rate, $c'(e_i) < 0$, $c''(e_i) > 0$. In order to guarantee a nontrivial solution where some, but not all, individuals choose to acquire education, the individual with the highest ability faces a very low cost of education, c(1) = 0, and the individual with the lowest ability level face very high costs of education, i.e. $\lim_{e_i\to 0} c(e_i) = \infty$. $\Gamma(m)$, m = h, l, is an indicator function, taking the value zero if the worker does not acquire education and one, if the worker acquires education. Hence, $\Gamma(h) = 1$ and $\Gamma(l) = 0$.⁵

⁵We assume that the educational cost is a cost to acquire and maintain education or skills. This is a simplifying

2.4 Wages

We assume that wages are determined by Nash bargaining and that the bargaining power is a half, so that $X_i^m = E_i^m - U_i^m$, i = N, I, m = h, l, where from equation (4) we have that $X_i^m = ky^m/(q(\theta_i^m)) = (y^m - w_i^m)/(\rho + s_i - \lambda_i^m q(\theta_i^m))$. We assume that the hiring cost parameter, k, is equal across firms, but that productivity and therefore actual hiring costs are higher for firms employing educated workers. This gives that $ky^m = X_i^m q(\theta_i^m)$ and thereby

$$X_{i}^{m} = \frac{y^{m} - w_{i}^{m} + \lambda_{i}^{m} k y^{m}}{(\rho + s_{i}^{m})}, \ m = h, l.$$
(8)

Subtracting equation (5) from equation (6) or (7) and then using $X_i^m = E_i^m - U_i^m$ and (8) give the wage equations

$$w_N^m = 0.5 \cdot y^m \left(1 + (\lambda_N^m + \theta_N^m) \, k \right), \tag{9}$$

$$w_I^m = 0.5 \cdot y^m \left(1 + (\lambda_I^m + \theta_I^m) \, k \right).$$
(10)

We note that wages are increasing in labour market tightness, networking and productivity. Substituting for wages into the equation determining labour market tightness, we obtain the equations for labour market tightness (8) as a function of parameter values and independently of productivity as hiring costs are a function of productivity:

$$k(\rho + s_I^m)2 = (1 - \theta_I^m k + \lambda_I^m k) q(\theta_I^m), \tag{11}$$

$$k(\rho + s_N)2 = (1 - \theta_N^m k + \lambda_N^m k) q(\theta_N^m).$$
(12)

We note the following. Regarding relative separation rates we have that, if the separation rate of both high and low productivity immigrants is greater than the separation rate of natives $s_I^m > s_N$, then the left hand side of (11) is larger than the left hand side of (12) tending to reduce labour market tightness for firms employing immigrants and thereby the transition rate for immigrants. Considering networking, labour market tightness is increasing in labour networking: $d\theta_i^m / (d\lambda_i^m) =$ $kq(\theta_i^m)/D_i^m > 0, i = N, I, m = h, l$, where $D_i^m = -((1 - \theta_i^m k + \lambda_i^m k) q'(\theta_i^m) - \theta_i^m kq(\theta_i^m)) > 0$. If networking is higher for immigrants than natives, $\lambda_I^m > \lambda_N^m$, this tends to increase θ_I^m relatively

assumption and is not important for the results. The assumption enables us to use a model without having workers continuously being born and dying. Such a model would deliver similar qualitative expressions.

to θ_N^m . However, if $s_I^m > s_N$ this tends to increase θ_N^m relatively to θ_I^m . Therefore, if there the separation rate is greater for immigrants than for natives, $s_I^m > s_N$, and networking for natives is greater than or equal to networking for immigrants, $\lambda_I^m \leq \lambda_N^m$, then the labour market tightness for natives is higher than labour market tightness for immigrants, $\theta_I^m < \theta_N^m$. If networking among immigrants is greater than networking among natives, $\lambda_I^m > \lambda_N^m$, and $s_I^m > s_N$ then the relative size of labour market tightness is ambiguous.

For the rest of the theoretical analysis we assume that educated and uneducated workers face the same networking effect, hence $\lambda_i^h = \lambda_i^l = \lambda_i$, i = N, I. With this assumption we obtain that labour market tightness is the same for high and low-educated natives, $\theta_N^h = \theta_N^l = \theta_N$ whereas we have two scenarios for immigrants. In the first case, negative attitudes is present for both high and low productivity workers and hence $s_I^h = s_I^l = s_I$ resulting in $\theta_I^h = \theta_I^l = \theta_I$. In the second case, negative attitudes exist for educated workers only and hence $s_N = s_I^h < s_I^l$ resulting in $\theta_I^h > \theta_I^l$. This assumption allows us to consider the impact of a change in attitudes and immigration on labour market tightness, education and unemployment, without making any assumptions about the relative importance of networking for educated or uneducated workers. We will in Section 6 below discuss how the results are modified in the case of heterogeneous networking effects. We have the following result.

Result: In case 1, where negative attitudes are present in both the high and low productivity sector, $a^h = a^l > 0$, and networking of natives is larger than or equal to networking of immigrants, $\lambda_N \ge \lambda_I$ then labour market tightness for natives is higher than labour market tightness for immigrants, $\theta_N > \theta_I$, and natives' wages are thus higher than immigrants' wages, $w_N^m > w_I^m$.

In case 2, when negative attitudes are present in the low productivity sector only, $a^l > a^h = 0$, and networking of natives is larger than or equal to networking of immigrants, $\lambda_N \ge \lambda_I$, then: (i) for low productivity workers, labour market tightness for natives is higher than labour market tightness facing immigrants, $\theta_N^l > \theta_I^l$, and low productivity natives' wages are thus higher than low productivity immigrants' wages, $w_N^l > w_I^l$, and (ii) for high productivity workers, labour market tightness and wages of natives and immigrants are equal, $\theta_N^h = \theta_I^h$ and $w_N^h = w_I^h$.

When networking of natives is less than networking of immigrants, $\lambda_N < \lambda_I$, then the relative sizes of labour market tightness, θ_N^m and θ_I^m , and wages, w_N^m and w_I^m , for natives and immigrants are indeterminate.

Note that given the assumption above that $\lambda_I = tI$ and $\lambda_N = t(1-I)$, where 0 < t < 1 we

have that $\lambda_N > \lambda_I$ as long as I < 0.5, which is the most realistic case. In case the networking function takes another form, namely if it is increasing in the number of the worker's own ethnicity but at a decreasing rate, for example, $\lambda_I = tI^{1/2}$ and $\lambda_N = t(1-I)^{1/2}$, we will still have that $\lambda_N > \lambda_I$ as long as I < 0.5, but the impact of an additional labour force participant is larger for immigrants than natives as long as immigrants are the minority.

2.5 Education

When individuals decide on whether to educate or not, they compare the value of acquiring education to the value of remaining uneducated. That is, at each point in time, as an unemployed worker, they compare the value of being unemployed as a educated worker to the value of being unemployed as an uneducated worker. Workers with high educational costs find it too costly to obtain education, whereas high ability workers and low educational costs individuals find it more than worthwhile to do so. The marginal worker has the ability level, \hat{e}_i , i = N, I, which makes the worker just indifferent between acquiring education or remaining uneducated. For simplicity, we assume that natives and immigrants are identical with respect to the distribution of educational costs. We write the condition determining the educational costs of the marginal worker as

$$\rho U_i^h\left(\hat{e}_i\right) = \rho U_i^l, \ i = N, I.$$
(13)

The higher \hat{e}_i is, the higher is the ability level of the marginal worker acquiring education. Hence, fewer workers acquire education, and a smaller fraction of the workers will be educated. Use equations (5)-(7) and (13), the bargaining condition together with the free entry condition, to obtain the following simplified condition in the first case where $a^h = a^l$ for immigrants and a = 0for natives:

$$\left(y^{h} - y^{l}\right)\theta_{i}k = c\left(\hat{e}_{i}\right), i = N, I.$$
(14)

Equation (14) gives $\hat{e}_i, i = N, I$ as a function of the endogenous labour market tightness variables, $\theta_i, i = N, I$. The higher the productivity difference is, the higher are wage differences, and then the more people will acquire higher education. For equal networking rate, labour market tightness facing natives is higher than labour market tightness facing immigrants, which results in that natives acquire more education than immigrants, that is, $\hat{e}_I > \hat{e}_N$.

In the second case, the result changes for immigrants whereas the natives' educational decision

is still given by equation (14), i.e. when $a^{h} = 0$ and $a^{l} > 0$ then we obtain:

$$\left(y^{h}\theta_{I}^{h}-y^{l}\theta_{I}^{l}\right)k=c\left(\hat{e_{I}}\right).$$
(15)

In this case, with equal networking rate for all workers, we now obtain that $\hat{e}_N > \hat{e}_I$ as low productivity immigrants are worse of than natives in terms of a lower transition rate into a job, $\theta_I^l < \theta_N^l$ and lower wages and high productivity immigrants have the same wages and employment probability as natives, $\theta_I^h = \theta_N^h$. Hence, due to that the uneducated immigrants are relative worse of than natives, immigrants in this case experience stronger incentives for acquiring education that natives. This is summarised in the following result.

Result: In case 1, where negative attitudes are present in both the high and low productivity sector, $a^h = a^l > 0$, and networking of natives is equal to networking of immigrants, $\lambda_N = \lambda_I$, natives acquire more education than immigrants, that is, $\hat{e}_I > \hat{e}_N$.

In case 2, where negative attitudes are present in the low productivity sector only, $a^l > a^h = 0$, and networking of natives is equal to networking of immigrants, $\lambda_N = \lambda_I$, then immigrants acquire more education than natives $\hat{e}_I < \hat{e}_N$.

Notice here the significance of the networking assumption. In section 6 below we discuss the impact of including heterogeneity and we discussed the nonproportionality of the networking function above.

2.6 Unemployment

In equilibrium, inflows are equal to outflows. The equilibrium flows characterising the labour market for workers are then, $f(\theta_i^m) \mu_i^m = s_i^m n_i^m$, i = N, I, m = h, l, and $n_i^h + \mu_i^h = (1 - \hat{e}_i)i$, $i = N, I, n_i^l + \mu_i^l = \hat{e}_i i$, i = N, I, where employment is n_i^m , i = N, I, m = h, l and unemployment is μ_i^m , i = N, I, m = h, l. The labour force is normalised to one, N + I = 1, giving the following expression for natives' unemployment rates: u_N^m , m = h, l: $u_N^h = u_N^l = u_N = s_N / (f(\theta_N) + s_N)$, as $\theta_N^h = \theta_N^l$. In the first case, we have the separation rate for high and low productivity immigrants is equal, $s_I^h = s_I^l$, and hence labour market tightness is equal, $\theta_i^h = \theta_i^l$, then unemployment rates are the following:

$$u_i^h = u_i^l = u_i = \frac{s_i}{f(\theta_i) + s_i}, \ i = N, I.$$
 (16)

Unemployment rates for educated workers are equal to unemployment rates of uneducated workers. This results stems from the assumption that hiring costs are proportional to productivity.

In the second case, where the separation rate of low productivity immigrants is higher than that of high productivity immigrants, $s_I^h < s_I^l$ as $a^h = 0$ and $a^l > 0$, then $s_I^l = s_N(1 + a^l) > s_I^h = s_N$ and therefore $f\left(\theta_I^l\right) < f\left(\theta_I^h\right) = f\left(\theta_N\right)$ which results in the following unemployment rates

$$u_N = u_I^h < u_I^l = \frac{s_I}{f(\theta_I^l) + s_I}, \ i = N, I.$$
(17)

The result is the following.

Result: In case 1, where negative attitudes are present in both the high and low productivity sector, $a^h = a^l > 0$, and networking of natives is larger than or equal to networking of immigrants, $\lambda_N \ge \lambda_I$, the unemployment rate of natives is smaller than the unemployment rate of immigrants, $u_N < u_I$.

In case 2, where negative attitudes are present in the low productivity sector only, $a^l > a^h = 0$, and networking of natives is larger than or equal to networking of immigrants, $\lambda_N \ge \lambda_I$, then the unemployment rate of low productivity immigrants is larger than that of high productivity immigrants and the unemployment rate of high productivity immigrants is greater than or equal to that of natives, $u_I^l > u_I^h \ge u_N$.

When $\lambda_N < \lambda_I$ then the relative sizes of the unemployment rates facing natives and immigrants, u_N and u_I are indeterminate.

3 Negative Attitudes

In this section, we examine what happens to labour market tightness, wages, education and unemployment when immigrants face more severe negative attitudes. For simplicity, we consider the case where $\lambda_N = \lambda_I$. The impact on labour market tightness, wages and unemployment as well as education will differ dependent on whether negative attitudes towards immigrants exists in both sectors or in the low productivity sector only. We have the following proposition.

Proposition: In case 1 where negative attitudes are present for both high and low productivity workers, $a^h = a^l > 0$, then when negative attitudes increase, immigrants' wages fall and their unemployment rates increase. Education of immigrants is also reduced. In case 2, when only low productivity workers face negative attitudes, $a^l > a^h = 0$, then wages for low productivity immigrants fall and their unemployment rate increases, whereas high productivity immigrants are not affected, increasing education for immigrants. There is no impact on natives.

Proof: see Appendix A.

In the first case, where $a^h = a^l = a$, an increase in negative attitudes increases the separation of immigrants and therefore makes it less profitable to open a vacancy. The reduction in labour market tightness for immigrants reduces their bargaining power and thereby their wages. Immigrants' transition rate falls which together with their higher separation rate increases their unemployment rate. Concerning educational choice, the impact depends on the impact on employment perspectives for high productivity workers relatively to the impact on low productivity workers. The reduced employment perspectives, through lower employment chances and lower wages, affect both high productivity and low productivity workers. However, due to higher productivity, the reduction in wages is going to be larger for high productivity workers than for low productivity workers and therefore the incentives to acquire education fall. The result is that fewer immigrants acquire education. As negative attitudes have no impact on the separation rate of natives, they are not affected.

For the second case, that is, where $a^l > a^h = 0$, an increase in negative attitudes only increases the separation of low productivity workers and only for the low productivity firms hiring immigrants, there is a reduction in the profitability of opening a vacancy. The resulting reduced labour market tightness for low productivity firms hiring immigrants increases uneducated immigrants' unemployment rate. High productivity immigrants are not affected as their separation rate is not affected. When we turn to educational choice, the result changes compared to in case 1. The employment perspectives for high productivity workers are not affected and as the employment perspectives of low productivity workers worsens, and the incentives to acquire education increase. In this case, we therefore obtain the opposite result compared to in case 1, namely that more immigrants acquire education. Again, as negative attitudes have no impact on the separation rate of natives, they are not affected.

As a caveat, notice, that we could allow for the possibility that negative attitudes affect the value of being unemployed also directly, and not only indirectly through wages and employment chances. In this case, the impact on unemployment will not be affected, but if, in case 1, negative attitudes directly diminish the value of being unemployment equally for uneducated and educated workers, then there is no impact on education. In case 2, the direct impact will also, as the indirect through employment and wages, tend to increase education.

4 Immigration

In this section, we examine the impact on labour market tightness, wages, education and unemployment from more immigration. Notice that $\lambda_I = tI$ and $\lambda_N = tN = t(1 - I)$. The impact on labour market tightness, wages and unemployment as well as education will differ dependent on whether negative attitudes towards immigrants exists in both sectors or in the low productivity sector only. We have the following proposition.

Proposition: When the fraction of immigrants increases, the unemployment rate of immigrants falls and their wages increase. The improved labour market prospects of immigrants raise their level of education in both case 1 and case 2 and the opposite holds for natives.

Proof: see Appendix A.

More immigrants will induce the fraction of immigrants to increase, improving networking and thus labour market tightness for firms hiring immigrants and therefore immigrants' transition rate. Similarly, networking among natives fall, and thereby labour market tightness for natives falls. As networking both directly and indirectly has a positive impact on immigrants' wages, their wages increase whereas natives' wages fall. Furthermore, the increase in immigrant's transition rate reduces their unemployment rate and the corresponding reduction in natives' transition rate raise their unemployment rate. Finally, concerning education for immigrants, improved labour market conditions due to more networking are better for high productivity workers than low productivity workers, wherefore education increases.

As an illustration, consider the situation where a = 0 and hence $s_I = s_N$ and initially N = I. In this case, labour market tightness facing immigrants is equal to labour market tightness facing natives. The fraction of educated immigrants and natives are also identical, $\hat{e}_I = \hat{e}_N$ and thereby $c'(\hat{e}_I) = c'(\hat{e}_N)$. The increase in educated natives is therefore equal to the fall in the fraction of educated immigrants. However, a more realistic setup is where N > I so that $\theta_N > \theta_I$ and thus $\hat{e}_I > \hat{e}_N$ (the fraction of natives acquiring skills is higher than the fraction of immigrants acquiring skills). In this case, $c(\hat{e}_I) < c(\hat{e}_N)$, and $|c'(\hat{e}_I)| > c'(\hat{e}_N)$, the impact through the lower educational costs will increase the impact on education. However, substituting from equation (11) and (12) we obtain that the positive impact of networking is smaller for immigrants than the negative impact from networking for the natives, $|d\theta_I/dI| < d\theta_N/dI$. Hence, given N > I initially, the impact from an increase in the number of immigrants on their educational level may be smaller or larger than the negative impact on the educational level facing natives.

5 Immigration and Negative Attitudes

In this section we expand the model by allowing for the possibility that a higher fraction of immigrants aggravates negative attitudes, giving for case 1, $s_I^h = s_I^l = s_N (1 + a(I))$ and for case 2, $s_I^h = s_N$ and $s_I^l = s_N (1 + a(I))$. The idea is that more immigrants around increases the possibility of a multi-ethnic society, which for some people is a negative development. As results now in general becomes ambiguous we consider the special case where the matching function takes the form $X_I^m = \sqrt{v_I^m u_I^m}$ and that a'(I) = 1. The impact on natives is identical to the impacts above.

Proposition: Natives are affected as above. For immigrants we have the following. In the first case, differentiating equations (11), including the matching function, $X_I^m = \sqrt{v_I^m u_I^m}$ with respect to I where now a(I) we obtain $\frac{d\theta_I}{dI}|_{a(I)} = \frac{-k(s_N a'(I)\sqrt{\theta_I} - t)}{(k(\rho + s_I^m)/\sqrt{\theta_I} + k)}$. Substituting for the solution for labour market tightness we obtain the condition for a'(I) = 1 that $\frac{d\theta_I}{dI}|_{a(I)} \leq 0 \Leftrightarrow z \geq I$, where $z = \frac{s_N(2\rho tk + s_N(1-2tk)) - t^2k}{s_N^2 tk}$. This implies that for case 1 we obtain $\frac{dw_I^m}{dI}|_{a(I)} \leq 0$ and $\frac{d\hat{e}_I}{dI}|_{a(I)} \geq 0$ for $z \geq I$. In the second case, we obtain where $da^l/dI = 1$ that $\frac{d\theta_I^l}{dI}|_{a(I)} = \frac{-k(s_N\sqrt{\theta_I} - t)}{k(\rho + s_I^l)/\sqrt{\theta_I^l} + k}$ and $\frac{d\theta_I^h}{dI}|_{a^l(I)} = \frac{tk}{k(\rho + s_I^l)/\sqrt{\theta_I^l} + k} > 0$. For wages we have $\frac{dw_I^l}{dI}|_{a^l(I)} \leq 0$ $z \geq I$, $\frac{dw_I^h}{dI}|_{a^l(I)} > 0$, and education increases, $\frac{d\hat{e}_I}{dI}|_{a^l(I)} < 0$ and unemployment increases if $\frac{du_I^l}{dI}|_{a(I)} > 0$ for $z \geq I$, and $\frac{du_I^h}{dI}|_{a(I)} < 0$.

The impact of immigration on labour market performance for immigrants now becomes ambiguous. The reason is that more immigration improves networking and thereby employment chances and wages, but at the same time, negative attitudes may become more severe which reduce labour market tightness again. In the first case, where $a^h = a^l = a(I) > 0$, the positive impact through networking on labour market tightness is more important than the negative impact through increased negative attitudes if the fraction of immigrants is sufficiently high. The condition for a positive sign for labour market tightness is dependent on the separation rate and the networking effect, so that in this case, the separation rate has to be low, low s, relatively to the networking effect, high t. In the second case, $a^h = 0$, $a^l = a(I) > 0$, high productivity workers are not affected and low productivity workers are effected as in case 1, implying that education unambiguously increases, as the relative gain of acquiring education increases.

6 Heterogeneous Networking Effects

In this section we allow the networking effects to differ for uneducated and educated workers as well as for natives and immigrants.⁶ First, we consider the case where $s_I^m > s_N$, m = h, l, which results in the left hand side of (11) being larger than the left hand side of (12) and therefore tends to reduce labour market tightness for firms employing immigrants and thereby the transition rate for immigrants. Therefore, when immigrants face more networking than natives, $\lambda_I^m > \lambda_N^m$, we cannot determine the relative size of θ_I^m and θ_N^m as this networking effect would tend to increase labour market tightness for immigrants relative to natives. For the rest of this section we therefore consider the case where networking for immigrants is lower than networking for natives, $\lambda_N^m > \lambda_I^m$, m = h, l.

Regarding education, we now need to consider the more general equation, allowing labour market tightness to differ both for educated and uneducated natives as well as immigrants :

$$\left(y^{h}\theta_{i}^{h}-y^{l}\theta_{i}^{l}\right)k=c\left(\hat{e}_{i}\right),\ i=N,I.$$
(18)

To begin with, we assume that networking is the same for educated and uneducated immigrants and consider first the case where $\lambda_N^h > \lambda_N^l$. In this case, educated natives are more efficient using the network and we obtain that $\theta_N^h > \theta_N^l > \theta_I^h = \theta_I^l = \theta_I$, resulting in higher wages for educated native workers than uneducated native workers, (see equation (9)), who then in turn, as before, receive higher wages than immigrants (see equation (9) relative to (10) inserting for labour market tightness and networking). Furthermore, considering education, using equation (18) we obtain that a higher fraction of natives than immigrants acquire education, $\hat{e}_I > \hat{e}_N$, as $y^h \theta_I^h - y^l \theta_I^l < y^h \theta_N^h - y^l \theta_N^l$ if and only if $y^l \left(\theta_N^l - \theta_I^l\right) < y^h \left(\theta_N^h - \theta_I^h\right)$ as there is a larger gain involved for natives than immigrants acquiring education.

If instead, uneducated native workers are better at networking than educated natives workers,

⁶As we do not allow networking effects to depend on the number of each educational type (as then labour market tightness would be a function of \hat{e}_i , i = N, I) then this corresponds to assuming that t^m is different for the two different educational types.

 $\lambda_N^h < \lambda_N^l$, then there are relative more vacancies supplied towards uneducated native workers than educated native workers and hence, $\theta_N^l > \theta_N^h > \theta_I^h = \theta_I^l = \theta_I$. In this case, the order of educated native wages and uneducated native wages become ambiguous as the higher productivity of educated natives will tend to raise w_N^h relative to w_N^l whereas the higher networking effect for uneducated natives both directly and indirectly through a higher labour market tightness will tend to increase w_N^l relative to w_N^h . In terms of education, we cannot tell whether $\hat{e}_I > \hat{e}_N$ or $\hat{e}_I \leq \hat{e}_N$ as $y^l < y^h$ but $\theta_N^l - \theta_I^l > \theta_N^h - \theta_I^h$ as $\theta_N^l > \theta_N^h$ and $\theta_I^h = \theta_I^l = \theta_I$. This is the case as good networking for uneducated natives means that being uneducated tends to be more attractive for natives, but on the other hand, as networking is still better for educated natives than immigrants, this will tend to increase the number of educated natives.

Next, we allow the networking variable to vary also for immigrants. First, we consider the case where uneducated immigrants are more efficient using their network, that is, $\lambda_I^l > \lambda_I^h$. When $\lambda_N^h = \lambda_N^l$ then $\theta_N^h = \theta_N^l > \theta_I^l > \theta_I^h$, implying that natives are better paid but we cannot tell whether the uneducated immigrants or the educated immigrants earn the most, as the higher networking and labour market tightness for uneducated workers compared to educated workers tends to raise wages for this group but the latter group of immigrants has a higher productivity than the former. In this case, $\hat{e}_I > \hat{e}_N$ as $\theta_N^h = \theta_N^l$ and $\theta_I^l > \theta_I^h$ implying that uneducated immigrants are relative better off than educated immigrants in terms of transition into work. When instead $\lambda_N^h > \lambda_N^l$ then $\theta_N^h > \theta_N^l > \theta_I^l > \theta_I^h$, the conclusion concerning relative wages for natives is as above when we considered the same relative networking effects for natives and indeterminate for immigrants. We also have that $\hat{e}_I > \hat{e}_N$ as the uneducated immigrants are relative better off than educated immigrants in terms of transition into work and the reverse holds for natives, making the impact even stronger. Finally, when $\lambda_N^l > \lambda_N^h$ then $\theta_N^l > \theta_N^h > \theta_I^l > \theta_I^h$. Here the relative sizes of low and educated wages are ambiguous for both natives and immigrants and we cannot determine the relative skill levels for natives and immigrants as we do not know the relative size of $\theta_N^l - \theta_N^h$ and $\theta_I^l - \theta_I^h$.

When there are more educated immigrants than uneducated immigrants, that is, $\lambda_I^h > \lambda_I^l$, then $\theta_I^h > \theta_I^l$, and relative labour market tightness for natives and thereby wages will vary as above dependent on the relative size of λ_N^h and λ_N^l . Here, we know that $w_I^h > w_I^l$ as networking, labour market tightness and productivity move in the same direction. We cannot determine whether a higher fraction of immigrants or natives acquire education as we do not know the relative size of $\theta_N^l - \theta_N^h \text{ and } \theta_I^l - \theta_I^h.$

In the second case, when negative attitudes exist for low productivity workers only, that is when, $s_N = s_I^h < s_I^l$, then if networking is the same for uneducated and educated immigrants, but higher for educated natives than uneducated natives, i.e. $\lambda_N^h > \lambda_N^l > \lambda_I^h = \lambda_I^l$, then we obtain $\theta_N^h > \theta_N^l > \theta_I^h > \theta_I^l$ and the wage order is similar and the order of unemployment rates is $u_N^h < u_N^l < u_I^h < u_I^l$. We cannot determine the relative size of \hat{e}_I and \hat{e}_N unless we know the relative size of $\theta_N^h - \theta_I^h$ and $\theta_N^l - \theta_I^l$. If instead, uneducated native workers have a better network than educated natives workers, $\lambda_N^h < \lambda_N^l$, then there are relative more vacancies supplied towards uneducated native workers than educated native workers and hence, $\theta_N^l > \theta_N^h > \theta_I^h > \theta_I^l$. As above with equal separation rates for immigrants, we cannot determine the relative size of immigrant wages and education, as $y^l < y^h$ but $\theta_N^l - \theta_I^l > \theta_N^h - \theta_I^h$.

When uneducated immigrants have a better network than educated, that is, $\lambda_I^l > \lambda_I^h$, then when $\lambda_N^h = \lambda_N^l$ we obtain that $\theta_N^h = \theta_N^l$ but we cannot determine the relative size of θ_I^l and θ_I^h , something which still holds when $\lambda_N^h > \lambda_N^l$ where for natives we now obtain $\theta_N^h > \theta_N^l$ and we can still not determine the relative size of labour market tightness for immigrants. Finally, when $\lambda_N^l > \lambda_N^h$ then $\theta_N^l > \theta_N^h$ and the relative size of θ_I^h and θ_I^l remain ambiguous. Relative wages follow the order of labour market tightness for natives and are indeterminate for immigrants. We cannot determine whether $\hat{e}_I > \hat{e}_N$ or $\hat{e}_I \leq \hat{e}_N$ as we do not know the relative size of θ_I^h and θ_I^l .

When educated immigrants are more efficient using their network than uneducated, that is, $\lambda_I^h > \lambda_I^l$ then we obtain the same relative labour market tightness for educated and uneducated natives as above as well as wages. In none of these situations, we can determine the relative skill levels for natives and immigrants as we do not know the relative size of $\theta_N^l - \theta_N^h$ and $\theta_I^l - \theta_I^h$. See Table A1 in Appendix B for a detailed summary of the results.

7 Empirical Analysis

In this section, we test the model predictions regarding the impact of networking and negative attitudes on education for 1^{st} and 2^{nd} generation immigrants.⁷

⁷In what follows, we use the terms 1st and 2nd generation immigrants and immigrants interchangably.

7.1 Data

We consider data for Denmark in 2002 and examine both the variation between municipalities as well as educational decisions at the individual level. Denmark is chosen for two reasons. Firstly, there is detailed and rich data available both at the municipality and individual level (where we have data for the whole population) which enables the impact of negative attitudes on education to be examined both at the macro and micro level. Secondly, the educational structure of Denmark allows us to examine an individual's decision of whether to attend high school, which is noncompulsory, after the completion of compulsory education at a young age. In Denmark, students' first year in high school will be when they are 15-16 years old and only the first 9 years of schooling are obligatory.⁸ As this decision takes place at such a young age, mobility of the student is of little concern, which is discussed in further detail below. While high school is optional, most of the students beginning high school will graduate with a high school degree. Moreover, in this paper we aim to evaluate whether the potential impact of negative attitudes and immigration on relative labour market performance for educated and uneducated workers influence the decision to begin high school.

In 2002, there are 275 municipalities in Denmark. 2002 is chosen as there is a general election in 2001 and it is well prior to 2007 when the 275 municipalities are merged into 98 municipalities. The fundamental idea behind using macro data is to potentially show some correlation between the general prevalence of negative attitudes in a municipality and the expected return to education, through employment perspectives and the fraction of immigrants attending high school. The impact may thus be both direct and indirect, and in this sense we may capture something different than if we were to only consider micro data. Furthermore, the macroeconomic nature of the theoretical model, makes the macro-econometric analysis a natural starting point.

The fundamental idea behind using micro data is we are able to precisely take into account many individual and household factors which may impact an individual's educational decision. In order to examine the relationship between immigrant high school attendance and negative attitudes towards immigrants and networks at the individual level, we use detailed Danish Register Data made available by Statistics Denmark. Danish Register Data is a database containing detailed information on every resident of Denmark from 1980-present. The data is interlinked across various government and administrative sources by an anonymous personal identification number, so indi-

⁸Only recently a grade zero has become obligatory.

viduals are also observed overtime. The data used contains similar, but more detailed information compared to the macro-econometric data and is composed of: education history (information such as where an individual attends school and what qualification they are studying towards and have already achieved); demographic information such as gender, age, and municipality of residence; immigration history (including an individual's nationality, exact date of immigration, and whether an individual is a 1st or 2nd generation immigrant; and household characteristics such as family composition and parental information. As individuals are linked to their parents, it is possible to include factors such as parental education, employment history, and marital status, all of which will likely affect an immigrant's high school decision.

Our main dependent variable is an immigrant's high school attendance in 2002. In the macroeconometric analysis, this is measured by the fraction of immigrants attending high school in the municipality while in the micro-econometric section this is measured by whether or not an individual is attending high school. We measure negative attitudes using the fraction of votes for two political parties, Fremskridtspartiet and Dansk Folkeparti, due to their emphasis on reducing immigration.⁹ One potential concern in using voting data is that immigrants may not vote for parties who emphasise reducing immigration. This would be problematic for our measure of negative attitudes as municipalities with higher concentrations of immigrants could actually have few votes for Fremskridtspartiet and Dansk Folkeparti. However, the voting behaviour of immigrants is unable to influence our measure of negative attitudes as only natives are permitted to vote in general elections. As such, the voting data is a good measure of the negative attitudes of natives against immigrants. Furthermore, the 2001 general election campaign had a huge emphasis of immigration, implying a large increase in the votes for Fremskridtspartiet and Dansk Folkeparti.

For the network variable we consider two possible variables: (1) the fraction of immigrants residing in a municipality and (2) the fraction of immigrants residing in a municipality who are educated and/or employed. The first of these measures captures whether the quantity of individuals

⁹Fremskridtpartiet's webpage: http://www.fremskridtspartiet.dk/page9.html (all in Danish: main idea: immigrants may stay for shorter or longer periods).

Dansk Folkeparti's webpage (http://www.danskfolkeparti.dk/ The_Party_Program_of_the_Danish_Peoples_Party): Denmark is not an immigrant-country and never has been. Thus we will not accept transformation to a multiethnic society.

Denmark belongs to the Danes and its citizens must be able to live in a secure community founded on the rule of law, which develops along the lines of Danish culture.

It ought to be possible to absorb foreigners into Danish society provided however, that this does not put security and democratic government at risk. To a limited extent and according to special rules and in conformity with the stipulations of the Constitution, foreign nationals should be able to obtain Danish citizenship.

Other Danish parties may also be interested in limiting immigration but not to such an extent that it is on their official webpage.

matter for networking while the second captures whether the quality of individuals matter. In the macro-econometric analysis, this second measure is captured by the fraction of Western/non-Western immigrants in employment. Throughout the micro-econometric analysis, we construct exact sums of immigrants from a certain nation residing in every municipality. By interacting these sums with an immigrant's own nationality, we recreate an individual's potential network based on the fraction of the population living in his municipality from his homeland (excluding the immigrant him or herself). Using education and employment data, we are able to construct the education and employment levels of immigrants by nationality for each municipality. From theory it is not evident which measure to choose and while exposure to educated or employed immigrants may have a far greater impact on an individual's educational decision. Even though the employment and educated fractions are considered closest to our theoretical model presented above, we include both networking measures for completeness. When the regressions are run for natives, we translate these fractions into the fraction of natives.

Figures 1 and 2 show the dispersion of the fraction of votes for Fremskridtspartiet and Dansk Folkeparti and the fraction of 1st and 2nd generation immigrants residing in the municipality. Votes for the two parties are relatively scattered across Denmark, with a high concentration of municipalities with a large fraction of votes for both parties near the Danish/German border. Immigrants are also scattered across Denmark, with the exceptions that they tend to reside closer to large cities (Aarhus, Odense, and the greater Copenhagen area) as well as near the Danish/German border.

7.2 Identification

In order to disregard mobility issues, we examine the high school decision as a function of immigration and attitudes as well as other explanatory variables. Our focus on the high school decision is important for our identification strategy as the parents decide where to live and young immigrants and natives then decide whether to continue to high school. As they are young when this decision is made, they will usually stay at home when attending high school. As such, the validity of our empirical analysis relies on two assumptions. Firstly, that young children, rather than their parents, are making the decision of whether to attend high school or not. As high school starts later in life than, for example, in the USA, students will be more mature and involved in their own education.

Figure 1: Municipal Fractions of Votes for Fremskridtpartiet and Dansk Folkeparti in 2001 General Election



This is especially true in our case as in Denmark, the young person, rather than parent, is heavily involved with the school in the education process at the time of deciding to attend high school - it is, for example the young person having the code needed in order to access the online application form and fill in the application form and not the parent. Secondly we assume, that parents, both in their emigration and subsequently, do not selectively locate into municipalities. These imply that the household placement is exogenous for the young person making the educational decision.

There are a number of potential threats to these assumptions, particularly the second assumption. Firstly, parents may move in order to place their child in a specific municipality before the high school decision is made. We examine this possibility by, as a robustness check for our microeconometric analysis, only considering the high school decision for families who have not moved recently. Secondly, high ability and resourceful parents may either move or emigrate to better municipalities, and these parents may also have high ability children. We overcome this problem by controlling for parental characteristics together with a host of municipality level controls. These high ability parents may also move in response to negative attitudes. Lastly, emigration as well as moving away from or towards certain municipalities could be driven by unobserved parental characteristics, and these unobservable characteristics could be correlated with our negative attitudes and networking measures as well as the child's educational decision. In the micro-econometric



Figure 2: Municipal Fractions of Immigrants Residing in Municipality

analysis, the last two concerns is partially addressed by the robustness check excluding those who have recently moved. However, this does not completely solve the unobservables problem, as the decision of staying and locating initially is also endogenous. As such, we investigate the importance of unobservables and examine the sensitivity of our results to the omitted variable bias problem using a procedure developed in Oster (2015) which tests the proportional importance of unobservables in relation to observables. In the macro-econometric analysis below, we address these concerns by exploiting the panel nature of our data and using a municipal fixed effects specification. We employ data from 1999 in order to account for any unobservable time-invariant factors at the municipal level which affect our measure of high school attendance. Additionally, a dynamic panel data specification allows us to comment on the theoretical predictions of Sections 3 and 4, which examine changes in the values of negative attitudes and immigrants.

7.3 Macro-econometric Model and Analysis

Table 1 presents descriptive statistics of the municipal data. While the total number of municipalities in Denmark during the time period is 275, we drop 27 municipalities as there were no 16 year old immigrants residing in them in either 2002 or 1999.¹⁰ This leaves us with 248 municipalities

¹⁰The municipalities we drop are: Hvalsø, Fuglebjerg, Gørlev, Høng, Holeby, Højreby, Ravnsborg, Rudbjerg, Egebjerg, Haarby, Langeskov, Marstal, Sydlangeland, Tommerup, Tranekær, Aarup, Gram, Blåvandshug, Hinnerup, Nørhald, Samsø, Fjends, Sallingsund, Læsø, Sejlflod, Sindal, Åbybro.
for our regressions. From Table 1, it is seen that the average share of 16 year old immigrants in high school is about 45.8% while it is larger for natives, 74%.

Table 1: Summary Statistics at the Municipal Level										
	(1) Mean	(2) Median	(3) Std Dev	(4)Min	(5) Max					
Immigrants Aged 16 in Any HS $(\%)$	45.8%	50.0%	27.0	0.0%	100%					
Natives Aged 16 in Any HS $(\%)$	74.0%	74.2%	5.9	54.9%	88.2%					
Votes Both Parties (%)	13.2%	13.0%	2.3	7.1%	21.0%					
Network for Immigrants (% of immigrants)	5.0%	3.9%	3.3	2.0%	25.2%					
'Network' for Natives (% of natives)	95.0%	96.1%	3.3	74.8%	98.0%					
Western Immigrants Employed $(\%)$	95.0%	95.5%	3.2	81.8%	100.0%					
Non Western Immigrants Employed $(\%)$	89.1%	90.9%	9.0	27.3%	100.0%					
Natives Employed (%)	96.5%	96.8%	1.3	90.4%	98.5%					
Gross Income Per Capita/100000	1.732	1.677	0.195	1.448	2.771					
Population Density/1000	0.285	0.073	0.846	0.021	10.413					
LF Short Tertiary Education $(\%)$	40.1%	40.2%	3.4	27.1%	50.3%					
LF Medium Tertiary Education (%)	10.7%	10.2%	2.8	6.3%	24.2%					
LF Long Tertiary Education $(\%)$	3.2%	2.3%	2.9	0.7%	20.4%					
Observations			248							

Our negative attitudes measure has an average value of 13.2%. The first network measure for immigrants (fraction of immigrants) is on average 5% and the two elements of the second network measure for immigrants (fraction of employed immigrants from Western and non-Western countries respectively) are on average 95% and 89.1%. It is not surprising that the employment ratio for Western immigrants is larger than for non-Western immigrants. The average (of average) gross income level per capita is 173200 DKK and the average population density is 285 people per square kilometre. Population density is included to account for the degree of urbanisation of a municipality. The average share of the population with a short, medium and long education are respectively 40.1%, 10.7% and 3.2%. When comparing the mean values and the median values of all variables we see that in some cases - especially for variables involving immigrants - the distributions are a bit skewed.

As stated above, we want to disregard mobility taken by the individual due to different labour market conditions or attitudes. We therefore estimate the high school decision at the municipal level using a fixed effects model:

$$(1 - \hat{e}_{rt}) = \beta_0 + \beta_1 a_{rt} + \beta_2 \lambda_{rt} + \sum_{\eta} \beta_{rt\eta} Controls_{rt\eta} + \gamma_t + c_r + \varepsilon_r,$$
(19)
$$r = 1, ..., 266 \ t = 1999, 2002.$$

The left hand side variable, $(1 - \hat{e}_{rt})$, is either the fraction of young 16 years old immigrants attending high school in year t (our main group of interest) or, for control purposes, the same fraction for natives in municipality r. Ideally, we would expect both the attitude and the network variables to be significant for immigrants while being insignificant for natives. We examine whether negative attitudes, a_{rt} , and networking, λ_{rt} , have any impact on the fraction of young immigrants (16 year olds) attending high school. For networking, we consider two potential measures for λ_{rt} to assess whether the quantity of individuals or the quality of individuals matters: (1) the fraction of immigrants residing in a municipality (quantity) and (2) the fraction of immigrants residing in a municipality who are employed (quality). The inclusion of both year fixed effects (γ_t) and municipality fixed effects (c_r) control for differences in the fraction of 16 year olds attending high school over time and for municipality specific time-invariant characteristics. As additional control variables, we include gross income per capita in the municipality, population density, and the percentage of the labour force (LF) with short, medium and long tertiary education.

Table 2 presents estimation results of equation 19, where columns (1) and (2) use the fraction of natives/immigrants as a networking measure for natives and immigrants respectively, while columns (3) and (4) use the fraction of natives/immigrants employed for natives and immigrants respectively. Across all specifications, we find no significant effects of negative attitudes on the school choice for either immigrants or natives. For immigrants, these effects are positive, but never significant, while for natives, these effects are close to zero. This is contrary to the expectations of our theoretical model, which predicts a negative impact of negative attitudes on immigrant education in case 1 and a positive impact in case 2. For networking, the first measure indicates a negative and (marginally) significant effect for immigrants and natives, while using the fractions employed, arguably a better network in relation to our theoretical model, shows a marginally significant positive effect for immigrants and no effect for natives. The overall weak effects found may be due to the fixed effect estimation which uses variation within municipalities. However, as the fixed effect approach is necessary to best handle the potential endogeneity problem, the results at the macro level fail to provide much insight into the predictions of our theoretical model.

Table 2. High School I alterpation for miningrants and Wattyes at the Municipal Level										
	(1)	(2)	(3)	(4)						
	Frac Native	Frac Imm	Frac Native	Frac Imm						
	Enrolled HS	Enrolled HS	Enrolled HS	Enrolled HS						
% Votes Both Parties	-0.10	0.88	-0.10	0.74						
	(0.22)	(0.87)	(0.22)	(0.87)						
% Native	-2.44**									
	(1.20)									
% Immigrants		-14.42*								
		(7.33)								
% Natives Employed			-0.03							
			(0.65)							
% Western Imm Employed			× ,	0.57						
				(0.93)						
% Non Western Imm Employed				0.40^{*}						
				(0.23)						
Controls?	Yes	Yes	Yes	Yes						
\mathbb{R}^2	0.234	0.076	0.195	0.082						
Ν	248	248	248	248						

Table 2: High School Participation for Immigrants and Natives at the Municipal Level

Robust standard errors are reported in parentheses. Negative attitudes are measured as the share of votes for the Fremskridtspartiet or Dansk Folkeparti. Not reported are controls for gross income per capita, population density, and the fraction of the labour force with short, medium, and long tertiary education. Vocational education is included in the short tertiary category. Medium education also includes bachelor degrees. An intercept is included in the model but not reported in the table. * p<0.10, ** p<0.05, *** p<0.01.

The natural next step is therefore to further examine the possibility of a positive networking effect for immigrants when using the fraction of immigrants employed and the positive, but insignificant, effect of negative attitudes using micro data. Moving to the individual level enables many individual and family level factors to be taken into account which cannot be at the municipal level. We are also able to more precisely measure an individual's potential network using information about their home country. Hence in the next subsection we model the educational choice of the individual young immigrants, and again natives for comparison, based on information about negative attitudes and networking possibilities in their municipality.

7.4 Micro-Econometric Model and Results

Using individual level data on 16 year old individuals living in Denmark in 2002, Table 3 presents summary statistics of all relevant variables included in the estimation. As before we see that immigrants on average attend high school less than natives. Our negative attitudes measure has a similar average value as the macro section, 12.8%.

Table 5. Summary Statistics at	
	(1)
Votes Both Parties (%)	12.74%
	(2.41)
Munc. Unemployment Rate (%)	5.21%
	(1.50)
Munc. Population Density/1000	0.7098
	(1.5072)
Munc. Gross Income Per Cap/100000	1.7592
	(0.20569)
Native $(\%)$	93.99%
	(23.77)
1^{st} Generation Immigrant (%)	3.82%
	(19.18)
2^{nd} Generation Immigrant (%)	2.19%
	(14.63)
Natives Aged 16 in Any HS $(\%)$	74.24%
	(0.00)
Imm./Desc. Aged 16 in Any HS $(\%)$	52.65%
	(0.00)
Male $(\%)$	51.06%
	(49.99)
Mother's Education	12.55
	(2.92)
Father's Education	12.33
	(3.04)
Parents Married (%)	68.53%
	(0.46)
Father Employed (%)	84.85%
	(35.86)
Mother Employed $(\%)$	82.99%
	(37.57)
Household Income/100000	6.2022
	(5.0702)
High School in Municipality $(\%)$	72.41%
	(44.7)
Observations	53256

 Table 3: Summary Statistics at the Individual Level

Mean values shown for 16 year old individuals residing in Denmark in 2002 unless otherwise indicated. Standard deviations in parentheses.

We estimate the following equation separately for natives and immigrants:

$$(1 - \hat{e}_i) = \beta_0 + \beta_1 a_r + \beta_2 \lambda_r + \beta_3 ParentEdu_p + \sum_{\mu} \beta_{r\mu} MuncControls_{r\mu} + \sum_{\eta} \beta_{i\eta} HHControls_{i\eta} + Origin_i + \varepsilon_i,$$
(20)

where $(1 - \hat{e}_i)$ is the educational decision of individual *i* represented by a dummy variable if an individual is attending any high school or not, which is determined by: a_r , negative attitudes captured by the fraction of votes for both Fremskridtspartiet and Dansk Folkeparti in municipality r; λ_r , potential networking of same nationality individuals residing the same municipality r; $ParentEdu_p$, the years of education of parent p where p = mother, father; $MuncControls_r$, municipal factors such as population density and the fraction of immigrants/natives unemployed which may affect an individuals education decision; $HHControls_i$, additional household controls such as parental employment status in December of the previous year, total household income, and parental marital status; $Origin_i$, origin country dummies that capture educational differences across specific immigrant home countries; and ε_i , residual unobservables which are clustered at the municipality level. As in the macro-econometric estimation, we use separate measures of networking in order to assess whether the quantity or quality of an individual's potential network matters where for the former, networking is measured by the fraction of own nationality individuals residing in the same municipality r where for the latter, networking is measured by both the fraction of own nationality individuals residing same municipality r who have (at least) a high school education and the fraction who are employed. As before, in order to identify the effects for immigrants, we separately estimate equations (20) for natives and for immigrants.

In the micro section, we also consider the high school choice of both 1^{st} and 2^{nd} generation immigrants in one measure. We do so in order to explore how the effects of negative attitudes on education depend on an individual's gender, as sample sizes when combining the two groups are sufficiently large.¹¹ Tables 4 and 5 present results for males and females. Columns (1) and (2) present results for natives and immigrants respectively using only the municipal fraction of own nationality individuals as a measure of networking. Columns (3) and (4) use the detailed networking measures of educated and employed own nationality individuals.

 $^{^{11}}$ Section 8.3 provides a comparison of the high school choice of 1^{st} and 2^{nd} generation immigrants.

	(1)	(2)	(3)	(4)
	Native - Any	Immigrant - Any	Native - Any	Immigrant - Any
	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing
% Votes Both Parties	0.063	1.483***	-0.028	1.275**
	(0.178)	(0.568)	(0.156)	(0.551)
Frac. of Own Nat. Educated			-0.104	0.141
			(0.093)	(0.150)
Frac. of Own Nat. Employed			0.087	0.468^{***}
			(0.179)	(0.167)
Frac of Own Nat.	0.092	-1.115		
	(0.150)	(0.724)		
Mother's Education	0.014^{***}	0.010^{***}	0.014^{***}	0.010^{***}
	(0.001)	(0.003)	(0.001)	(0.003)
Father's Education	0.016^{***}	0.006^{*}	0.016^{***}	0.006^{*}
	(0.001)	(0.003)	(0.001)	(0.003)
Parents Married	0.064^{***}	0.113^{***}	0.064^{***}	0.113^{***}
	(0.007)	(0.034)	(0.007)	(0.033)
Father Employed	0.053^{***}	0.081^{***}	0.053^{***}	0.074^{***}
	(0.009)	(0.028)	(0.009)	(0.028)
Mother Employed	0.102^{***}	0.082^{***}	0.102^{***}	0.068^{**}
	(0.009)	(0.025)	(0.009)	(0.026)
Household Income/100000	0.003^{**}	0.014^{**}	0.003^{**}	0.012^{*}
	(0.001)	(0.006)	(0.001)	(0.006)
% Natives Unemployed	-0.304		-0.252	
	(0.277)		(0.396)	
% Western Imm Unemployed		-0.136		-0.050
		(0.636)		(0.625)
% Non Western Imm Unemployed		-0.032		0.094
		(0.217)		(0.211)
Population Density/1000	-0.004	0.009^{*}	-0.004	0.007
	(0.004)	(0.006)	(0.003)	(0.005)
High School in Municipality	0.016^{*}	0.063	0.017^{**}	0.036
	(0.008)	(0.039)	(0.008)	(0.038)
Origin Country Dummies?	No	Yes	No	Yes
\mathbb{R}^2	0.064	0.186	0.064	0.192
Ν	25526	1669	25526	1669

Table 4: High School Participation for Male 16 Year Olds by Immigrant Status

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). * p<0.10, ** p<0.05, *** p<0.01.

	(1)	(2)	(3)	(4)
	Native - Any	Immigrant - Any	Native - Any	Immigrant - Any
	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing
% Votes Both Parties	0.021	0.869	-0.373**	1.184**
	(0.226)	(0.590)	(0.173)	(0.576)
Frac. of Own Nat. Educated	, , , , , , , , , , , , , , , , , , ,		-0.368***	-0.101
			(0.115)	(0.174)
Frac. of Own Nat. Employed			0.173	0.175
2 0			(0.207)	(0.193)
Frac of Own Nat.	0.501^{***}	1.824^{*}		
	(0.162)	(1.029)		
Mother's Education	0.014***	0.010***	0.014^{***}	0.010***
	(0.001)	(0.003)	(0.001)	(0.003)
Father's Education	0.014***	0.007**	0.014***	0.007**
	(0.001)	(0.003)	(0.001)	(0.003)
Parents Married	0.067***	0.045	0.066***	0.049
	(0.008)	(0.038)	(0.008)	(0.039)
Father Employed	0.077***	0.062^{**}	0.076***	0.061^{**}
1 0	(0.009)	(0.025)	(0.009)	(0.025)
Mother Employed	0.111***	0.075***	0.111***	0.074***
1 0	(0.008)	(0.026)	(0.008)	(0.027)
Household Income/100000	0.006**	-0.000	0.006**	-0.000
7	(0.003)	(0.007)	(0.003)	(0.007)
% Natives Unemployed	-0.839***	()	-0.829*	()
	(0.303)		(0.433)	
% Western Imm Unemployed	()	-1.351**	()	-1.376**
1 0		(0.589)		(0.603)
% Non Western Imm Unemployed		0.153		0.198
1 0		(0.221)		(0.222)
Population Density/1000	0.002	-0.000	-0.001	0.000
1 07	(0.003)	(0.004)	(0.003)	(0.005)
High School in Municipality	0.014	0.154***	0.016*	0.159***
S and the first of	(0.009)	(0.043)	(0.009)	(0.044)
Origin Country Dummies?	No	Yes	No	Yes
\mathbb{R}^2	0.081	0.235	0.081	0.233
Ν	24528	1533	24528	1533

Table 5: High School Participation for Female 16 Year Olds by Immigrant Status

Standard errors reported in parentheses are clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). * p<0.10, ** p<0.05, *** p<0.01.

Examining males in Table 4, networking significantly increases the propensity of an immigrant to attend high school, where a one percentage point increase in the fraction of own nationality immigrants employed residing in the municipality would lead to a 0.47 percentage point increase in the probability of attending any high school. No equivalent significant effects are seen for natives, a finding which is consistent with networking amongst immigrants. The other networking variables, the fraction of own nationality individuals residing in the same municipality and the fraction of own nationality individuals with at least a high school education, are imprecisely estimated for both natives and immigrants. This is consistent with our theoretical model, where employment prospects are key in determining the level of education an individual obtains, and it is reassuring that networking in terms of employed immigrants matters. The negative attitudes measure significantly increases an immigrant's probability of attending high school, where a 1 percentage point increase in the fraction of votes for either political party significantly increases the probability of attending any high school by 1.3-1.5 percentage points, while no effects are seen for natives. For both natives and immigrants, household controls matter a lot for an individual's propensity to attend high school, with education, employment, and marital status of both parents significantly increasing the probability of attending high school in all specifications.

For females, in Table 5, a different picture is seen. While the fraction of own nationality immigrants leads to a significant and positive increase in the probability of attending high school for immigrants, there are no significant effects, either positive or negative, of the fraction of own nationality immigrants who are educated or employed. For natives, a significant positive effect is seen for the fraction of natives while a significant negative effect is seen for the fraction of natives with education to high school or beyond. For female immigrants, negative attitudes increase high school attendance, but is imprecisely estimated depending on which networking measure is used. For natives, a similar pattern emerges when networking is measured as the fraction of natives residing in a municipality. Similar to the male estimation, parental education, employment, and marital status significantly increase the probability of attending high school for both natives and immigrants.

We find evidence that for males, negative attitudes towards immigrants increase the propensity of an immigrant to attend high school. For females, this effect is positive, but imprecisely estimated. For natives, no significant effects are seen for males nor for females. This is consistent with the macro-econometric analysis, which finds a positive, but insignificant, effect of negative attitudes on immigrant high school attendance. For male immigrants, networking seems to matter most in terms of employed immigrants of the same nationality, while for female immigrants, networking seems to matter in terms of the presence of any immigrants of the same nationality. The positive effect of negative attitudes on male immigrant education supports the second case of the theoretical model, where negative attitudes may have differential effects on immigrants propensity to attend high school depending on their productivity levels. In the case where high productivity workers are comparable to natives, as outlined in Section 3, negative attitudes only affect low productivity workers. This leads to lower employment perspectives for these low productivity workers, lowering the future wages that young immigrants expect to receive and increasing the incentives of young immigrants to acquire education.

8 Robustness of Micro-econometric Results

We examine the robustness of our micro-econometric analysis in the following subsections. In particular, we examine how the mobility of immigrants within Denmark affect the results obtained in Section 7.4, how sensitive our results on male immigrants are to the presence of unobservables, and how negative attitudes affect 1st generation immigrants compared to 2nd generation immigrants.

8.1 Exploring Mobility of Immigrants

While the results presented in Section 7.4 are supportive of the second case of the theoretical model, it is possible our estimation fails to properly estimate the impact of negative attitudes on high school attendance. In particular, we focus on the educational decision of 16 year old individuals in order to disregard mobility concerns, as students of this age are likely to reside at home in this period. While this may be the case, it could be that parents either selectively locate to certain municipalities or move as a reaction to negative attitudes in a municipality. We examine the possibility that movers are driving the positive effect of negative attitudes we see for males in Appendix C by examining if our results are stable to restricting the sample to individuals who have resided in the same house for 3 or more years and 6 or more years. Similarly, we look at years since immigration for the non-native sample in order to see if recent immigrants, who could have selectively located within Denmark, are driving our results.

On the whole, the results presented in Appendix C are very similar to the main results. For males, the negative attitude variable always increases the propensity to attend high school and is of a similar magnitude for immigrants, while there is no effect seen for natives. This is true for both the 3 or more years restriction as well as the 6 or more years restriction. For females, the effects of negative attitudes for immigrants is insignificant across specifications when imposing the years since moved restriction, which is consistent with the positive but imprecise effects seen in Table 5. The networking variables are less robust where, for males, the fraction of own nationality immigrants educated now matters while the fraction employed no longer does. For females, there are no effects of networking on the propensity to attend high school.

A similar pattern is seen for estimation restricting the time since immigration for immigrants;

the positive estimated effect of negative attitudes on the propensity to attend high school remains for males and is imprecisely estimated for females.¹² The effect of negative attitudes on high school attendance is remarkably robust for male immigrants and seems to not be driven by mobility within Denmark, while the results on networking fluctuate excluding families who have recently moved, indicative of a less stable relationship between networking and high school attendance. For female immigrants, these relationships remain insignificant.

8.2 Unobservables and Observables

In the estimation reported up to this point, unobservables could be problematic in the sense that immigrant location could be driven by unobserved parental characteristics, and these unobservable characteristics could be correlated with the high school attendance of their children. In order to make statements beyond simple associations between negative attitudes and the decision to attend high school, we examine the influence of unobservables biasing our previous results.

Oster (2015) formalises a relationship between the association between selection on observables and unobservables and R-squared values to explain the potential importance of omitted variable bias. In the method, which builds upon the work of Altonji et al. (2005), movements in the coefficients and R-squared values identify the bias resulting from omitted variables. Formally, this assumes (i) a relationship between the covariance of the variable of interest, X, and observables and the covariance of X and unobservables (proportional selection, δ) and (ii) an R-squared from a hypothetical regression of the outcome of interest, Y, on X, observables, and unobservables (R_{max}) in order to quantify the ratio in the movements of the R-squared from a regression of Y on X (\hat{R}), to a regression of Y on X and observables (\hat{R}), to this R_{max} value. Under these two assumptions, the resulting omitted variable bias is a function of δ and R_{max} . When $\delta = 1$, unobservables and observables are equally important. When $\delta = 2$, selection on unobservables is twice as important as selection on observables. Both Altonji et al. (2005) and Oster (2015) suggest equal selection, when $\delta = 1$, as a reasonable bound. When $R_{max} = 1$, Y is fully explained by X and both observable and unobservable controls. As this may not be reasonable to assume,¹³ Oster suggests a reasonable choice of $R_{max} = 1.3\tilde{R}$.

Two types of statements on the importance of unobservables result from these assumptions:

 $^{^{12}2^{}nd}$ generation immigrants, by definition, have "immigrated" when they are born, so all and 2^{nd} generation immigrants are included in both the years since immigrated tables.

¹³For example, this may overstate the total explanatory power of these potential variables. If $R_{max} = 1$, high school attendance can be completely explained, implying no optimization errors in an individual's optimal schooling and zero measurement error in data.

(i) the degree of selection on unobservables compared to observables, δ , required to completely explain away the estimated effect of X on Y for a given R_{max} and (ii) statements of bounds of the estimated effect of X on Y for a given level of δ and R_{max} . We examine the sensitivity of the positive and significant effect of negative attitudes on high school attendance we see previously for males to unobservables in Table 6. Column (1) reports the raw effect of negative attitudes on immigrant's high school decision from a regression of Y on X. Column (2) is the controlled effect, including the observables, reported in column (4) of Table 4 and acts as an upper bound for the estimated effect of negative attitudes on high school attendance. The reported R-squared value corresponds to R.

Table 6: Importance of Unobservables and Omitted Variable Bias Male 16 Year Olds										
	(1) Baseline Effect	(2) Controlled Effect	(3) Identified Set	$\begin{array}{c} (4)\\ \delta \text{ for}\\ \beta = 0 \end{array}$	(5) Bias-adjusted β if $\delta = 2$	(6) Bias-adjusted β if $R_{max} = 1$				
$\%$ Votes Both Parties $\rm R^2$	$\begin{array}{c} 1.454^{**} \\ (0.560) \\ 0.006 \end{array}$	$\begin{array}{c} 1.275^{**} \\ (0.551) \\ 0.192 \end{array}$	[1.214,1.275]	13.844	1.150	0.072				
N	1669	1669	1669	1669	1669	1669				

T 1 1 0 T

Standard errors reported in parentheses clustered at municipality level. * p<0.10, ** p<0.05, *** p<0.01. R^2 reported in columns (1) and (2) correspond to \mathring{R} and \tilde{R} respectively in Oster (2015). Column (3) presents bounded set for effect of negative attitudes on high school attendance accounting for role of unobservables assuming $\delta = 1$ and $R_{max} = 1.3\tilde{R}$. Column (4) presents the level of selection on unobservables relative to observables required for estimated effect in column (2) to be zero. Column (5) presents the lower bound estimate of column (3) when $\delta = 2$. Column (6) presents the lower bound estimate of column (3) when $R_{max} = 1$. Calculations done using psacalc developed in Oster (2015).

Column (3) reports the bounded set taking the value from column (2) and the lower bound estimate assuming $\delta = 1$ and $R_{max} = 1.3\tilde{R}$ using the procedure developed by Oster (2015). The resulting bound suggests that, using reasonable values for the necessary assumptions, we can exclude the possibility that negative attitudes decrease high school attendance for male immigrants. Despite these relatively tight bounds, we can further investigate what properties, in terms of unobservables, would have to hold in order to explain away the entire positive effect of negative attitudes on education. This is crucial to examine in further detail as unobservables may have a major role in explaining the effects of negative attitudes on high school attendance seen previously. Column (4) reports the level of δ required for the effect of negative attitudes on high school attendance to be zero, and indicates that selection on unobservables would have to be far greater than selection on observables, nearly 14 times more important, in order for $\beta = 0$. Columns (5) and (6) present two additional results: the lower bound estimate from increasing δ from column

(3) from 1 to 2 and the lower bound estimate from increasing the R_{max} from column (3) to 1. Even if we assume that high school attendance can be fully explained by negative attitudes and both observables and unobservables, the effects of negative attitudes on high school attendance are still non-negative. The results of Table 6 indicate that for males, the effect of negative attitudes on high school attendance is bounded away from zero, and that negative attitudes do not decrease high school attendance. This effectively excludes the first case of our theoretical model, when negative attitudes affect all immigrants and decrease immigrant education, for male immigrants.

8.3 Comparing 1st and 2nd Generation Immigrants

Table A8 analyses the high school decision of 1st and 2nd generation immigrants separately, allowing the effects of negative attitudes on education to affect the two groups differently. While 1st and 2nd generation immigrants can both be impacted by negative attitudes, 2nd generation immigrants have been raised in Denmark and may have assimilated more compared to 1st generation immigrants. As we expect that 2nd generation immigrants may be less adversely impacted by negative attitudes, finding a larger impact of negative attitudes on 1st generation immigrants than on 2nd generation immigrants would support that the voting data captures negative attitudes while finding a larger impact for 2nd generation immigrants would raise concerns about the validity of this measure. In addition, we expect that networking may be more important to 1st generation immigrants, as they may be less assimilated. Due to the few numbers of 2nd generation immigrants in a given municipality, males and females are combined into one sample, and a dummy for male is included as a control variable. For the same reason, the results presented in Table A8 should be interpreted with some caution, as there are some municipalities with very few 2nd generation immigrants residing in them.

Estimation using only 1^{st} generation immigrants with the different networking measures is reported in columns (1) and (3) of Table A8 while estimation on a sample of only 2^{nd} generation immigrants is reported in columns (2) and (4). Comparing these two columns reveals that the positive impact of negative attitudes on high school attendance seen previously is driven by the impact of negative attitudes on 1^{st} generation immigrants, while virtually no effect is seen on the probability of attending high school for 2^{nd} generation immigrants. For networking, there is an effect seen for 1^{st} generation immigrants, in terms of the fraction of own nationality immigrants employed, while there is no effect for 2^{nd} generation immigrants. It is also reassuring that household and municipality controls, which are not reported, are relatively similar for 1st and 2nd generation immigrants, as while negative attitudes may affect the two groups differently, there is less reason to believe that the two types of immigrants would be differentially affected by other controls.

9 Conclusion

We considered the impact of negative attitudes and immigration on educational choice of immigrants and natives. We did this theoretically and empirically.

Theoretically, we formulated a Becker-style taste discrimination model within a search and wage bargaining setting. We assumed that potential negative tastes towards immigrants implied that their separation rate from the job was higher than the separation rate of a native worker. Furthermore, we allowed for networking effects, which increased the probability of obtaining employment. We included endogenous education, where a higher expected income as educated in terms of both employment chances and wages relative to the expected income as uneducated, increases the number of educated workers. We considered two different cases. In the first case, discrimination existed for all immigrants, while it was only present in the sector employing uneducated workers in the second case. We found that an increase in negative attitudes reduced education for immigrants in the first case, increased education in the second case, where there was no impact on natives in the two cases. We also found that more immigration improved employment perspectives for immigrants and thereby increased the fraction of educated immigrants due to increased networking. Finally, we considered endogenous negative attitudes in the sense that more immigration increased negative attitudes. In this case, the impact of more immigration on the educational level of immigrants was ambiguous.

Empirically, we considered an immigrant's high school attendance as a function of the variables in the theoretical model. Exploiting the panel nature of the data at the macro-econometric level, we found positive, but insignificant, effects of negative attitudes on the fraction of immigrants attending high school. We found little of impact of networking, and turned to estimation at the individual level in order to more precisely measure an individual's potential network but also to account for important household level controls. At the micro-econometric level, we found positive impacts of negative attitudes on young immigrant's high school attendance which were significant for males, but less precisely estimated for females. We estimated positive impacts of networking on high school attendance for immigrants, where the quality of networks mattered for males while only the quantity of individuals in a potential network mattered for females. Overall, our baseline empirical findings are consistent with the second case of the theoretical model, where negative attitudes are prevalent in the sector hiring low skilled workers and more severe negative attitudes increase the incentives to acquire education.

Considering high school attendance in the institutional context of the Danish education system allowed us to, plausibly, disregard mobility issues for the individual acquiring education. As these individuals usually live at home while attending high school, the parents decide where to live and the young immigrants and natives, placed by their parents in a particular municipality, decide whether to attend high school. We therefore obtain a situation were placement is plausibly exogenous for the person taking the educational decision. Despite this, we addressed concerns that unobservable factors could drive parents, either in their emigration or subsequent relocation, to locate in order to give the young immigrant or native a better choice of high school prior to this decision in two robustness checks. Firstly, we examined the possibility that parents could re-locate within Denmark in order to affect the young person's educational decision by estimating the effects of negative attitudes on high school attendance for young people who had not moved recently, in the last 3 and 6 years. Excluding these movers, we see very similar effects as our baseline individual level findings for the impact of negative attitudes and, to a lesser extent, networking. However, as the decision to stay is also endogenously determined we, secondly, directly assessed the importance of unobservable characteristics compared to observables in explaining our estimated effects of negative attitudes on high school attendance using a procedure developed in Oster (2015). While there are reasons to believe that concerns over the importance of unobservables may be mitigated due to the timing of the decision to attend high school, we were able to directly quantify how our estimates change depending on the degree of omitted variable bias due to unobservables. Doing so, we found that, under reasonable assumptions of the importance of unobservables, we were able to bound the estimated effect for males away from zero. As such, we could confidently assert that even accounting for potentially correlated unobservables that negative attitudes have a positive impact on high school attendance for male immigrants. We also saw that our findings are driven by 1st generation immigrants rather than 2nd generation immigrants, who are likely more assimilated.

If negative attitudes are most prevalent in the uneducated sector, and the second case of the

model is the most realistic one, negative attitudes cannot be the explanation behind the lower fraction of immigrants attending high school. Rather, negative attitudes towards immigrants may induce more immigrants to continue school beyond the 9th grade and thus improve their employment chances, wages, and expected lifetime income significantly. On the other hand, as female immigrants tend to benefit from networking with a quantity of individuals of their own nationality and male immigrants tend to benefit from networking with quality individuals of their own nationality, a low fraction of immigrants and quality immigrants in a municipality seems to partly explain the low fraction of female and male immigrant high school attendance. Hence, our results indicate that potential immigrant high school students have higher incentives to attend high school in areas where many of their own nationality or many of their own nationality who are employed live as it improves their networking chances. While we cannot exclude that variables other than negative attitudes and networking can explain the high school educational gap between immigrant and natives, the main empirical finding of this paper is that, at least for males, discrimination towards immigrants cannot explain the educational gap between immigrants and natives, as male immigrants tend to avoid discrimination by acquiring further education.

References

- Andersson, F., Burgess, S. and Lane, J., 2009: Do as the Neighbors Do: The Impact of Social Networks on Immigrant Employment, IZA Discussion Papers 4423, Institute for the Study of Labor (IZA).
- [2] Altonji, J., Elder, T., and Taber, C., 2005: Selection on Observed and Unobserved Variables: Assessing the Effectiveness of Catholic Schools, Journal of Political Economy, vol 113(1), 2005, 151-184, February.
- [3] Bowlus, A. and Eckstein, Z., 2002: Discrimination and skill Differences in an Equilibrium Search Model, International Economic Review, vol 43(4), 2002, 1309-1345, November.
- [4] Calvo-Armengol, A. and Jackson, M. O. 2004: The Effect of Social Networks on Employment and Inequality. American Economic Review 94(3), pp. 426-454.
- [5] Damm, A. P., 2014: Neighborhood quality and labor market outcomes: Evidence from quasirandom neighborhood assignment of immigrants, Journal of Urban Economics 79, 139–166.
- [6] Dansk Data Arkiv, Survey 21174. For year 2002.
- [7] http://www.danskfolkeparti.dk/The_Party_Program_of_the_Danish_Peoples_Party
- [8] Fontaine, F., 2007: A Simple Matching Model with Social Networks. Economic Letters, 94 (3).

- [9] Flabbi, L., 2010: Gender Discrimination Estimation In A Search Model With Matching And Bargaining, International Economic Review, vol. 51(3), 745-783, 08.
- [10] Galenianos, M., 2014: Hiring through referrals, Journal of Economic Theory, 152, 304-323.
- [11] http://www.fremskridtspartiet.dk/page9.html
- [12] Hellerstein, J., McInerney, M. and Neumark, D., 2009. Spatial Influences on the Employment of U.S. Hispanics: Spatial Mismatch, Discrimination, or Immigrant Networks?, Working Papers 09-03. Center for Economic Studies, U.S. Census Bureau.
- [13] Nielsen, H. S. and Rangvid, B.S., 2012: The impact of parents' years since migration on children's academic achievement. IZA Journal of Migration1:6.
- [14] Kofi Charles, K. and Guryan, J., 2008: Prejudice and Wages: An Empirical Assessment of Becker's The Economics of Discrimination. Journal of Political Economy. Vol. 116, No. 5 (October 2008), 773-809.
- [15] Kramarz, F. and Skans, O. N., 2014: When Strong Ties are Strong: Networks and Youth Labour Market Entry, Review of Economic Studies, 81, 1174-1200.
- [16] Mailath, G. J., Samuelson, L. and Shaked, A., 2000: Endogenous Inequality in Integrated Labor Markets with Two-Sided Search. American Economic Review, vol. 90(1), 46-72, March.
- [17] Lang, K., Manove, M. and Dickens, W. T., 2005: Racial Discrimination in Labor Markets with Posted Wage Offers. American Economic Review, vol. 95(4), 1327-1340, September.
- [18] Larsen, B. and Waisman, G., 2012: Labour Market Performance Effects of Discrimination and Loss of Skill. Working Paper, Copenhagen Business School.
- [19] OECD 2006: Where immigrant students succeed A comparative review of performance and engagement in Pisa 2003.
- [20] Oster, E., 2015: Unobservable Selection and Coefficient Stability: Theory and Evidence. Mimeo.
- [21] Solignac, M. and Tô, M., 2013: Neighbourhood Effect and Transitions from school to work. Mimeo.
- [22] Waisman, G. and Larsen, B., 2015: Wages, Amenities and Negative Attitudes. Mimeo.

Appendices

A Proofs of Theoretical Propositions

A.1 Proof for Negative Attitudes Proposition in Section 3

First case: Differentiating equations (11), (12), (10) and (9) with respect to $a^{h} = a^{l} = a$ shows that there is a negative impact on labour market tightness and wages facing immigrants but no impact for natives $\frac{d\theta_{I}}{da} = \frac{-k2}{D_{I}}\frac{ds_{I}}{da} < 0$, $\frac{d\theta_{N}}{da} = 0$, $\frac{dw_{I}^{m}}{da} = 0.5y^{m}\frac{d\theta_{I}}{da}k < 0$, $\frac{dw_{N}}{da} = 0$. Concerning unemployment and education, we differentiate equations (16) and (14) with respect to a to obtain:

$$\frac{du_I}{da} = \frac{-s_I f'(\theta_I)}{(f(\theta_I) + s_I)^2} \frac{d\theta_I}{da} > 0, \ \frac{du_N}{da} = 0, \ \frac{d\hat{e}_I}{da} = \frac{(y^h - y^l) \frac{d\theta_I}{da}}{c'(\hat{e}_I)} > 0, \ \frac{d\hat{e}_N}{da} = 0.$$

In the second case, the results for immigrants change to $\frac{d\theta_I^l}{da^l} = \frac{-k2}{D_I^l} \frac{ds_I^l}{da^l} < 0, \quad \frac{d\theta_I^h}{da^l} = 0, \quad \frac{dw_I^l}{da^l} = 0, \quad$

A.2 Proof for Immigration Proposition in Section 4

For both cases: Differentiating equations (11), (12), (10) and (9) with respect to I delivers a positive impact on labour market tightness for immigrants and a negative impact on natives:

 $\left(d\theta_{I}^{m}\right)/\left(dI\right)=\left(tkq(\theta_{I}^{m})\right)/D_{I}^{m}>0,\ m=h,l,\ \left(d\theta_{N}\right)/\left(dI\right)=ktq(\theta_{N})/D_{N}<0,$

$$\left(dw_{I}^{m}\right)/\left(dI\right)=0.5y^{m}\left(t+\left(d\theta_{I}^{m}\right)/\left(dI\right)\right)k>0,\ \frac{dw_{N}^{m}}{dI}=0.5y^{m}\left(-t+\left(d\theta_{N}^{m}\right)/\left(dI\right)\right)k<0.$$

Again for both case 1 and 2, we differentiate equations (16) and (17) with respect to I to obtain:

$$\frac{du_{I}}{dI} = \frac{-s_{I}f'(\theta_{I})}{(f(\theta_{I}) + s_{I})^{2}}\frac{d\theta_{I}}{dI} < 0, \ \frac{du_{I}^{m}}{dI} = \frac{-s_{I}^{m}f'(\theta_{I}^{m})}{(f(\theta_{I}^{m}) + s_{I}^{m})^{2}}\frac{d\theta_{I}^{m}}{dI} < 0, \ \frac{du_{N}}{dI} = \frac{-s_{N}f'(\theta_{N})}{(f(\theta_{N}) + s_{N})^{2}}\frac{d\theta_{N}}{dI} > 0.$$

Concerning education, for case 1, we differentiate equation (14) with respect to I to obtain:

$$(d\hat{e}_{I}) / (dI) = (y^{h} - y^{l}) (d\theta_{I} / (dI)) / c'(\hat{e}_{I}) < 0, \quad (d\hat{e}_{N}) / (dI) = (y^{h} - y^{l}) (d\theta_{N} / (dI)) / c'(\hat{e}_{N}) > 0.$$

In the second case, the result for education for immigrants is:

 $\left(d\hat{e}_I \right) / \left(dI \right) = \left(y^h \left(d\theta_I^h / \left(dI \right) \right) - y^l \left(d\theta_I^l / \left(dI \right) \right) \right) / c' \left(\hat{e}_I \right) < 0, \text{ which is negative as } d\theta_I^h / \left(dI \right) > d\theta_I^l / \left(dI \right). \text{ Q.E.D.}$

Summary of Heterogeneous Networking Effects В

Table A1: Heterogeneous Networking Effects									
Separation	Networking immigrants	Networking natives	Labour Market Tightness	Education					
	$\lambda_I^h = \lambda_I^l$	$\lambda_N^h > \lambda_N^l$	$\theta^h_N > \theta^l_N > \theta^l_I = \theta^h_I$	$\hat{e}_I > \hat{e}_N$					
$s_I^l = s_I^h > s_N$	$\lambda_I^h = \lambda_I^l$	$\lambda_N^h < \lambda_N^l$	$\theta_N^l > \theta_N^h > \theta_I^l = \theta_I^h$	ambiguous					
	$\lambda_I^l > \lambda_I^h$	$\lambda_N^h \geq \lambda_N^l$	$\theta^h_N \geq \theta^l_N > \theta^l_I > \theta^h_I$	$\widehat{e}_I > \widehat{e}_N$					
	$\lambda_I^l > \lambda_I^h$	$\lambda_N^h < \lambda_N^l$	$\theta_N^l \geq \theta_N^h > \theta_I^l > \theta_I^h$	ambiguous					
	$\lambda_I^h > \lambda_I^l$	$\lambda^h_N \lesseqgtr \lambda^l_N$	$\theta^h_N \stackrel{<}{\scriptscriptstyle >} \theta^l_N > \theta^h_I > \theta^l_I$	ambiguous					
	$\lambda_I^h = \lambda_I^l$	$\lambda_N^h > \lambda_N^l$	$\theta^h_N > \theta^l_N > \theta^h_I > \theta^l_I$	ambiguous					
$s_{r}^{l} > s_{r}^{h} = s_{N}$	$\lambda^h_I = \lambda^l_I$	$\lambda^h_N < \lambda^l_N$	$\theta_N^l > \theta_N^h > \theta_I^h > \theta_I^l$	ambiguous					
$v_I > v_I = v_N$	$\lambda_I^l > \lambda_I^h$	$\lambda_N^h \geq \lambda_N^l$	$\theta^h_N \geq \theta^l_N > \theta^h_I ? \theta^l_I$	ambiguous					
	$\lambda_I^l > \lambda_I^h$	$\lambda_N^h < \lambda_N^l$	$\theta_N^l \geq \theta_N^h > \theta_I^h ? \theta_I^l$	ambiguous					
	$\lambda_I^h > \lambda_I^l$	$\lambda_N^h \lesseqgtr \lambda_N^l$	$\theta^h_N \stackrel{<}{_{>}} \theta^l_N > \theta^h_I > \theta^l_I$	ambiguous					

T-hla A1. Hotorogor Networking Effects

 s_i^m corresponds to the job separation rate for high and low productivity, m = h, l, immigrants and natives, i = N, I. λ_I^m corresponds to networking for high and low productivity, for immigrants. λ_N^m corresponds to networking for high and low productivity, for natives. θ_i^m corresponds to the labour market tightness for high and low productivity, m = h, l, immigrants and natives, i = N, I. \hat{e}_i corresponds to education for immigrants and natives, i = N, I.

C Robustness Checks

C.1 3 or More Years Since Moved

Table A2: High School Participation for Male 16 Year Olds by Immigrant Status - 3 or More Years Since Moved

	(1)	(2)	(3)	(4)
	Native - Any	Immigrant - Any	Native - Any	Immigrant - Any
	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing
% Votes Both Parties	0.115	1.680**	0.039	1.434**
	(0.188)	(0.697)	(0.168)	(0.696)
Frac. of Own Nat. Educated	, ,		-0.078	0.503**
			(0.096)	(0.237)
Frac. of Own Nat. Employed			0.038	0.015
			(0.175)	(0.228)
Frac. of Own Nat.	0.079	-1.815**		
	(0.159)	(0.810)		
Parental Education Controls?	Yes	Yes	Yes	Yes
Household Controls?	Yes	Yes	Yes	Yes
Municipality Controls?	Yes	Yes	Yes	Yes
Origin Country Dummies?	No	Yes	No	Yes
R^2	0.052	0.117	0.052	0.118
Ν	21861	1276	21861	1276

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 4. * p<0.10, ** p<0.05, *** p<0.01.

Table	A3:	High	School	Participation	for	Female	16	Year	Olds	by	Immigrant	Status	- 3	or	More
Years	Since	e Mov	red												

	(1)	(2)	(3)	(4)
	Native - Any	Immigrant - Any	Native - Any	Immigrant - Any
	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing
% Votes Both Parties	0.082	0.612	-0.330*	0.862
	(0.234)	(0.573)	(0.178)	(0.552)
Frac. of Own Nat. Educated			-0.358***	0.019
			(0.118)	(0.210)
Frac. of Own Nat. Employed			0.110	0.120
			(0.201)	(0.218)
Frac. of Own Nat.	0.551^{***}	1.258		
	(0.165)	(0.892)		
Parental Education Controls?	Yes	Yes	Yes	Yes
Household Controls?	Yes	Yes	Yes	Yes
Municipality Controls?	Yes	Yes	Yes	Yes
Origin Country Dummies?	No	Yes	No	Yes
R^2	0.062	0.153	0.062	0.152
Ν	20795	1171	20795	1171

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 5. * p<0.10, ** p<0.05, *** p<0.01.

C.2 6 or More Years Since Moved

	(1)	(2)	(3)	(4)
	Native - Any HS Ongoing	Immigrant - Any HS Ongoing	Native - Any HS Ongoing	Immigrant - Any HS Ongoing
% Votes Both Parties	0.201 (0.179)	2.308^{***} (0.848)	0.115 (0.159)	2.000^{**} (0.804)
Frac. of Own Nat. Educated	· · /		-0.042 (0.095)	0.525 (0.356)
Frac. of Own Nat. Employed			-0.100 (0.165)	(0.037) (0.284)
Frac. of Own Nat.	0.140 (0.141)	-2.075^{**} (1.006)	(01200)	(0.201)
Parental Education Controls?	Yes	Yes	Yes	Yes
Household Controls?	Yes	Yes	Yes	Yes
Municipality Controls?	Yes	Yes	Yes	Yes
Origin Country Dummies?	No	Yes	No	Yes
\mathbb{R}^2	0.045	0.109	0.045	0.108
Ν	18259	928	18259	928

Table A4: High School Participation for Male 16 Year Olds by Immigrant Status - 6 or More Years Since Moved

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 4. * p<0.10, ** p<0.05, *** p<0.01.

Table .	A5:	High School	Participation	for	Female	16	Year	Olds by	^r Immigrant	Status	- 6	or	More
Years S	Since	e Moved											

	(1) Native - Any HS Ongoing	(2) Immigrant - Any HS Ongoing	(3) Native - Any HS Ongoing	(4) Immigrant - Any HS Ongoing	
% Votes Both Parties	0.116	0.206	-0.282	0.371	
Frac. of Own Nat. Educated	(0.254)	(0.639)	(0.202) - 0.326^{**}	(0.626) -0.110 (0.265)	
Frac. of Own Nat. Employed			(0.129) 0.051 (0.226)	(0.205) 0.132 (0.280)	
Frac. of Own Nat.	0.582^{***}	0.895 (1.030)	(0.220)	(0.200)	
Parental Education Controls?	Yes	Yes	Yes	Yes	
Household Controls?	Yes	Yes	Yes	Yes	
Municipality Controls?	Yes	Yes	Yes	Yes	
Origin Country Dummies?	No	Yes	No	Yes	
\mathbb{R}^2	0.054	0.133	0.053	0.132	
Ν	17435	853	17435	853	

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 5. * p<0.10, ** p<0.05, *** p<0.01.

C.3 3 or More Years Since Immigrated

	(1) Male - Any HS Ongoing	(2) Male - Any HS Ongoing	(3) Female - Any HS Ongoing	(4) Female - Any HS Ongoing
% Votes Both Parties	1.603^{***} (0.601)	1.508^{**} (0.598)	$0.874 \\ (0.561)$	1.178^{**} (0.554)
Frac. of Own Nat. Educated		0.500^{**} (0.218)		-0.067 (0.226)
Frac. of Own Nat. Employed		0.105 (0.204)		-0.040 (0.211)
Frac. of Own Nat.	-1.207^{*} (0.724)		$1.515 \\ (0.924)$	
Parental Education Controls?	Yes	Yes	Yes	Yes
Household Controls?	Yes	Yes	Yes	Yes
Municipality Controls?	Yes	Yes	Yes	Yes
Origin Country Dummies?	Yes	Yes	Yes	Yes
R^2	0.132	0.137	0.155	0.154
Ν	1524	1524	1398	1398

Table A6: High School Participation for Immigrant 16 Year Olds - 3 or More Years Since Immigrated_____

Standard errors reported in parentheses clustered at municipality level. * p<0.10, ** p<0.05, *** p<0.01. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 4 and 5.

C.4 6 or More Years Since Immigrated

Table A7:	High	School	Participatio	on fo	r Immigra	ant 16	5 Year	Olds -	- 6 or	More	Years	Since	Immig-
rated													

	(1)	(2)	(3)	(4)
	Male - Any	Male - Any	Female - Any	Female - Any
_	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing
% Votes Both Parties	1.669^{***}	1.529**	0.643	0.882*
	(0.597)	(0.590)	(0.523)	(0.521)
Frac. of Own Nat. Educated		0.515**		-0.226
		(0.235)		(0.207)
Frac. of Own Nat. Employed		0.040		0.069
		(0.215)		(0.203)
Frac. of Own Nat.	-1.298*		1.324	
	(0.779)		(0.839)	
Parental Education Controls?	Yes	Yes	Yes	Yes
Household Controls?	Yes	Yes	Yes	Yes
Municipality Controls?	Yes	Yes	Yes	Yes
Origin Country Dummies?	Yes	Yes	Yes	Yes
R^2	0.118	0.122	0.135	0.134
Ν	1393	1393	1274	1274

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs (apprenticeships). Includes same parental education, household, and municipality controls as Table 4 and 5. * p<0.10, ** p<0.05, *** p<0.01.

C.5 1^{st} and 2^{nd} Generation Immigrants

10010 1101 11181 2011					
	(1)	(2)	(3)	(4)	
	^{1st} Gen Any	2 nd Gen Any	^{1st} Gen Any	2 nd Gen Any	
	HS Ongoing	HS Ongoing	HS Ongoing	HS Ongoing	
% Votes Both Parties	1.575^{***}	-0.035	1.652^{***}	0.035	
	(0.511)	(0.553)	(0.505)	(0.548)	
Frac. of Own Nat. Educated			0.065	-0.077	
			(0.111)	(0.256)	
Frac. of Own Nat. Employed			0.344**	0.309	
			(0.134)	(0.241)	
Frac. of Own Nat.	0.919	0.474		· · ·	
	(1.524)	(0.699)			
Male	-0.054**	-0.073**	-0.054**	-0.074**	
	(0.024)	(0.029)	(0.023)	(0.029)	
Parental Education Controls?	Yes	Yes	Yes	Yes	
Household Controls?	Yes	Yes	Yes	Yes	
Municipality Controls?	Yes	Yes	Yes	Yes	
Origin Country Dummies?	Yes	Yes	Yes	Yes	
R^2	0.230	0.096	0.234	0.097	
Ν	2037	1165	2037	1165	

Table A8: High School Participation for 1st and 2nd Generation Immigrants

Standard errors reported in parentheses clustered at municipality level. Any HS Ongoing corresponds to either enrollment in regular high school, business high school, or vocational training programs

(apprentices hips). Includes same parental education, household, and municipality controls as Table 4 and 5. * p<0.10, ** p<0.05, *** p<0.01.

Conclusion

This Ph.D. thesis examined the determinants and social implications of education, crime, and job displacement. This was done using detailed Danish Register Data which provided the detail and comprehensiveness required for such analysis. The first chapter, "The Heterogeneous Effects of Education on Crime: Evidence from Danish Administrative Twin Data", assessed whether education is equally effective in lowering crime for all individuals, and found that education is most effective in lowering crime for males, for those of low educated households, and for the completion of high school. No effects of education on crime were seen for those of highly educated households, and education was found to lower crime irrespective of the level of crime an individual is exposed to during childhood. The results were robust to using data on incarcerations, to imputing the estimated effects of education on crime that would result from using a sample of identical twins, to excluding twins with large differences in education, and to directly estimating the reverse causality that exists between education and crime. The results found in Chapter 1 imply that these "at risk" individuals for whom education lowers crime obtain levels of education which are less than socially optimal, as they fail to consider the additional social benefits of education in terms of reduced crime.

The second chapter, "Job Displacement and Crime", examined the individual level relationship that exists between unemployment and crime, and found that individuals who lost their job in a sudden and unexpected mass-layoff event were significantly more likely than non-displaced individuals to engage in criminal activity after displacement, but not significantly different in the time leading up to displacement. These effects were found to be long lasting, and exist up to 7 years after displacement. These effects were found mainly for property crimes and to be concentrated at the lower end of the educational distribution, for those with less than high school education and for those with vocational education. The potential for an intergenerational effect of displacement on children's crime was examined, and revealed that, at best, father's displacement has a marginal short-run impact on son's crime. Large and long-lasting effects were found for those unmarried or living with no other adults. Job displacement impacted crime over and above what was explained by individual earnings losses, while weeks spent in unemployment explained substantially more of the impact of displacement on crime. Our results were robust to a variety of specification changes and alterations to mass-layoff criteria, and suggest that both employers and employees fail to take into account the social costs of job separation, creating the potential for active public policy.

The third chapter, "Negative Attitudes, Network and Education", assessed potential explanations behind the educational gap existing between natives and immigrants. This was done both theoretically and empirically using two potential measures, negative attitudes towards immigrants and networking. Theoretically, we determined that negative attitudes can decrease the levels of education obtained by immigrants if negative attitudes affect all types of immigrants, while if they affect only low-skilled immigrants, negative attitudes can increase the levels of education of immigrants who educate in order to improve their employment prospects. Empirically, we analyzed an immigrant's educational choice at the macro- and micro-levels, and found a positive impact of negative attitudes on educational attainment, particularly for male immigrants. We found suggestive evidence of a positive impact of networking on education which differed across genders. We determined that, under reasonable assumptions about the importance of unobservables, while these unobservable factors do explain some of our results, we were able to eliminate the possibility that negative attitudes have a negative impact on immigrant education. As such, negative attitudes are determined to not explain the educational gap between immigrants and natives as they motivate immigrants to obtain more education, while a lack of networking could explain some of this gap.

Overall, this thesis emphasizes that taking into account specific factors in and the social implications of individual-level decisions matter greatly, both in terms of heterogeneity and social spillovers. This thesis has shown that these externalities can have meaningful impacts both on the social costs and benefits of public expenditure. As such, accounting for heterogeneity and externalities is necessary not only for the sake of drawing accurate conclusions about the relevant factors assessed in this thesis but also for public policy to be designed effectively.

TITLER I PH.D.SERIEN:

2004

- 1. Martin Grieger Internet-based Electronic Marketplaces and Supply Chain Management
- 2. Thomas Basbøll LIKENESS A Philosophical Investigation
- 3. Morten Knudsen Beslutningens vaklen En systemteoretisk analyse of moderniseringen af et amtskommunalt sundhedsvæsen 1980-2000
- 4. Lars Bo Jeppesen Organizing Consumer Innovation A product development strategy that is based on online communities and allows some firms to benefit from a distributed process of innovation by consumers
- 5. Barbara Dragsted SEGMENTATION IN TRANSLATION AND TRANSLATION MEMORY SYSTEMS An empirical investigation of cognitive segmentation and effects of integrating a TM system into the translation process
- 6. Jeanet Hardis Sociale partnerskaber Et socialkonstruktivistisk casestudie af partnerskabsaktørers virkelighedsopfattelse mellem identitet og legitimitet
- 7. Henriette Hallberg Thygesen System Dynamics in Action
- 8. Carsten Mejer Plath Strategisk Økonomistyring
- 9. Annemette Kjærgaard Knowledge Management as Internal Corporate Venturing

– a Field Study of the Rise and Fall of a Bottom-Up Process

- 10. Knut Arne Hovdal De profesjonelle i endring Norsk ph.d., ej til salg gennem Samfundslitteratur
- Søren Jeppesen Environmental Practices and Greening Strategies in Small Manufacturing Enterprises in South Africa – A Critical Realist Approach
- 12. Lars Frode Frederiksen Industriel forskningsledelse – på sporet af mønstre og samarbejde i danske forskningsintensive virksomheder
- 13. Martin Jes Iversen
 The Governance of GN Great Nordic
 in an age of strategic and structural transitions 1939-1988
- 14. Lars Pynt Andersen The Rhetorical Strategies of Danish TV Advertising A study of the first fifteen years with special emphasis on genre and irony
- 15. Jakob Rasmussen Business Perspectives on E-learning
- Sof Thrane
 The Social and Economic Dynamics of Networks
 – a Weberian Analysis of Three
 Formalised Horizontal Networks
- 17. Lene Nielsen Engaging Personas and Narrative Scenarios – a study on how a usercentered approach influenced the perception of the design process in the e-business group at AstraZeneca
- S.J Valstad
 Organisationsidentitet
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur

- 19. Thomas Lyse Hansen Six Essays on Pricing and Weather risk in Energy Markets
- 20. Sabine Madsen Emerging Methods – An Interpretive Study of ISD Methods in Practice
- 21. Evis Sinani The Impact of Foreign Direct Investment on Efficiency, Productivity Growth and Trade: An Empirical Investigation
- 22. Bent Meier Sørensen Making Events Work Or, How to Multiply Your Crisis
- 23. Pernille Schnoor Brand Ethos Om troværdige brand- og virksomhedsidentiteter i et retorisk og diskursteoretisk perspektiv
- 24. Sidsel Fabech Von welchem Österreich ist hier die Rede? Diskursive forhandlinger og magtkampe mellem rivaliserende nationale identitetskonstruktioner i østrigske pressediskurser
- 25. Klavs Odgaard Christensen Sprogpolitik og identitetsdannelse i flersprogede forbundsstater Et komparativt studie af Schweiz og Canada
- 26. Dana B. Minbaeva Human Resource Practices and Knowledge Transfer in Multinational Corporations
- 27. Holger Højlund Markedets politiske fornuft Et studie af velfærdens organisering i perioden 1990-2003
- 28. Christine Mølgaard Frandsen A.s erfaring Om mellemværendets praktik i en

transformation af mennesket og subjektiviteten

29. Sine Nørholm Just
The Constitution of Meaning
– A Meaningful Constitution?
Legitimacy, identity, and public opinion in the debate on the future of Europe

2005

- 1. Claus J. Varnes Managing product innovation through rules – The role of formal and structured methods in product development
- Helle Hedegaard Hein Mellem konflikt og konsensus

 Dialogudvikling på hospitalsklinikker
- 3. Axel Rosenø Customer Value Driven Product Innovation – A Study of Market Learning in New Product Development
- 4. Søren Buhl Pedersen Making space An outline of place branding
- 5. Camilla Funck Ellehave Differences that Matter An analysis of practices of gender and organizing in contemporary workplaces
- 6. Rigmor Madeleine Lond Styring af kommunale forvaltninger
- 7. Mette Aagaard Andreassen Supply Chain versus Supply Chain Benchmarking as a Means to Managing Supply Chains
- 8. Caroline Aggestam-Pontoppidan From an idea to a standard The UN and the global governance of accountants' competence
- 9. Norsk ph.d.
- 10. Vivienne Heng Ker-ni An Experimental Field Study on the

Effectiveness of Grocer Media Advertising Measuring Ad Recall and Recognition, Purchase Intentions and Short-Term Sales

- 11. Allan Mortensen Essays on the Pricing of Corporate Bonds and Credit Derivatives
- 12. Remo Stefano Chiari Figure che fanno conoscere Itinerario sull'idea del valore cognitivo e espressivo della metafora e di altri tropi da Aristotele e da Vico fino al cognitivismo contemporaneo
- 13. Anders McIlquham-Schmidt Strategic Planning and Corporate Performance An integrative research review and a meta-analysis of the strategic planning and corporate performance literature from 1956 to 2003
- 14. Jens Geersbro The TDF – PMI Case Making Sense of the Dynamics of Business Relationships and Networks
- 15 Mette Andersen Corporate Social Responsibility in Global Supply Chains Understanding the uniqueness of firm behaviour
- 16. Eva Boxenbaum Institutional Genesis: Micro – Dynamic Foundations of Institutional Change
- 17. Peter Lund-Thomsen Capacity Development, Environmental Justice NGOs, and Governance: The Case of South Africa
- 18. Signe Jarlov Konstruktioner af offentlig ledelse
- 19. Lars Stæhr Jensen Vocabulary Knowledge and Listening Comprehension in English as a Foreign Language

An empirical study employing data elicited from Danish EFL learners

- 20. Christian Nielsen Essays on Business Reporting Production and consumption of strategic information in the market for information
- 21. Marianne Thejls Fischer Egos and Ethics of Management Consultants
- 22. Annie Bekke Kjær Performance management i Procesinnovation – belyst i et social-konstruktivistisk perspektiv
- 23. Suzanne Dee Pedersen GENTAGELSENS METAMORFOSE Om organisering af den kreative gøren i den kunstneriske arbejdspraksis
- 24. Benedikte Dorte Rosenbrink Revenue Management Økonomiske, konkurrencemæssige & organisatoriske konsekvenser
- 25. Thomas Riise Johansen Written Accounts and Verbal Accounts The Danish Case of Accounting and Accountability to Employees
- 26. Ann Fogelgren-Pedersen The Mobile Internet: Pioneering Users' Adoption Decisions
- 27. Birgitte Rasmussen Ledelse i fællesskab – de tillidsvalgtes fornyende rolle
- 28. Gitte Thit Nielsen *Remerger* skabende ledelseskræfter i fusion og opkøb
- 29. Carmine Gioia A MICROECONOMETRIC ANALYSIS OF MERGERS AND ACQUISITIONS

- 30. Ole Hinz Den effektive forandringsleder: pilot, pædagog eller politiker? Et studie i arbejdslederes meningstilskrivninger i forbindelse med vellykket gennemførelse af ledelsesinitierede forandringsprojekter
- 31. Kjell-Åge Gotvassli *Et praksisbasert perspektiv på dynami ske læringsnettverk i toppidretten* Norsk ph.d., ej til salg gennem Samfundslitteratur
- 32. Henriette Langstrup Nielsen Linking Healthcare An inquiry into the changing performances of web-based technology for asthma monitoring
- 33. Karin Tweddell Levinsen Virtuel Uddannelsespraksis Master i IKT og Læring – et casestudie i hvordan proaktiv proceshåndtering kan forbedre praksis i virtuelle læringsmiljøer
- 34. Anika Liversage Finding a Path Labour Market Life Stories of Immigrant Professionals
- 35. Kasper Elmquist Jørgensen Studier i samspillet mellem stat og erhvervsliv i Danmark under 1. verdenskrig
- 36. Finn Janning A DIFFERENT STORY Seduction, Conquest and Discovery
- 37. Patricia Ann Plackett Strategic Management of the Radical Innovation Process Leveraging Social Capital for Market Uncertainty Management

2006

1. Christian Vintergaard Early Phases of Corporate Venturing

- 2. Niels Rom-Poulsen Essays in Computational Finance
- 3. Tina Brandt Husman Organisational Capabilities, Competitive Advantage & Project-Based Organisations The Case of Advertising and Creative Good Production
- Mette Rosenkrands Johansen
 Practice at the top
 how top managers mobilise and use
 non-financial performance measures
- 5. Eva Parum Corporate governance som strategisk kommunikations- og ledelsesværktøj
- 6. Susan Aagaard Petersen Culture's Influence on Performance Management: The Case of a Danish Company in China
- 7. Thomas Nicolai Pedersen The Discursive Constitution of Organizational Governance – Between unity and differentiation The Case of the governance of environmental risks by World Bank environmental staff
- 8. Cynthia Selin Volatile Visions: Transactons in Anticipatory Knowledge
- 9. Jesper Banghøj Financial Accounting Information and Compensation in Danish Companies
- 10. Mikkel Lucas Overby Strategic Alliances in Emerging High-Tech Markets: What's the Difference and does it Matter?
- 11. Tine Aage External Information Acquisition of Industrial Districts and the Impact of Different Knowledge Creation Dimensions

A case study of the Fashion and Design Branch of the Industrial District of Montebelluna, NE Italy

- 12. Mikkel Flyverbom Making the Global Information Society Governable On the Governmentality of Multi-Stakeholder Networks
- 13. Anette Grønning Personen bag Tilstedevær i e-mail som interaktionsform mellem kunde og medarbejder i dansk forsikringskontekst
- 14. Jørn Helder One Company – One Language? The NN-case
- 15. Lars Bjerregaard Mikkelsen Differing perceptions of customer value Development and application of a tool for mapping perceptions of customer value at both ends of customer-supplier dyads in industrial markets
- 16. Lise Granerud Exploring Learning Technological learning within small manufacturers in South Africa
- 17. Esben Rahbek Pedersen Between Hopes and Realities: Reflections on the Promises and Practices of Corporate Social Responsibility (CSR)
- Ramona Samson The Cultural Integration Model and European Transformation. The Case of Romania

2007

1. Jakob Vestergaard Discipline in The Global Economy Panopticism and the Post-Washington Consensus

- 2. Heidi Lund Hansen Spaces for learning and working A qualitative study of change of work, management, vehicles of power and social practices in open offices
- 3. Sudhanshu Rai Exploring the internal dynamics of software development teams during user analysis A tension enabled Institutionalization Model; "Where process becomes the objective"
- 4. Norsk ph.d. Ej til salg gennem Samfundslitteratur
- 5. Serden Ozcan *EXPLORING HETEROGENEITY IN ORGANIZATIONAL ACTIONS AND OUTCOMES A Behavioural Perspective*
- 6. Kim Sundtoft Hald Inter-organizational Performance Measurement and Management in Action

 An Ethnography on the Construction of Management, Identity and Relationships
- 7. Tobias Lindeberg Evaluative Technologies Quality and the Multiplicity of Performance
- 8. Merete Wedell-Wedellsborg Den globale soldat Identitetsdannelse og identitetsledelse i multinationale militære organisationer
- Lars Frederiksen Open Innovation Business Models Innovation in firm-hosted online user communities and inter-firm project ventures in the music industry – A collection of essays
- 10. Jonas Gabrielsen Retorisk toposlære – fra statisk 'sted' til persuasiv aktivitet

- Christian Moldt-Jørgensen Fra meningsløs til meningsfuld evaluering. Anvendelsen af studentertilfredshedsmålinger på de korte og mellemlange videregående uddannelser set fra et psykodynamisk systemperspektiv
- 12. Ping Gao Extending the application of actor-network theory Cases of innovation in the telecommunications industry
- Peter Mejlby Frihed og fængsel, en del af den samme drøm? Et phronetisk baseret casestudie af frigørelsens og kontrollens sameksistens i værdibaseret ledelse!
- 14. Kristina Birch Statistical Modelling in Marketing
- 15. Signe Poulsen Sense and sensibility: The language of emotional appeals in insurance marketing
- 16. Anders Bjerre Trolle Essays on derivatives pricing and dynamic asset allocation
- 17. Peter Feldhütter Empirical Studies of Bond and Credit Markets
- 18. Jens Henrik Eggert Christensen Default and Recovery Risk Modeling and Estimation
- Maria Theresa Larsen Academic Enterprise: A New Mission for Universities or a Contradiction in Terms? Four papers on the long-term implications of increasing industry involvement and commercialization in academia

- 20. Morten Wellendorf Postimplementering af teknologi i den offentlige forvaltning Analyser af en organisations kontinuerlige arbejde med informationsteknologi
- 21. Ekaterina Mhaanna Concept Relations for Terminological Process Analysis
- 22. Stefan Ring Thorbjørnsen Forsvaret i forandring Et studie i officerers kapabiliteter under påvirkning af omverdenens forandringspres mod øget styring og læring
- 23. Christa Breum Amhøj Det selvskabte medlemskab om managementstaten, dens styringsteknologier og indbyggere
- 24. Karoline Bromose Between Technological Turbulence and Operational Stability

 An empirical case study of corporate venturing in TDC
- 25. Susanne Justesen Navigating the Paradoxes of Diversity in Innovation Practice

 A Longitudinal study of six very different innovation processes – in practice
- 26. Luise Noring Henler Conceptualising successful supply chain partnerships

 Viewing supply chain partnerships from an organisational culture perspective
- 27. Mark Mau Kampen om telefonen Det danske telefonvæsen under den tyske besættelse 1940-45
- 28. Jakob Halskov The semiautomatic expansion of existing terminological ontologies using knowledge patterns discovered

on the WWW – an implementation and evaluation

- 29. Gergana Koleva European Policy Instruments Beyond Networks and Structure: The Innovative Medicines Initiative
- 30. Christian Geisler Asmussen Global Strategy and International Diversity: A Double-Edged Sword?
- 31. Christina Holm-Petersen Stolthed og fordom Kultur- og identitetsarbejde ved skabelsen af en ny sengeafdeling gennem fusion
- 32. Hans Peter Olsen Hybrid Governance of Standardized States Causes and Contours of the Global Regulation of Government Auditing
- 33. Lars Bøge Sørensen Risk Management in the Supply Chain
- 34. Peter Aagaard Det unikkes dynamikker De institutionelle mulighedsbetingelser bag den individuelle udforskning i professionelt og frivilligt arbejde
- 35. Yun Mi Antorini Brand Community Innovation An Intrinsic Case Study of the Adult Fans of LEGO Community
- 36. Joachim Lynggaard Boll Labor Related Corporate Social Performance in Denmark Organizational and Institutional Perspectives

2008

- 1. Frederik Christian Vinten Essays on Private Equity
- 2. Jesper Clement Visual Influence of Packaging Design on In-Store Buying Decisions

- Marius Brostrøm Kousgaard Tid til kvalitetsmåling?

 Studier af indrulleringsprocesser i forbindelse med introduktionen af kliniske kvalitetsdatabaser i speciallægepraksissektoren
- 4. Irene Skovgaard Smith Management Consulting in Action Value creation and ambiguity in client-consultant relations
- 5. Anders Rom Management accounting and integrated information systems How to exploit the potential for management accounting of information technology
- 6. Marina Candi Aesthetic Design as an Element of Service Innovation in New Technologybased Firms
- Morten Schnack Teknologi og tværfaglighed

 – en analyse af diskussionen omkring indførelse af EPJ på en hospitalsafdeling
- 8. Helene Balslev Clausen Juntos pero no revueltos – un estudio sobre emigrantes norteamericanos en un pueblo mexicano
- 9. Lise Justesen Kunsten at skrive revisionsrapporter. En beretning om forvaltningsrevisionens beretninger
- 10. Michael E. Hansen The politics of corporate responsibility: CSR and the governance of child labor and core labor rights in the 1990s
- 11. Anne Roepstorff Holdning for handling – en etnologisk undersøgelse af Virksomheders Sociale Ansvar/CSR

- 12. Claus Bajlum Essays on Credit Risk and Credit Derivatives
- 13. Anders Bojesen The Performative Power of Competence – an Inquiry into Subjectivity and Social Technologies at Work
- 14. Satu Reijonen Green and Fragile A Study on Markets and the Natural Environment
- 15. Ilduara Busta Corporate Governance in Banking A European Study
- 16. Kristian Anders Hvass A Boolean Analysis Predicting Industry Change: Innovation, Imitation & Business Models The Winning Hybrid: A case study of isomorphism in the airline industry
- 17. Trine Paludan De uvidende og de udviklingsparate Identitet som mulighed og restriktion blandt fabriksarbejdere på det aftayloriserede fabriksgulv
- 18. Kristian Jakobsen Foreign market entry in transition economies: Entry timing and mode choice
- 19. Jakob Elming Syntactic reordering in statistical machine translation
- 20. Lars Brømsøe Termansen Regional Computable General Equilibrium Models for Denmark Three papers laying the foundation for regional CGE models with agglomeration characteristics
- 21. Mia Reinholt The Motivational Foundations of Knowledge Sharing

- 22. Frederikke Krogh-Meibom The Co-Evolution of Institutions and Technology – A Neo-Institutional Understanding of Change Processes within the Business Press – the Case Study of Financial Times
- 23. Peter D. Ørberg Jensen OFFSHORING OF ADVANCED AND HIGH-VALUE TECHNICAL SERVICES: ANTECEDENTS, PROCESS DYNAMICS AND FIRMLEVEL IMPACTS
- 24. Pham Thi Song Hanh Functional Upgrading, Relational Capability and Export Performance of Vietnamese Wood Furniture Producers
- 25. Mads Vangkilde Why wait? An Exploration of first-mover advantages among Danish e-grocers through a resource perspective
- 26. Hubert Buch-Hansen Rethinking the History of European Level Merger Control A Critical Political Economy Perspective

2009

- 1. Vivian Lindhardsen From Independent Ratings to Communal Ratings: A Study of CWA Raters' Decision-Making Behaviours
- 2. Guðrið Weihe Public-Private Partnerships: Meaning and Practice
- 3. Chris Nøkkentved Enabling Supply Networks with Collaborative Information Infrastructures An Empirical Investigation of Business Model Innovation in Supplier Relationship Management
- 4. Sara Louise Muhr Wound, Interrupted – On the Vulnerability of Diversity Management

- 5. Christine Sestoft Forbrugeradfærd i et Stats- og Livsformsteoretisk perspektiv
- 6. Michael Pedersen Tune in, Breakdown, and Reboot: On the production of the stress-fit selfmanaging employee
- Salla Lutz
 Position and Reposition in Networks

 Exemplified by the Transformation of the Danish Pine Furniture Manufacturers
- 8. Jens Forssbæck Essays on market discipline in commercial and central banking
- 9. Tine Murphy Sense from Silence – A Basis for Organised Action How do Sensemaking Processes with Minimal Sharing Relate to the Reproduction of Organised Action?
- 10. Sara Malou Strandvad Inspirations for a new sociology of art: A sociomaterial study of development processes in the Danish film industry
- Nicolaas Mouton On the evolution of social scientific metaphors: A cognitive-historical enquiry into the divergent trajectories of the idea that collective entities – states and societies, cities and corporations – are biological organisms.
- 12. Lars Andreas Knutsen Mobile Data Services: Shaping of user engagements
- 13. Nikolaos Theodoros Korfiatis Information Exchange and Behavior A Multi-method Inquiry on Online Communities

- 14. Jens Albæk Forestillinger om kvalitet og tværfaglighed på sygehuse – skabelse af forestillinger i læge- og plejegrupperne angående relevans af nye idéer om kvalitetsudvikling gennem tolkningsprocesser
- 15. Maja Lotz The Business of Co-Creation – and the Co-Creation of Business
- 16. Gitte P. Jakobsen Narrative Construction of Leader Identity in a Leader Development Program Context
- 17. Dorte Hermansen "Living the brand" som en brandorienteret dialogisk praxis: Om udvikling af medarbejdernes brandorienterede dømmekraft
- 18. Aseem Kinra Supply Chain (logistics) Environmental Complexity
- 19. Michael Nørager How to manage SMEs through the transformation from non innovative to innovative?
- 20. Kristin Wallevik Corporate Governance in Family Firms The Norwegian Maritime Sector
- 21. Bo Hansen Hansen Beyond the Process Enriching Software Process Improvement with Knowledge Management
- 22. Annemette Skot-Hansen Franske adjektivisk afledte adverbier, der tager præpositionssyntagmer indledt med præpositionen à som argumenter En valensgrammatisk undersøgelse
- 23. Line Gry Knudsen Collaborative R&D Capabilities In Search of Micro-Foundations

- 24. Christian Scheuer Employers meet employees Essays on sorting and globalization
- 25. Rasmus Johnsen The Great Health of Melancholy A Study of the Pathologies of Performativity
- 26. Ha Thi Van Pham Internationalization, Competitiveness Enhancement and Export Performance of Emerging Market Firms: Evidence from Vietnam
- 27. Henriette Balieu Kontrolbegrebets betydning for kausativalternationen i spansk En kognitiv-typologisk analyse

2010

- 1. Yen Tran Organizing Innovationin Turbulent Fashion Market Four papers on how fashion firms create and appropriate innovation value
- 2. Anders Raastrup Kristensen Metaphysical Labour Flexibility, Performance and Commitment in Work-Life Management
- 3. Margrét Sigrún Sigurdardottir Dependently independent Co-existence of institutional logics in the recorded music industry
- Ásta Dis Óladóttir Internationalization from a small domestic base: An empirical analysis of Economics and Management
- 5. Christine Secher E-deltagelse i praksis – politikernes og forvaltningens medkonstruktion og konsekvenserne heraf
- 6. Marianne Stang Våland What we talk about when we talk about space:

End User Participation between Processes of Organizational and Architectural Design

- 7. Rex Degnegaard Strategic Change Management Change Management Challenges in the Danish Police Reform
- Ulrik Schultz Brix Værdi i rekruttering – den sikre beslutning En pragmatisk analyse af perception og synliggørelse af værdi i rekrutterings- og udvælgelsesarbejdet
 - Jan Ole Similä Kontraktsledelse Relasjonen mellom virksomhetsledelse og kontraktshåndtering, belyst via fire norske virksomheter

9.

- 10. Susanne Boch Waldorff Emerging Organizations: In between local translation, institutional logics and discourse
- 11. Brian Kane Performance Talk Next Generation Management of Organizational Performance
- 12. Lars Ohnemus Brand Thrust: Strategic Branding and Shareholder Value An Empirical Reconciliation of two Critical Concepts
- 13. Jesper Schlamovitz Håndtering af usikkerhed i film- og byggeprojekter
- Tommy Moesby-Jensen Det faktiske livs forbindtlighed Førsokratisk informeret, ny-aristotelisk ήθος-tænkning hos Martin Heidegger
- 15. Christian Fich Two Nations Divided by Common Values French National Habitus and the Rejection of American Power
- 16. Peter Beyer Processer, sammenhængskraft og fleksibilitet Et empirisk casestudie af omstillingsforløb i fire virksomheder
- 17. Adam Buchhorn Markets of Good Intentions Constructing and Organizing Biogas Markets Amid Fragility and Controversy
- 18. Cecilie K. Moesby-Jensen Social læring og fælles praksis Et mixed method studie, der belyser læringskonsekvenser af et lederkursus for et praksisfællesskab af offentlige mellemledere
- 19. Heidi Boye Fødevarer og sundhed i senmodernismen
 – En indsigt i hyggefænomenet og de relaterede fødevarepraksisser
- 20. Kristine Munkgård Pedersen Flygtige forbindelser og midlertidige mobiliseringer Om kulturel produktion på Roskilde Festival
- 21. Oliver Jacob Weber Causes of Intercompany Harmony in Business Markets – An Empirical Investigation from a Dyad Perspective
- 22. Susanne Ekman Authority and Autonomy Paradoxes of Modern Knowledge Work
- 23. Anette Frey Larsen
 Kvalitetsledelse på danske hospitaler
 Ledelsernes indflydelse på introduktion og vedligeholdelse af kvalitetsstrategier i det danske sundhedsvæsen
- 24. Toyoko Sato Performativity and Discourse: Japanese Advertisements on the Aesthetic Education of Desire

- 25. Kenneth Brinch Jensen Identifying the Last Planner System Lean management in the construction industry
- 26. Javier Busquets Orchestrating Network Behavior for Innovation
- 27. Luke Patey The Power of Resistance: India's National Oil Company and International Activism in Sudan
- 28. Mette Vedel Value Creation in Triadic Business Relationships. Interaction, Interconnection and Position
- 29. Kristian Tørning Knowledge Management Systems in Practice – A Work Place Study
- 30. Qingxin Shi An Empirical Study of Thinking Aloud Usability Testing from a Cultural Perspective
- 31. Tanja Juul Christiansen Corporate blogging: Medarbejderes kommunikative handlekraft
- Malgorzata Ciesielska Hybrid Organisations.
 A study of the Open Source – business setting
- 33. Jens Dick-Nielsen Three Essays on Corporate Bond Market Liquidity
- 34. Sabrina Speiermann Modstandens Politik Kampagnestyring i Velfærdsstaten. En diskussion af trafikkampagners styringspotentiale
- 35. Julie Uldam Fickle Commitment. Fostering political engagement in 'the flighty world of online activism'

- 36. Annegrete Juul Nielsen Traveling technologies and transformations in health care
- 37. Athur Mühlen-Schulte Organising Development Power and Organisational Reform in the United Nations Development Programme
- 38. Louise Rygaard Jonas Branding på butiksgulvet Et case-studie af kultur- og identitetsarbejdet i Kvickly

- 1. Stefan Fraenkel Key Success Factors for Sales Force Readiness during New Product Launch A Study of Product Launches in the Swedish Pharmaceutical Industry
- 2. Christian Plesner Rossing International Transfer Pricing in Theory and Practice
- Tobias Dam Hede
 Samtalekunst og ledelsesdisciplin
 en analyse af coachingsdiskursens genealogi og governmentality
- 4. Kim Pettersson Essays on Audit Quality, Auditor Choice, and Equity Valuation
- 5. Henrik Merkelsen The expert-lay controversy in risk research and management. Effects of institutional distances. Studies of risk definitions, perceptions, management and communication
- 6. Simon S. Torp Employee Stock Ownership: Effect on Strategic Management and Performance
- 7. Mie Harder Internal Antecedents of Management Innovation

- 8. Ole Helby Petersen Public-Private Partnerships: Policy and Regulation – With Comparative and Multi-level Case Studies from Denmark and Ireland
- 9. Morten Krogh Petersen 'Good' Outcomes. Handling Multiplicity in Government Communication
- 10. Kristian Tangsgaard Hvelplund Allocation of cognitive resources in translation - an eye-tracking and keylogging study
- 11. Moshe Yonatany The Internationalization Process of Digital Service Providers
- 12. Anne Vestergaard Distance and Suffering Humanitarian Discourse in the age of Mediatization
- 13. Thorsten Mikkelsen Personligsheds indflydelse på forretningsrelationer
- 14. Jane Thostrup Jagd Hvorfor fortsætter fusionsbølgen udover "the tipping point"? – en empirisk analyse af information og kognitioner om fusioner
- 15. Gregory Gimpel Value-driven Adoption and Consumption of Technology: Understanding Technology Decision Making
- 16. Thomas Stengade Sønderskov Den nye mulighed Social innovation i en forretningsmæssig kontekst
- 17. Jeppe Christoffersen Donor supported strategic alliances in developing countries
- 18. Vibeke Vad Baunsgaard Dominant Ideological Modes of Rationality: Cross functional

integration in the process of product innovation

- 19. Throstur Olaf Sigurjonsson Governance Failure and Icelands's Financial Collapse
- 20. Allan Sall Tang Andersen Essays on the modeling of risks in interest-rate and inflation markets
- 21. Heidi Tscherning Mobile Devices in Social Contexts
- 22. Birgitte Gorm Hansen Adapting in the Knowledge Economy Lateral Strategies for Scientists and Those Who Study Them
- 23. Kristina Vaarst Andersen Optimal Levels of Embeddedness The Contingent Value of Networked Collaboration
- 24. Justine Grønbæk Pors Noisy Management A History of Danish School Governing from 1970-2010
- 25. Stefan Linder Micro-foundations of Strategic Entrepreneurship Essays on Autonomous Strategic Action 4.
- 26. Xin Li Toward an Integrative Framework of National Competitiveness An application to China
- 27. Rune Thorbjørn Clausen Værdifuld arkitektur Et eksplorativt studie af bygningers rolle i virksomheders værdiskabelse
- 28. Monica Viken Markedsundersøkelser som bevis i varemerke- og markedsføringsrett
- 29. Christian Wymann Tattooing The Economic and Artistic Constitution of a Social Phenomenon

- 30. Sanne Frandsen Productive Incoherence A Case Study of Branding and Identity Struggles in a Low-Prestige Organization
- 31. Mads Stenbo Nielsen Essays on Correlation Modelling
- 32. Ivan Häuser Følelse og sprog Etablering af en ekspressiv kategori, eksemplificeret på russisk
- 33. Sebastian Schwenen Security of Supply in Electricity Markets

- Peter Holm Andreasen The Dynamics of Procurement Management - A Complexity Approach
- 2. Martin Haulrich Data-Driven Bitext Dependency Parsing and Alignment
- 3. Line Kirkegaard Konsulenten i den anden nat En undersøgelse af det intense arbejdsliv
 - Tonny Stenheim Decision usefulness of goodwill under IFRS
- 5. Morten Lind Larsen Produktivitet, vækst og velfærd Industrirådet og efterkrigstidens Danmark 1945 - 1958
- 6. Petter Berg Cartel Damages and Cost Asymmetries
- 7. Lynn Kahle Experiential Discourse in Marketing A methodical inquiry into practice and theory
- 8. Anne Roelsgaard Obling Management of Emotions in Accelerated Medical Relationships

- 9. Thomas Frandsen Managing Modularity of Service Processes Architecture
- 10. Carina Christine Skovmøller CSR som noget særligt Et casestudie om styring og meningsskabelse i relation til CSR ud fra en intern optik
- Michael Tell Fradragsbeskæring af selskabers finansieringsudgifter En skatteretlig analyse af SEL §§ 11, 11B og 11C
- 12. Morten Holm *Customer Profitability Measurement Models Their Merits and Sophistication across Contexts*
- 13. Katja Joo Dyppel Beskatning af derivater En analyse af dansk skatteret
- 14. Esben Anton Schultz Essays in Labor Economics Evidence from Danish Micro Data
- 15. Carina Risvig Hansen "Contracts not covered, or not fully covered, by the Public Sector Directive"
- 16. Anja Svejgaard Pors Iværksættelse af kommunikation - patientfigurer i hospitalets strategiske kommunikation
- 17. Frans Bévort Making sense of management with logics An ethnographic study of accountants who become managers
- 18. René Kallestrup The Dynamics of Bank and Sovereign Credit Risk
- 19. Brett Crawford Revisiting the Phenomenon of Interests in Organizational Institutionalism The Case of U.S. Chambers of Commerce

- 20. Mario Daniele Amore Essays on Empirical Corporate Finance
- 21. Arne Stjernholm Madsen The evolution of innovation strategy Studied in the context of medical device activities at the pharmaceutical company Novo Nordisk A/S in the period 1980-2008
- 22. Jacob Holm Hansen Is Social Integration Necessary for Corporate Branding? A study of corporate branding strategies at Novo Nordisk
- 23. Stuart Webber Corporate Profit Shifting and the Multinational Enterprise
- 24. Helene Ratner Promises of Reflexivity Managing and Researching Inclusive Schools
- 25. Therese Strand The Owners and the Power: Insights from Annual General Meetings
- 26. Robert Gavin Strand In Praise of Corporate Social Responsibility Bureaucracy
- 27. Nina Sormunen Auditor's going-concern reporting Reporting decision and content of the report
- 28. John Bang Mathiasen Learning within a product development working practice:
 - an understanding anchored in pragmatism
 - Philip Holst Riis Understanding Role-Oriented Enterprise Systems: From Vendors to Customers

29.

30. Marie Lisa Dacanay Social Enterprises and the Poor Enhancing Social Entrepreneurship and Stakeholder Theory

- 31. Fumiko Kano Glückstad Bridging Remote Cultures: Cross-lingual concept mapping based on the information receiver's prior-knowledge
- 32. Henrik Barslund Fosse Empirical Essays in International Trade
- 33. Peter Alexander Albrecht Foundational hybridity and its reproduction Security sector reform in Sierra Leone
- 34. Maja Rosenstock CSR - hvor svært kan det være? Kulturanalytisk casestudie om udfordringer og dilemmaer med at forankre Coops CSR-strategi
- 35. Jeanette Rasmussen Tweens, medier og forbrug Et studie af 10-12 årige danske børns brug af internettet, opfattelse og forståelse af markedsføring og forbrug
- 36. Ib Tunby Gulbrandsen 'This page is not intended for a US Audience' A five-act spectacle on online communication, collaboration & organization.
- 37. Kasper Aalling Teilmann Interactive Approaches to Rural Development
- 38. Mette Mogensen The Organization(s) of Well-being and Productivity (Re)assembling work in the Danish Post
- 39. Søren Friis Møller
 From Disinterestedness to Engagement 6.
 Towards Relational Leadership In the Cultural Sector
- 40. Nico Peter Berhausen Management Control, Innovation and Strategic Objectives – Interactions and Convergence in Product Development Networks

- 41. Balder Onarheim Creativity under Constraints Creativity as Balancing 'Constrainedness'
- 42. Haoyong Zhou Essays on Family Firms
- 43. Elisabeth Naima Mikkelsen Making sense of organisational conflict An empirical study of enacted sensemaking in everyday conflict at work

- 1. Jacob Lyngsie Entrepreneurship in an Organizational Context
- 2. Signe Groth-Brodersen Fra ledelse til selvet En socialpsykologisk analyse af forholdet imellem selvledelse, ledelse og stress i det moderne arbejdsliv
- 3. Nis Høyrup Christensen Shaping Markets: A Neoinstitutional Analysis of the Emerging Organizational Field of Renewable Energy in China
- 4. Christian Edelvold Berg As a matter of size THE IMPORTANCE OF CRITICAL MASS AND THE CONSEQUENCES OF SCARCITY FOR TELEVISION MARKETS
- 5. Christine D. Isakson Coworker Influence and Labor Mobility Essays on Turnover, Entrepreneurship and Location Choice in the Danish Maritime Industry
 - Niels Joseph Jerne Lennon Accounting Qualities in Practice Rhizomatic stories of representational faithfulness, decision making and control
- 7. Shannon O'Donnell Making Ensemble Possible How special groups organize for collaborative creativity in conditions of spatial variability and distance

- 8. Robert W. D. Veitch Access Decisions in a Partly-Digital World Comparing Digital Piracy and Legal Modes for Film and Music
- 9. Marie Mathiesen Making Strategy Work An Organizational Ethnography
- 10. Arisa Shollo The role of business intelligence in organizational decision-making
- 11. Mia Kaspersen The construction of social and environmental reporting
- 12. Marcus Møller Larsen The organizational design of offshoring
- 13. Mette Ohm Rørdam EU Law on Food Naming The prohibition against misleading names in an internal market context
- 14. Hans Peter Rasmussen GIV EN GED! Kan giver-idealtyper forklare støtte til velgørenhed og understøtte relationsopbygning?
- 15. Ruben Schachtenhaufen Fonetisk reduktion i dansk
- 16. Peter Koerver Schmidt Dansk CFC-beskatning I et internationalt og komparativt perspektiv
- 17.Morten Froholdt26.Strategi i den offentlige sektorEn kortlægning af styringsmæssigkontekst, strategisk tilgang, samtanvendte redskaber og teknologier forudvalgte danske statslige styrelser27.
- Annette Camilla Sjørup Cognitive effort in metaphor translation An eye-tracking and key-logging study 28.

- 19. Tamara Stucchi The Internationalization of Emerging Market Firms: A Context-Specific Study
- 20. Thomas Lopdrup-Hjorth "Let's Go Outside": The Value of Co-Creation
- 21. Ana Alačovska Genre and Autonomy in Cultural Production The case of travel guidebook production
- 22. Marius Gudmand-Høyer Stemningssindssygdommenes historie i det 19. århundrede Omtydningen af melankolien og manien som bipolære stemningslidelser i dansk sammenhæng under hensyn til dannelsen af det moderne følelseslivs relative autonomi. En problematiserings- og erfaringsanalytisk undersøgelse
- 23. Lichen Alex Yu Fabricating an S&OP Process Circulating References and Matters of Concern
- 24. Esben Alfort The Expression of a Need Understanding search
- 25. Trine Pallesen Assembling Markets for Wind Power An Inquiry into the Making of Market Devices
- 26. Anders Koed Madsen Web-Visions Repurposing digital traces to organize social attention
 - Lærke Højgaard Christiansen BREWING ORGANIZATIONAL RESPONSES TO INSTITUTIONAL LOGICS

Tommy Kjær Lassen EGENTLIG SELVLEDELSE En ledelsesfilosofisk afhandling om selvledelsens paradoksale dynamik og eksistentielle engagement

- 29. Morten Rossing Local Adaption and Meaning Creation in Performance Appraisal
- 30. Søren Obed Madsen Lederen som oversætter Et oversættelsesteoretisk perspektiv på strategisk arbejde
- 31. Thomas Høgenhaven Open Government Communities Does Design Affect Participation?
- 32. Kirstine Zinck Pedersen Failsafe Organizing? A Pragmatic Stance on Patient Safety
- 33. Anne Petersen Hverdagslogikker i psykiatrisk arbejde En institutionsetnografisk undersøgelse af hverdagen i psykiatriske organisationer
- 34. Didde Maria Humle Fortællinger om arbejde
- 35. Mark Holst-Mikkelsen Strategieksekvering i praksis – barrierer og muligheder!
- 36. Malek Maalouf Sustaining lean Strategies for dealing with organizational paradoxes
- 37. Nicolaj Tofte Brenneche Systemic Innovation In The Making The Social Productivity of Cartographic Crisis and Transitions in the Case of SEEIT
- Morten Gylling The Structure of Discourse A Corpus-Based Cross-Linguistic Study
- 39. Binzhang YANG
 Urban Green Spaces for Quality Life
 Case Study: the landscape
 architecture for people in Copenhagen

- 40. Michael Friis Pedersen Finance and Organization: The Implications for Whole Farm Risk Management
- 41. Even Fallan Issues on supply and demand for environmental accounting information
- 42. Ather Nawaz Website user experience A cross-cultural study of the relation between users' cognitive style, context of use, and information architecture of local websites
- 43. Karin Beukel The Determinants for Creating Valuable Inventions
- 44. Arjan Markus External Knowledge Sourcing and Firm Innovation Essays on the Micro-Foundations of Firms' Search for Innovation

- 1. Solon Moreira Four Essays on Technology Licensing and Firm Innovation
- 2. Karin Strzeletz Ivertsen Partnership Drift in Innovation Processes A study of the Think City electric car development
- 3. Kathrine Hoffmann Pii Responsibility Flows in Patient-centred Prevention
- 4. Jane Bjørn Vedel Managing Strategic Research An empirical analysis of science-industry collaboration in a pharmaceutical company
- 5. Martin Gylling Processuel strategi i organisationer Monografi om dobbeltheden i tænkning af strategi, dels som vidensfelt i organisationsteori, dels som kunstnerisk tilgang til at skabe i erhvervsmæssig innovation

- Linne Marie Lauesen Corporate Social Responsibility in the Water Sector: How Material Practices and their Symbolic and Physical Meanings Form a Colonising Logic
- 7. Maggie Qiuzhu Mei LEARNING TO INNOVATE: The role of ambidexterity, standard, and decision process
- 8. Inger Høedt-Rasmussen Developing Identity for Lawyers Towards Sustainable Lawyering
- 9. Sebastian Fux Essays on Return Predictability and Term Structure Modelling
- 10. Thorbjørn N. M. Lund-Poulsen Essays on Value Based Management
- 11. Oana Brindusa Albu Transparency in Organizing: A Performative Approach
- 12. Lena Olaison Entrepreneurship at the limits
- 13. Hanne Sørum DRESSED FOR WEB SUCCESS? An Empirical Study of Website Quality in the Public Sector
- 14. Lasse Folke Henriksen Knowing networks How experts shape transnational governance
- 15. Maria Halbinger Entrepreneurial Individuals Empirical Investigations into Entrepreneurial Activities of Hackers and Makers
- 16. Robert Spliid Kapitalfondenes metoder og kompetencer

- 17. Christiane Stelling Public-private partnerships & the need, development and management of trusting A processual and embedded exploration
- 18. Marta Gasparin Management of design as a translation process
- 19. Kåre Moberg Assessing the Impact of Entrepreneurship Education From ABC to PhD
- 20. Alexander Cole Distant neighbors Collective learning beyond the cluster
- 21. Martin Møller Boje Rasmussen Is Competitiveness a Question of Being Alike? How the United Kingdom, Germany and Denmark Came to Compete through their Knowledge Regimes from 1993 to 2007
- 22. Anders Ravn Sørensen Studies in central bank legitimacy, currency and national identity Four cases from Danish monetary history
- 23. Nina Bellak Can Language be Managed in International Business? Insights into Language Choice from a Case Study of Danish and Austrian Multinational Corporations (MNCs)
- 24. Rikke Kristine Nielsen Global Mindset as Managerial Meta-competence and Organizational Capability: Boundary-crossing Leadership Cooperation in the MNC The Case of 'Group Mindset' in Solar A/S.
- 25. Rasmus Koss Hartmann User Innovation inside government Towards a critically performative foundation for inquiry

- 26. Kristian Gylling Olesen Flertydig og emergerende ledelse i folkeskolen Et aktør-netværksteoretisk ledelsesstudie af politiske evalueringsreformers betydning for ledelse i den danske folkeskole
- 27. Troels Riis Larsen Kampen om Danmarks omdømme 1945-2010 Omdømmearbejde og omdømmepolitik
- 28. Klaus Majgaard Jagten på autenticitet i offentlig styring
- 29. Ming Hua Li Institutional Transition and Organizational Diversity: Differentiated internationalization strategies of emerging market state-owned enterprises
- 30. Sofie Blinkenberg Federspiel IT, organisation og digitalisering: Institutionelt arbejde i den kommunale digitaliseringsproces
- 31. Elvi Weinreich
 Hvilke offentlige ledere er der brug for
 når velfærdstænkningen flytter sig
 er Diplomuddannelsens lederprofil
 svaret?
- 32. Ellen Mølgaard Korsager
 Self-conception and image of context in the growth of the firm
 – A Penrosian History of Fiberline Composites
- 33. Else Skjold The Daily Selection
- 34. Marie Louise Conradsen The Cancer Centre That Never Was The Organisation of Danish Cancer Research 1949-1992
- 35. Virgilio Failla Three Essays on the Dynamics of Entrepreneurs in the Labor Market

- 36. Nicky Nedergaard Brand-Based Innovation Relational Perspectives on Brand Logics and Design Innovation Strategies and Implementation
- 37. Mads Gjedsted Nielsen Essays in Real Estate Finance
- 38. Kristin Martina Brandl Process Perspectives on Service Offshoring
- 39. Mia Rosa Koss Hartmann In the gray zone With police in making space for creativity
- 40. Karen Ingerslev Healthcare Innovation under The Microscope Framing Boundaries of Wicked Problems
- 41. Tim Neerup Themsen Risk Management in large Danish public capital investment programmes

- 1. Jakob Ion Wille Film som design Design af levende billeder i film og tv-serier
- 2. Christiane Mossin Interzones of Law and Metaphysics Hierarchies, Logics and Foundations of Social Order seen through the Prism of EU Social Rights
- 3. Thomas Tøth TRUSTWORTHINESS: ENABLING GLOBAL COLLABORATION An Ethnographic Study of Trust, Distance, Control, Culture and Boundary Spanning within Offshore Outsourcing of IT Services
- 4. Steven Højlund Evaluation Use in Evaluation Systems – The Case of the European Commission

- 5. Julia Kirch Kirkegaard *AMBIGUOUS WINDS OF CHANGE – OR FIGHTING AGAINST WINDMILLS IN CHINESE WIND POWER A CONSTRUCTIVIST INQUIRY INTO CHINA'S PRAGMATICS OF GREEN MARKETISATION MAPPING CONTROVERSIES OVER A POTENTIAL TURN TO QUALITY IN CHINESE WIND POWER*
- 6. Michelle Carol Antero A Multi-case Analysis of the Development of Enterprise Resource Planning Systems (ERP) Business Practices

Morten Friis-Olivarius The Associative Nature of Creativity

- Mathew Abraham
 New Cooperativism:
 A study of emerging producer
 organisations in India
- 8. Stine Hedegaard Sustainability-Focused Identity: Identity work performed to manage, negotiate and resolve barriers and tensions that arise in the process of constructing or ganizational identity in a sustainability context
- 9. Cecilie Glerup Organizing Science in Society – the conduct and justification of resposible research
- 10. Allan Salling Pedersen Implementering af ITIL® IT-governance - når best practice konflikter med kulturen Løsning af implementeringsproblemer gennem anvendelse af kendte CSF i et aktionsforskningsforløb.
- 11. Nihat Misir A Real Options Approach to Determining Power Prices
- 12. Mamdouh Medhat MEASURING AND PRICING THE RISK OF CORPORATE FAILURES

13. Rina Hansen Toward a Digital Strategy for Omnichannel Retailing

- 14. Eva Pallesen In the rhythm of welfare creation A relational processual investigation moving beyond the conceptual horizon of welfare management
- 15. Gouya Harirchi In Search of Opportunities: Three Essays on Global Linkages for Innovation
- 16. Lotte Holck Embedded Diversity: A critical ethnographic study of the structural tensions of organizing diversity
- 17. Jose Daniel Balarezo Learning through Scenario Planning
- 18. Louise Pram Nielsen Knowledge dissemination based on terminological ontologies. Using eye tracking to further user interface design.
- 19. Sofie Dam PUBLIC-PRIVATE PARTNERSHIPS FOR INNOVATION AND SUSTAINABILITY TRANSFORMATION An embedded, comparative case study of municipal waste management in England and Denmark
- 20. Ulrik Hartmyer Christiansen Follwoing the Content of Reported Risk Across the Organization
- 21. Guro Refsum Sanden Language strategies in multinational corporations. A cross-sector study of financial service companies and manufacturing companies.
- 22. Linn Gevoll
 Designing performance management
 for operational level
 A closer look on the role of design
 choices in framing coordination and
 motivation

- 23. Frederik Larsen Objects and Social Actions – on Second-hand Valuation Practices
- 24. Thorhildur Hansdottir Jetzek The Sustainable Value of Open Government Data Uncovering the Generative Mechanisms of Open Data through a Mixed Methods Approach
- 25. Gustav Toppenberg Innovation-based M&A – Technological-Integration Challenges – The Case of Digital-Technology Companies
- 26. Mie Plotnikof Challenges of Collaborative Governance An Organizational Discourse Study of Public Managers' Struggles with Collaboration across the Daycare Area
- 27. Christian Garmann Johnsen Who Are the Post-Bureaucrats? A Philosophical Examination of the Creative Manager, the Authentic Leader 39. and the Entrepreneur
- Jacob Brogaard-Kay Constituting Performance Management 40. A field study of a pharmaceutical company
- 29. Rasmus Ploug Jenle Engineering Markets for Control: Integrating Wind Power into the Danish Electricity System
- 30. Morten Lindholst Complex Business Negotiation: Understanding Preparation and Planning
- 31. Morten Grynings TRUST AND TRANSPARENCY FROM AN ALIGNMENT PERSPECTIVE
- 32. Peter Andreas Norn Byregimer og styringsevne: Politisk lederskab af store byudviklingsprojekter

- 33. Milan Miric Essays on Competition, Innovation and Firm Strategy in Digital Markets
- 34. Sanne K. Hjordrup The Value of Talent Management Rethinking practice, problems and possibilities
- 35. Johanna Sax
 Strategic Risk Management
 Analyzing Antecedents and
 Contingencies for Value Creation
- 36. Pernille Rydén Strategic Cognition of Social Media
- 37. Mimmi Sjöklint
 The Measurable Me
 The Influence of Self-tracking on the User Experience
- 38. Juan Ignacio Staricco Towards a Fair Global Economic Regime? A critical assessment of Fair Trade through the examination of the Argentinean wine industry
 - Marie Henriette Madsen Emerging and temporary connections in Quality work
 - Yangfeng CAO Toward a Process Framework of Business Model Innovation in the Global Context Entrepreneurship-Enabled Dynamic Capability of Medium-Sized Multinational Enterprises
- 41. Carsten Scheibye Enactment of the Organizational Cost Structure in Value Chain Configuration A Contribution to Strategic Cost Management

- 1. Signe Sofie Dyrby Enterprise Social Media at Work
- 2. Dorte Boesby Dahl The making of the public parking attendant Dirt, aesthetics and inclusion in public service work
- 3. Verena Girschik Realizing Corporate Responsibility Positioning and Framing in Nascent Institutional Change
- 4. Anders Ørding Olsen IN SEARCH OF SOLUTIONS Inertia, Knowledge Sources and Diversity in Collaborative Problem-solving
- 5. Pernille Steen Pedersen Udkast til et nyt copingbegreb En kvalifikation af ledelsesmuligheder for at forebygge sygefravær ved psykiske problemer.
- 6. Kerli Kant Hvass Weaving a Path from Waste to Value: Exploring fashion industry business models and the circular economy
- 7. Kasper Lindskow Exploring Digital News Publishing Business Models – a production network approach
- 8. Mikkel Mouritz Marfelt The chameleon workforce: Assembling and negotiating the content of a workforce
- 9. Marianne Bertelsen Aesthetic encounters Rethinking autonomy, space & time in today's world of art
- 10. Louise Hauberg Wilhelmsen EU PERSPECTIVES ON INTERNATIONAL COMMERCIAL ARBITRATION

- 11. Abid Hussain On the Design, Development and Use of the Social Data Analytics Tool (SODATO): Design Propositions, Patterns, and Principles for Big Social Data Analytics
- 12. Mark Bruun Essays on Earnings Predictability
- 13. Tor Bøe-Lillegraven BUSINESS PARADOXES, BLACK BOXES, AND BIG DATA: BEYOND ORGANIZATIONAL AMBIDEXTERITY
- 14. Hadis Khonsary-Atighi ECONOMIC DETERMINANTS OF DOMESTIC INVESTMENT IN AN OIL-BASED ECONOMY: THE CASE OF IRAN (1965-2010)
- 15. Maj Lervad Grasten Rule of Law or Rule by Lawyers? On the Politics of Translation in Global Governance
- 16. Lene Granzau Juel-Jacobsen SUPERMARKEDETS MODUS OPERANDI – en hverdagssociologisk undersøgelse af forholdet mellem rum og handlen og understøtte relationsopbygning?
- 17. Christine Thalsgård Henriques In search of entrepreneurial learning – Towards a relational perspective on incubating practices?
- 18. Patrick Bennett Essays in Education, Crime, and Job Displacement

TITLER I ATV PH.D.-SERIEN

1992

1. Niels Kornum Servicesamkørsel – organisation, økonomi og planlægningsmetode

1995

2. Verner Worm Nordiske virksomheder i Kina Kulturspecifikke interaktionsrelationer ved nordiske virksomhedsetableringer i Kina

1999

3. Mogens Bjerre Key Account Management of Complex Strategic Relationships An Empirical Study of the Fast Moving Consumer Goods Industry

2000

4. Lotte Darsø Innovation in the Making Interaction Research with heterogeneous Groups of Knowledge Workers creating new Knowledge and new Leads

2001

5. Peter Hobolt Jensen Managing Strategic Design Identities The case of the Lego Developer Network

2002

- 6. Peter Lohmann The Deleuzian Other of Organizational Change – Moving Perspectives of the Human
- 7. Anne Marie Jess Hansen To lead from a distance: The dynamic interplay between strategy and strategizing – A case study of the strategic management process

2003

- Lotte Henriksen Videndeling

 om organisatoriske og ledelsesmæssige udfordringer ved videndeling i praksis
- 9. Niels Christian Nickelsen Arrangements of Knowing: Coordinating Procedures Tools and Bodies in Industrial Production – a case study of the collective making of new products

2005

10. Carsten Ørts Hansen Konstruktion af ledelsesteknologier og effektivitet

TITLER I DBA PH.D.-SERIEN

2007

1. Peter Kastrup-Misir Endeavoring to Understand Market Orientation – and the concomitant co-mutation of the researched, the re searcher, the research itself and the truth

2009

1. Torkild Leo Thellefsen Fundamental Signs and Significance effects

A Semeiotic outline of Fundamental Signs, Significance-effects, Knowledge Profiling and their use in Knowledge Organization and Branding

2. Daniel Ronzani When Bits Learn to Walk Don't Make Them Trip. Technological Innovation and the Role of Regulation by Law in Information Systems Research: the Case of Radio Frequency Identification (RFID)

2010

1. Alexander Carnera Magten over livet og livet som magt Studier i den biopolitiske ambivalens